World Development 138 (2021) 105229

Contents lists available at ScienceDirect

World Development

journal homepage: www.elsevier.com/locate/worlddev

Regular Research Article

COVID-19 response needs to broaden financial inclusion to curb the rise in poverty

Roxana Gutiérrez-Romero^{a,*}, Mostak Ahamed^b

^a Queen Mary University of London, School of Business and Management, London, UK ^b University of Sussex, Business School, University of Sussex, Brighton, UK

ARTICLE INFO

ABSTRACT

Article history: Available online 22 October 2020

JEL codes: G21 G22 I30 I32 C53

Keywords: Financial inclusion Poverty Inequality COVID-19 Forecasts

1. Introduction

The ongoing COVID-19 pandemic will have a profound health and economic impact, particularly in the developing world. Millions of people in these economies are employed in the informal sector often without regular access to welfare or pension rights (Gutiérrez-Romero, 2021). With the social distancing and lockdown measures implemented to control the spread of COVID-19 millions of people suddenly lost their livelihoods and can no longer rely on their daily earnings to survive. Urgent cash transfers and labour initiatives have been implemented in 181 countries to try mitigating some of the immediate economic impacts of the pandemic (Gentilini et al., 2020). However, other short-term and medium-term policies will be needed to help households receive government transfers and build financial buffers to spread resources over the likely prolonged crisis. Globally, there are 1.7

The ongoing COVID-19 pandemic risks wiping out years of progress made in reducing global poverty. In this paper, we explore to what extent financial inclusion could help mitigate the increase in poverty using cross-country data across 79 low- and lower-middle-income countries. Unlike other recent cross-country studies, we show that financial inclusion, particularly financial outreach, is a key driver of poverty reduction in these countries. This effect is not direct, but indirect, by mitigating the detrimental effect that inequality has on poverty. Our findings are consistent across all the different measures of poverty used and robust to instrumental variables. Our forecasts suggest that by 2021 the world's population living on less than \$5.50 dollars a day would increase by 231 million people, of which nearly 107.8 million people would be pushed into extreme poverty living on less than \$1.90 per day. However, urgent improvements in financial inclusion could substantially reduce the impact on poverty.

© 2020 Elsevier Ltd. All rights reserved.

billion adults without an account at a financial institution or a mobile money provider (World Bank, 2018). In this context, it is more important than ever to understand how severe the economic impact of the ongoing pandemic would be on poverty, and to what extent financial inclusion could contribute to reducing some of this impact.

In this paper, we address both questions. We start our analysis by assessing to what extent changes in poverty have been the product of improvements in income or distributional gains using cross-country data on poverty and inequality over the last two decades. To this end, we follow the poverty decomposition approach proposed by Datt and Ravallion (1992).¹ We then extend this decomposition regressions to estimate to what extent financial inclusion has played a direct effect on poverty reduction or an indirect effect by also mediating the impact that inequality and growth have on poverty. Various cross-country studies have found that financial inclusion is beneficial for poverty reduction, particularly when using aggregate measures of financial development such as private credit (Beck, Demirgüç-Kunt, & Levine, 2007) and in wealthy







^{*} Corresponding author at: Francis Bancroft Building, 4th Floor, School of Business and Management, Queen Mary University of London, Mile End, E1 4NS, UK.

E-mail addresses: r.gutierrez@qmul.ac.uk (R. Gutiérrez-Romero), M.Ahamed@-sussex.ac.uk (M. Ahamed).

¹ We take these series from PovcalNet, a dataset provided by the World Bank (Atamanov et al., 2018).

countries (Park & Mercado, 2018).² However, it is still unclear whether financial inclusion is indeed beneficial for poverty reduction in low- and low-middle income countries giving home to the majority of the population living in extreme poverty. Similarly, it is unclear how financial inclusion could be most effective if by improving financial outreach or financial usage. In this paper, we address these important gaps by assessing the link between poverty and an overall index of financial inclusion we construct using the Financial Access Survey over the period between 2004 and 2018 (IMF, 2019).³ We decompose this index into a sub-index of financial outreach and financial usage to ascertain how improvements in inclusion help poverty reduction the most. Based on this analysis, we then forecast changes in poverty using the latest IMF and World Bank projection of global growth in 2020 and 2021. We also use a range of forecast scenarios to estimate to what extent improvements in financial inclusion could curb increases in poverty. Our focus is on low- and low-middle income countries. Nonetheless, we also provide forecasts on global poverty, for all 121 countries included in our sample.

Our paper offers three key findings. First, we show that over the last two decades, poverty has been relatively unresponsive to economic growth and has been deeply affected by inequality. These findings are in line with earlier literature (Datt & Ravallion, 1992; van der Weide & Branko, 2018). Second, we show that financial inclusion does not directly reduce poverty; however, it strongly reduces the detrimental effect that inequality has on poverty. In other words, unlike other recent cross-country studies, we show that financial inclusion is a key element in reducing extreme forms of poverty in low- and lower-middle-income countries (e.g. Goksu et al., 2017; Park & Mercado, 2018). Moreover, we show that improvement in financial outreach is the most strongly associated with poverty reduction. Our results are robust to using the three components of poverty measures: incidence, intensity, and inequality among the poor and three different international poverty lines (\$1.90, \$3.20 and \$5.50 dollars a day). Since financial inclusion or lack thereof may be a consequence of the persistence of poverty, we show that our results are robust to using instrumental variables specifications (IV). As external instruments, we consider whether the country was colonised, the origin of their legal tradition as well as improvements in mobile phone subscription. A large body of literature has shown these historical conditions as well as improvements in communication technology influence financial development over time (Beck, Demirgüc-Kunt, & Martinez Peria, 2007; La Porta, Lopez-De-Silanes, Shleifer, & Vishny, 1997). Our findings are also robust to using alternative measures of financial inclusion taken from the World Bank's Global Findex, which suggest that improving financial inclusion towards men and women is beneficial, but particularly towards women.

Third, our forecast analysis suggests that the world's population living on less than \$5.50 dollars a day would increase by 231 million people, of which nearly 107.8 million people would be pushed into extreme poverty living on less than \$1.90 per day if no urgent and adequate measures are implemented. Our forecast analysis also suggests that with improvements in financial inclusion, particularly financial outreach, poverty increases could be curbed. We also provide policy implications and recommendations towards reducing the impact of COVID-19 on poverty. The remainder of the paper is structured as follows. The next section provides a brief overview of the literature. Section 3 describes the data and discusses our estimation strategy. Section 4 presents the empirical results, while Section 5 shows additional robustness checks. Section 6 shows the forecast analysis and Section 7 presents our conclusion.

2. Literature review

The empirical literature on financial inclusion falls broadly into three main categories. The first one uses randomised control experiments to ascertain the impacts of offering financial services or improving outreach to individuals, households, and firms. The second strand uses quasi-experiments and case studies using mostly *ad-hoc* measures of financial inclusion. The third strand evaluates the impacts of financial inclusion using cross-country aggregate analysis. What we can learn from this extensive literature is that the role of financial inclusion in poverty reduction is far from conclusive.

During the 1980s and the 1990s, it was widely believed that financial inclusion, particularly in the form of providing microcredits, could be vastly beneficial for poverty reduction (Morduch, 1999; Yunus, 2013). However, recent evidence of the micro-credit revolution stemming from randomised control trials is more nuanced. Micro-credits do improve people's ability to earn a living and help some to create and expand small businesses, but the evidence on poverty reduction is negligible. Systematic reviews of micro-credits have failed to find positive effects on household income (Duvendack, Palmer-Jones, & Vaessen, 2014; Stewart et al., 2012), including a meta-analysis from Grameen Bank micro-credits (Yang & Stanley, 2014). Although there are no dramatic changes in poverty reduction, there is no evidence either that these micro-credits lure vulnerable people into indebtedness, as some isolated anecdotal evidence might suggest (Banerjee, Duflo, Glennerster, & Kinnan, 2015).

The literature has also focused on understanding the extent to which broader financial inclusion can be pro-poor and what sort of broader inclusion is needed. Is it merely expanding the outreach, so that poor people can access financial services, or is increasing financial usage more beneficial? The evidence is again, inconclusive. From case studies, we know that increasing financial outreach can be beneficial for poverty reduction, even when outreach expansion might be motivated by political reasons (Cole, 2009). However, questions remain as to whether this is the most effective resource allocation instead of, for instance, direct cash transfers. For example, the largest mandated bank branch expansion established in rural areas in India during 1969–1990 helped to reduce poverty (Burgess & Pande, 2005). Still, the bank loan default rate was 40%, and questions remain about its cost-effectiveness, relative to potential alternative programmes.

Over the last decade, financial inclusion has also focused not only on improving access to credits, but also on broadening access to financial services, such as savings, insurance, and mobile banking (Cai et al., 2009; Dercon, 2005; Flory, 2018). The evidence from randomised control trials of improving household income, thanks to providing access to micro-savings and insurance, is promising. There is also evidence that expanding access to savings can particularly benefit those users that have been typically constrained and reduce sharp gender inequalities. For instance, the first randomised control trial of this kind provided access to non-interest-bearing bank accounts to young women and men in Kenya (Dupas & Robinson, 2013). The experiment showed that, despite hefty withdrawal fees, the majority of women used the accounts, and they were able to save more than men and to increase their investment and expenditures. This study suggests that women, particularly in

² An extensive empirical literature has analysed the link between financial inclusion and poverty reduction, ranging from randomised control trials, case studies and cross-country studies. See The next section briefly discusses the findings and gaps of that literature. See Beck, Demirgüç-Kunt, and Levine (2007) for a discussion of the theoretical literature on why financial inclusion could be beneficial for the wealthy and not neccesarily for the poor.

 $^{^{3}}$ In the Appendix, we present such indices by country during the 2004–2018 period.

rural areas, face negative private returns on money if they cannot find secure forms of saving. Similar findings have been found in Nepal and Malawi (Flory, 2018; Prina, 2015).

These recent randomised control trials suggest that financial inclusion might not only help poor people have more productive investments, reduce their consumption from idiosyncratic or local shocks but also help to reduce inequality. Recent quasi-experimental and case studies in developing countries such as in China, India, Nigeria and Ghana also suggest that increasing financial inclusion, in the form of increasing outreach and usage, can help to reduce household vulnerability to poverty, particularly in areas with financial services in distant places (Churchill & Marisetty, 2020; Dimova & Adebowale, 2018; Koomson, Villano, & Hadley, 2020; Li, 2018). But a remaining question is whether financial inclusion helps reducing poverty directly or via its impact on reducing income and gender inequalities. From cross-country studies, there is some mixed evidence. For instance, Goksu, Deléchat, Newiak, and Yang (2017), using a micro-data set across 140 countries, found a non-linear relationship between financial inclusion and inequality. Their findings suggest that in earlier stages of development, only a small group, the wealthy, benefit from financial inclusion progress, but with a broader level of financial inclusion gradually all other groups benefit. Similarly, using cross-country analysis, Park and Mercado (2018) show that financial inclusion is positively associated with lower levels of poverty in high- and upper-middle-income economies, but not in middle-low and low-income economies. This mixed evidence is perhaps unsurprising, given that high-income economies have a broader welfare system and better regulatory conditions that can further the impact of financial inclusion. However, this earlier analysis needs to be broadened to understand the main factors that drive poverty changes and how financial inclusion might affect poverty, whether directly or indirectly. In this respect, the literature has found two challenges. The first regards how the multidimensional aspects of financial inclusion should be measured, and the second, how to estimate whether financial inclusion affects poverty directly or indirectly by dampening the detrimental effects of inequality.

Over the last decade, the literature has proposed several different measures of financial inclusion, mainly drawing from individual financial surveys, or drawing from the global financial surveys conducted by the World Bank or the IMF (Ahamed & Mallick, 2019). The World Bank has recently made available the Global Financial Inclusion database, which provides information on more than 850 indicators across 151 economies, focusing on the demand side of financial services (Demirgüc-Kunt, Klapper, Singer, Ansar, & Hess, 2018). A main constraint of this database is its periodicity, only available for 2011, 2014, and 2017, which does not allow for an extended comparative analysis across countries over time. An alternative source is the Financial Access Survey, 2004–2018 gathered by the IMF (IMF, 2019). This annual series offers the largest global supply-side data on financial inclusion, including data on access to and usage of financial services by both firms and households that are comparable across countries and over time.

As reviewed here, all strands of the empirical literature are equally relevant for policy analysis. From the experimental literature and quasi-experimental case studies, we learn that financial inclusion needs to consider more than micro-credits. We also learn that financial inclusion might not reduce poverty directly, but indirectly by reducing inequalities in financial access and by broadening financial usage among typically disadvantaged groups. Cross-country analysis has a different advantage. It allows one to use the same measure of financial inclusion to make comparative analysis across countries and over time. Cross-country analysis can also estimate the likely direct and indirect impacts of financial inclusion on poverty. Next, we take advantage of the poverty-growth-inequality decomposition method proposed by Datt and Ravallion (1992) to understand whether financial inclusion affects poverty reduction directly or indirectly. At the aggregate level, poverty from one period to the next might change as a result of changes in the Gross Domestic Product (GDP), ceteris paribus, or whether GDP is distributed any differently. These simple poverty decompositions, theoretically underpinned by Lorenz Curve principles, can be empirically estimated. Extensive research has shown that inequality in particular is detrimental for poverty reduction since increases in GDP are often captured by middle or upper classes, with a limited trickledown effect for the poor (Gutiérrez-Romero & Méndez-Errico, 2017; Ravallion, 2005; van der Weide & Branko, 2018).

In our empirical analysis, we extend these decomposition regressions to include the potential role of financial inclusion in poverty reduction. Our first hypothesis is whether financial inclusion might affect poverty reduction directly. Our second hypothesis is that financial inclusion might reduce poverty indirectly by mitigating the detrimental effect that inequality has on poverty. To test our two key hypotheses, we construct a financial inclusion index along two associated dimensions of financial outreach and financial usage.

3. Empirical Strategy: Data and method

3.1. Data sources

We use three main data sources. First, to construct the financial inclusion index, described in sub-Section 3.2, we use the Financial Access Survey (FAS) database, available during the 2004–2018 period (IMF, 2019). As an alternative indicator of financial inclusion, we also use the percentage of the population that owns a bank account, taken from the World Bank's Global Findex database, available during the 2011–2018 period (Demirgüç-Kunt et al., 2018).

Our second key data source is on countries' annual GDP growth rate from 2004 until 2019. In Section 6, we forecast global levels of poverty post-COVID-19 for which we use the latest IMF economic growth forecasts (at the time of this writing) for the years 2020 and 2021 (IMF, 2020b). As a robustness check of our forecasts, we also use the latest World Bank economic growth projections for the years 2020 and 2021 (World Bank, 2020).

Our third key data source is PovcalNet, which provides inequality and poverty statistics, at the national level, based on over two million randomly sampled households around the globe since the 1980s. We constrain our analysis to the period 2004-2018 where we also have data on financial inclusion. Following standard practice in the literature, we measure the incidence of poverty in the poorest countries of the world using three international poverty lines of \$1.90, \$3.20 and \$5.50 dollars a day (at 2011 purchasing power parity).⁴ Specifically, we use the poverty headcount ratio which measures the proportion of the population who are poor; the poverty gap index which measures the depth of poverty by considering how far, on average, the poor are from that poverty line, expressed as a percentage; the poverty gap squared, similar to the earlier measure but based on the sum of squared poverty deficits; and the Watts poverty index which is more sensitive to changes in the lowest part of the income distribution. That is, transfers to the poorest of the poor counts more in the Watts index in terms of poverty reduction than transfers to the relatively wealthier of the poor.⁵

⁴ For any gap year on poverty statistics, we use the poverty statistics from the earlier waves of the household/consumption survey.

⁵ The Watts index is estimated as $\frac{1}{N}\sum_{j=1}^{q} \ln\left(\frac{z}{y_j}\right)$, where *N* denotes the total population, *q* the total population of the poor living under the poverty line, *z*, and *y_j* is the income of the poor individual *j*.



Financial Inclusion Index 2018

Fig. 1. Financial inclusion around the globe. Source: Own estimates using 2019 Financial Access Survey (IMF, 2019).

3.2. Constructing the financial inclusion index

We construct an overall index of financial inclusion which we decompose into two sub-indices of financial outreach and financial usage. This approach follows the advice of the recent literature that has suggested to focus on these two dimensions to measure progress made in financial inclusion (Ahamed & Mallick, 2019; Amidžic, Massara, & Mialou, 2014).

Our sub-index of financial usage includes the number of bank accounts per 1,000 people (that is the sum of loan and deposit accounts), intended to measure the depth of the financial access.⁶ We estimate the sub-index of financial outreach using principal component analysis that helps us to combine the four indicators described next (Fo1, Fo2, Fo3 and Fo4). Following the approach suggested by Beck, Demirgüc-Kunt, and Martinez Peria (2007), to capture the demographic penetration of bank branches we include the number of bank branches (Fo1) and the number of ATMs per 100,000 people (Fo2). To capture the geographic penetration of bank branches, we include the number of bank branches (Fo3) and the number of ATMs per 1,000 square kilometres (Fo4). Thus, this financial outreach sub-index helps to capture the physical proximity to the point of financial services, which is considered to be one of the most important impediments to inclusive financial development (Allen et al. 2014).

After estimating the financial outreach dimension using principal component analysis, we then construct the overall index of financial inclusion using again principal component analysis, which helps us to weight the financial outreach and financial usage sub-indices, as expressed in Eq. (1).⁷

Financial inclusion =
$$\sum_{i=1}^{n} w_{ij} X_i$$
 (1)

where w_{ij} are the component's loadings or weights, and X_i are the financial outreach and financial usage dimensions.⁸ To ease interpretation, we normalise the overall index of financial inclusion and assign each country a score from 0 (lowest) to 1 (highest) scale of inclusive financial development.

Fig. 1 shows the overall index of financial inclusion around the globe. In the Appendix, in Tables A1–A3, we provide the financial inclusion index on an annual basis for each country for which there are available data, including the two associated dimensions of financial outreach and financial usage from 2004 to 2018. In Table A1, we also include a category of whether the country is low, lower-middle, upper-middle, or high-income. Since some countries have changed income category over time, we present this income category for the year 2018 only. The distinction of income levels is particularly relevant for our analysis because as explained in the next section, we focus on low- and low-middle income countries given that they concentrate most of the population living in extreme levels of poverty (according to all the poverty lines used here of \$1.90, \$3.20 and \$5.50 dollars a day).

⁶ As the data on the number of people having bank accounts are limited, we use the number of bank accounts per capita where double counting cannot be eliminated if a person has multiple accounts. For further discussion on this see Beck, Demirgüc-Kunt, and Martinez Peria (2007). Note that data on all indicators in constructing a financial inclusion index are based on commercial banks. We also winsorise each indicator at the 95th percentile levels to reduce the influence at the upper tail. We also normalise each indicator to have values between zero and one.

⁷ In constructing the overall index of financial inclusion we find that the first principal component (the sub-index of financial outreach) explains about 85% of the corresponding sample variance of the two components included, with an eigenvalue of more than one, that is, 1.70.

⁸ As a robustness check we also aggregated the financial inclusion index in an alternative way, by allowing all the components of the financial outreach (Fo1, Fo2, Fo3 and Fo4) and financial usage sub-indices to dictate the natural intercorrelation of the one summary index. Using this alternative financial inclusion index, we re-run all our analysis finding no differences in our core results. This is not surprising since this alternative financial inclusion for 0.98 to the overall financial inclusion presented here. Since the results are identical, in sub-section 5.2 as a robustness check we present instead our results using an alternative index of financial inclusion drawn from the World Bank Global Findex data.

3.3. Descriptive statistics

After combining the inequality (Gini index) and poverty statistics with the financial inclusion indices, we remain with 121 countries and 1,694 observations for the 2004–2018 period. We report the overall descriptive statistics in Table 1 over the whole 2004– 2018 period. We also present separately the statistics for the 79 low-and lower-middle-income countries contained in our sample and 66 upper-middle and high-income countries. Note that these countries do not add up to the total of countries included in our sample (121) because during the period analysed some countries changed of income category, and these countries, therefore, might appear in different income categories over time, albeit in different years only.

Table 1 shows that over the whole period analysed, only 2.6% of the population in upper-middle and high-income countries lived on less than \$1.90 dollars a day. In sharp contrast, an average of 27% of the population in low- and lower-middle-income countries lived on less than \$1.90 dollars a day. People in these poorer countries also faced significant disadvantages in terms of experiencing higher levels of inequality and significantly lower levels of financial inclusion than people in wealthier countries. In Section 5, we discuss further the statistics related to the other two poverty lines of \$3.20 and \$5.50 dollars a day. In Section 6 we show in further depth the progress made in reducing poverty over the period analysed, and the likely rises in poverty expected as a result of the COVID-19 economic crisis.

Table 2 shows the percentage of the population that own a bank account, according to the Global Findex database on average during 2011–2018, which can be interpreted as an alternative indicator of financial inclusion. This table shows there are sharp differences in financial inclusion between wealthier countries and those low-and lower-middle-income countries. Moreover, although in general, women have a lower percentage of bank ownership than men,

Table 1

Summary statistics of poverty and financial inclusion over 2004-2018.

women in low- and lower-middle-income countries are significantly constrained from these financial services (see Table 2).

Going back to the overall index of financial inclusion constructed for this paper, at the country level, we also note that, on average for the period of analysis 2004–2018, low- and lower-middle-income countries with higher levels of overall financial inclusion were associated with lower levels of poverty headcount ratio (using the \$1.90 a day) and lower levels of income inequality (Fig. 2). In the next section, we assess this potential relationship between financial inclusion, poverty and inequality over the period 2004–2018, which can inform the current policy debate on whether financial inclusion development is indeed related to poverty reduction, and the potential mechanisms involved. To do so, we first apply the poverty decomposition proposed by Datt and Ravallion (1992), which we then extend to include the indicator of financial inclusion.

3.4. Estimation framework

We start by applying the decomposition methodology proposed by Datt and Ravallion (1992) to examine to what extent poverty changes observed at the national level can be ascribed to changes in economic growth or income inequality. From the theoretical properties of the Lorenz curve, these authors show that the levels of poverty from one period to the next may change due to three key components. These are either a change in the mean income while holding the Lorenz curve constant at that of the reference year, the so-called growth component; or due to a change in the Lorenz curve, while holding the mean income constant, referred to as the redistribution component; or due to changes in a residual term that capture changes in poverty due to the interaction between these two other terms. Thus, at the national level, a change in poverty from one period to the next one can be expressed as in Eq. (2).

	All countries		Low- and low income coun	ver-middle- tries	Upper-middle and high- income countries			
	Mean	St dev.	Mean	St dev.	Mean	St dev.		
Headcount under \$1.9 a day	0.156	0.219	0.270	0.244	0.026	0.053		
Poverty gap under \$1.9 a day	0.060	0.101	0.105	0.121	0.009	0.019		
Poverty gap squared under \$1.9 a day	0.032	0.061	0.056	0.076	0.005	0.012		
Watts index under \$1.9 a day	0.091	0.168	0.161	0.205	0.011	0.024		
Gini	0.395	0.084	0.412	0.075	0.375	0.090		
GDP growth rate	0.037	0.038	0.045	0.039	0.028	0.035		
Financial inclusion index	0.267	0.230	0.124	0.117	0.428	0.219		
Financial outreach	0.225	0.221	0.102	0.099	0.365	0.237		
Financial usage	0.320	0.288	0.153	0.159	0.510	0.283		
Number of countries	121		79		66			
Number of observations	f observations 1694		899		795			

Source: Own estimates using PovcalNet and Financial Access Survey (IMF, 2019).

Table 2

Summary statistics of account ownership over 2011-2018.

	All countries		Low- and low income count	er-middle- ries	Upper-middle and high- income countries			
	Mean	St dev.	Mean	St dev.	Mean	St dev.		
Account ownership	0.473	0.296	0.269	0.165	0.686	0.247		
Account ownership (Male)	0.503	0.291	0.300	0.170	0.716	0.233		
Account ownership (Female)	0.443	0.305	0.238	0.168	0.657	0.266		
Number of countries	108		68		60			
Number of observations	1512	1512			739			

Source: Own estimates using World Bank's Global Findex database (Demirgüç-Kunt et al., 2018).



Panel A

Panel B

Fig. 2. Financial inclusion, poverty, and inequality in low- and lower-middle-income countries.

$$P_{f+n} = P_f = G\underbrace{(f, f+n; r)}_{Growth} + \underbrace{D(f, f+n; r)}_{Redistribution} + \underbrace{R(f, f+n; r)}_{Residual}$$
(2)

where *P* denotes the poverty measure at date *f*, or f + n that can be fully characterised by a poverty line *z*, and the Lorenz curve. G(.); D (.) and R(.) stand for the growth, redistribution, and residual components. The last argument *r* denotes the reference date with respect to which the observed change in poverty is being decomposed. According to Datt and Ravallion (1992), the residual term can be interpreted as the difference between the growth (redistribution) components evaluated at the terminal and initial Lorenz curves. Therefore, if the mean income or the Lorenz curve remain unchanged over the decomposition period, the residual vanishes.

This poverty decomposition can be easily implemented by regressing changes in poverty on changes in income inequality, proxied by changes in the Gini index, and on changes in the real Gross Domestic Product growth, as shown in Eq. (3).⁹ Note that the decomposition technique is not intended to prove causality, nor to account for the causal determinants of the differences in the levels of poverty, which can be numerous. Instead, this regression, based on the theoretical properties of the Lorenz curve, estimates to what extent changes in poverty at the macro-level can be attributed to changes in economic growth, redistribution changes, or the interaction between these two. Since this regression is usually run on several years, the literature usually merely adds year fixed effects as controls to capture to what extent fluctuations in particular years exacerbate changes in poverty, an approach that we follow here (Freije, 2014).

$$\Delta P_{it} = \alpha_i + \beta \Delta Gini_{it} + \gamma GDP \ Growth_{it} + \varphi year_t + \varepsilon_{it}$$
(3)

where ΔP_{it} denotes annual change in poverty in country *i* in annual period *t*. We separately use four measures of poverty (headcount ratio, poverty gap, poverty gap squared, and the Watts index). Δ Gini denotes annual changes in the Gini index, and φ represents the regression coefficient for the year fixed effects. We estimate this regression using panel fixed effects at the country level, clustering the Huber–White standard errors at the country level.

To consider to what extent changes in financial inclusion affect changes in poverty, we amend this poverty decomposition regression. We do so by adding the annual change in the financial inclusion index and the interaction between changes in the financial inclusion index and the change in the Gini index, as shown in Eq. (4). We add these two factors to understand whether improvements in financial inclusion contribute directly towards poverty reduction or indirectly by dampening the detrimental effect of increases in inequality.¹⁰

$$\Delta P_{it} = \alpha_i + \beta \Delta Gini_{it} + \gamma GDP \ Growth_{it} + \xi \Delta Financial \ inclusion_{it} + \eta \Delta Financial \ inclusion_{it} * \Delta Gini_{it} + \varphi year_t + \varepsilon_{it}$$
(4)

where Δ Financial inclusion_{it} stands for the change in the overall financial inclusion index in country *i* in annual period *t*. To understand further how financial inclusion might be helping poverty, we run three separate regressions, using either the change in the overall index of financial inclusion, or the change in the sub-index of financial outreach, or the change in the sub-index of financial usage. The regression coefficient η , is the interaction between changes in financial inclusion and change in inequality. A negative

⁹ This poverty decomposition tool has been used to forecast how changes in expected levels of economic growth, or inequality could affect poverty changes (Ravallion, 2013).

¹⁰ As mentioned in Section 4, in separate regressions we also added the interaction between changes in economic growth and changes in the financial inclusion index. However, we found this interaction to be statistically insignificant.

interaction coefficient η would suggest that improvements in financial inclusion would have a larger contribution to poverty reduction in countries with higher levels of income inequality. We estimate this regression using panel fixed effects at the country level, clustering the Huber–White standard errors at the country level.

This panel fixed effects regression can help us to forecast to what extent a major drop in GDP growth, such as the one expected as a result of COVID-19, or likely changes in inequality could impact poverty in the near future. More importantly, for policy insights, we can also learn to what extent improvements in financial inclusion, could help to reduce poverty directly or indirectly.

Our panel fixed effects regression specification helps to mitigate some potential concerns with endogeneity, particularly for any omitted time-invariant regressors that might be correlated with the error term. Nonetheless, we acknowledge that a potential limitation of this approach could be a potential reverse causality between changes in financial inclusion and changes in poverty. That is, changes in financial inclusion, or lack thereof, may be a consequence of changes in poverty over time, which are perhaps more likely to occur in poorer countries where with some exceptions, financial inclusion is still not widely spread. To test for and address such potential endogeneity bias we will also run a separate instrumental variable specification. As shown in Section 4, although we find some evidence of endogeneity our core conclusions remain the same as from our benchmark panel fixed specifications.

Specifically, to test and account for endogeneity we use a panel random effects (RE) model with two-stage least squares (IV-2SLS) instrumental variables. This RE model has the flexibility of simultaneously modelling both time-variant and time-invariant effects, unlike panel fixed effects (Bell & Jones, 2015). Eqs. (5) and (6) represent the first-stage regressions of our two potential endogenous variables, Δ Financial inclusion and therefore, its interaction term Δ Financial inclusion^{*} Δ Gini.

$$\Delta Financial inclusion_{it} = \mu_{1i} + \mu_2 Z_i + \mu_3 Z_{it} + \mu_4 \Delta Gini_{it} + \mu_5 GDPGrowth_{it} + \mu_6 year_t + u_{it}$$
(5)

$$\Delta Financial\ inclusion_{it} * \Delta Gini_{it} = \lambda_{1i} + \lambda_2 Z_i + \lambda_3 Z_{it} + \lambda_4 \Delta Gini_{it} + \lambda_5 GDPGrowth_{it} + \lambda_6 year_t + v_{it}$$
(6)

where Z_{it} and Z_i refer to the three external instruments used, u_{it} and v_{it} refer to the disturbance terms. The first instrument used, denoted by Z_{it} is the number of mobile phone subscriptions, per 10,000 people, ¹¹which varies on an annual basis during our sample period, and has been shown to be strongly associated with financial development (Beck, Demirgüç-Kunt, & Martinez Peria, 2007).¹² Since new mobile phone subscriptions are unlikely to turn immediately into more financial inclusion, we lag our instrument by two periods. We also use as external instruments the origin of countries' legal code and whether they were former colonies. Earlier research has shown that these external instruments account for substantial cross-country differences in financial development (La Porta et al., 1997, 2008; Beck et al., 2000). The origin of countries' legal code

can be broadly classified into two, the English common law and the civil law tradition (along its several sub-traditions, the French, Socialist, German and Scandinavian). These legal traditions are derived from the Roman law, which were developed over time into the distinct schools we know today and which through colonialism and occupation were transplanted from few countries to the rest of the world (Beck et al., 2000). Thus, we treat these legal traditions, as well as who was the Western coloniser (if any), as exogenous 'endowments' which up to date continue to affect financial development during our sample period.

The second-stage IV random-effects model estimates the impact of changes in financial inclusion on changes in poverty, as shown in Eq. (7).

$$\Delta P_{it} = \kappa_1 + \kappa_2 \Delta Gini_{it} + \kappa_3 GDP \ Growth_{it} + \kappa_4 DFinancial \ inclusion_{it} + \kappa_5 \Delta Financial \ inclusion_{it} * \Delta Gini_{it} + \kappa_6 year_t + (\kappa_i + e_{it})$$
(7)

where κ_4 and κ_5 are the regression coefficients of the instrumented endogenous variables. The so-called 'random' part of the model in parenthesis consists of the residual κ_i for country *i*, which allows for differential intercepts for countries, and e_{it} represents the time-varying residuals.

4. Results

4.1. Poverty decomposition

Table 3 reports the estimation results of the poverty decomposition shown in Eq. (3), estimated with a panel fixed effect specification, and robust standard errors clustered at the country level. This regression assesses the extent to which annual changes in poverty rates are attributable to changes in economic growth or distribution of income. We use four separate dependent variables. That is the change (denoted by Δ) of the poverty headcount ratio, poverty gap, poverty gap squared, and the Watts index. In columns 1-4, we estimate the poverty decomposition for all 121 countries in our sample for the whole period 2014–2018. In columns 5–8, we focus on only the 79 low- and lower-middle-income countries in the sample, and the remaining columns 9-12 refer to uppermiddle and high-income countries only. All these poverty statistics refer to the \$1.90 dollars a day poverty line. In Section 5, as a robustness check, we re-run our results for the poverty lines of \$3.20 and \$5.50 dollars a day.

The results show that the redistribution component (change in the Gini index) is statistically significant, suggesting that increases in inequality contribute to rises in poverty. The strength of this association is stronger for low- and lower-middle-income countries than for wealthier countries.¹³ In contrast, the regression coefficient of economic growth is statistically insignificant for all columns.

In Fig. 3, we plot the marginal effects of the regression coefficients shown in column 5 for low- and lower-middle-income countries. These marginal effects illustrate the null effect of growth on poverty reduction, but the strong effect of changes in inequality. Overall, these results imply that the poverty reduction seen over the whole period analysed 2004–2018 has been mostly driven by redistribution of income towards the poor.

 $[\]overline{}^{11}$ We take data on mobile phone subscriptions from the World Bank repository data.

¹² Higher mobile phone subscriptions help improve communication infrastructure, and cover formerly unbanked people by developing new ways to make a deposit and withdraw cash from an account, such as mobile money transfers (Allen et al., 2014). Better communication systems also help to lower the delivery cost of banking services and increase financial usage particularly in low- and lower-income countries that tend to suffer from limited physical and financial infrastructure (Aker & Mbiti, 2010).

¹³ Not shown in table 3, the magnitude of standardised regression coefficient for the change in the Gini index is about 0.50 for low-and lower-middle income countries, therefore of medium magnitude. In contrast, this standardised coefficient is about 0.20 for upper-middle and high-income countries, considered of small magnitude according to Cohen's (1988) guidelines.

Povertv	growth-redistribution	decomposition	of those	living under	\$1.90 dollars a day	1.
	a				+	

_	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)
_	All countries				Low- and low	ver-middle	-income countrie	es	Upper-middl	e and high	-income countri	es
	∆Headcount	∆Poverty gap	Δ Poverty gap squared	Δ Watts	∆Headcount	Δ Poverty gap	Δ Poverty gap squared	∆Watts	∆Headcount	∆Poverty gap	Δ Poverty gap squared	ΔWatts
∆Gini	0.604***	0.378***	0.273***	0.679***	0.738***	0.500***	0.359***	0.986***	0.411***	0.207***	0.151***	0.251***
	(0.162)	(0.093)	(0.064)	(0.187)	(0.278)	(0.156)	(0.107)	(0.297)	(0.074)	(0.053)	(0.046)	(0.054)
GDP growth	-0.011	-0.006	-0.005	-0.009	0.010	-0.002	-0.004	-0.007	-0.029	-0.010	-0.007	-0.008
rate	(0.022)	(0.013)	(0.009)	(0.024)	(0.036)	(0.020)	(0.014)	(0.037)	(0.019)	(0.010)	(0.006)	(0.015)
Constant	-0.004^{*}	-0.002	-0.001	-0.003	-0.006^{*}	-0.003^{*}	-0.002	-0.004	-0.002	-0.001	0.000	-0.002
	(0.002)	(0.001)	(0.001)	(0.002)	(0.003)	(0.002)	(0.001)	(0.003)	(0.003)	(0.001)	(0.001)	(0.002)
Year fixed effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Observations	1,694	1,694	1,694	1,694	899	899	899	899	795	795	795	795
Adjusted R ²	0.140	0.155	0.156	0.141	0.137	0.169	0.169	0.179	0.291	0.293	0.303	0.237
Number of countries	121	121	121	121	79	79	79	79	66	66	66	66

Note: Panel fixed effects model. Δ indicates change from one period to the next. Robust standard errors, clustered at the country level in parentheses. Significance level *** p < 0.01, ** p < 0.05, * p < 0.1.

Source: World Bank's PovcalNet database and IMF. Coverage: 2004–2018 period.



Fig. 3. Marginal effects of change in inequality and growth on changes on poverty headcount ratio in low- and low-middle-income countries.

4.2. The role of financial inclusion

We move on to assess the role of improvements in financial inclusion on poverty reduction. Since our interest is particularly on extreme levels of poverty, we focus exclusively on the 79 low- and lower-middle-income countries in our sample, which are home to the majority of the population living under a \$1.90 dollars a day.

In Table 4, we start by showing the panel fixed effects specification as expressed in Eq. (4). As before, we use four separate measures of poverty as dependent variables, all referring to the \$1.90 dollars a day poverty line. For each of these poverty measures, we separately estimate the association between changes in the overall index of financial inclusion, the sub-index financial outreach, and the sub-index financial usage.

Table 4 shows that changes in inequality (proxied by changes in the Gini index) are positive and statistically significantly associated with changes in poverty, as our earlier estimations had shown. Similarly, we find that GDP growth rate is not statistically associated with changes in poverty.

Growth-redistribution decomposition of poverty in low- and low-middle income countries: the role of financial inclusion.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)
Dependent variable:	Δ Headco	unt		Δ Poverty	gap		Δ Poverty	gap square	d	Δ Watts		
ΔGini	0.961***	0.953***	0.866***	0.638***	0.641***	0.575***	* 0.454***	0.459***	0.410***	1.246***	1.257***	1.127***
	(0.309)	(0.313)	(0.289)	(0.175)	(0.181)	(0.162)	(0.121)	(0.126)	(0.112)	(0.340)	(0.351)	(0.314)
GDP growth rate	0.007	0.011	0.009	-0.002	-0.001	-0.002	-0.003	-0.003	-0.003	-0.007	-0.006	-0.006
	(0.035)	(0.035)	(0.035)	(0.020)	(0.020)	(0.020)	(0.013)	(0.013)	(0.014)	(0.036)	(0.036)	(0.036)
Δ Financial inclusion	0.136*			0.044*			0.019			0.058		
	(0.072)			(0.025)			(0.015)			(0.041)		
Δ Gini $\times \Delta$ Financial inclusion	-34.622*	k .		-21.368*	*		-14.643**	*		-40.275**	•	
	(18.028)			(9.788)			(6.555)			(18.275)		
Δ Financial inclusion outreach		0.040			0.019			0.013			0.033	
		(0.058)			(0.032)			(0.022)			(0.059)	
Δ Gini $\times \Delta$ Financial inclusion outreach	1	-39.437*	ŧ		-25.917*	*		-18.275*	*		-49.745*	k
		(19.753)			(11.718)			(8.093)			(22.317)	
Δ Financial inclusion usage			0.089*			0.025			0.009			0.028
			(0.050)			(0.019)			(0.011)			(0.030)
Δ Gini $ imes \Delta$ Financial inclusion usage			-16.387			-9.767			-6.576*			-18.221^{*}
			(11.252))		(5.940)			(3.886)			(10.785)
Constant	-0.006^{*}	-0.006^{*}	-0.006^{*}	-0.003^{*}	-0.003^{*}	-0.003*	-0.002	-0.002	-0.002	-0.004	-0.004	-0.004
	(0.003)	(0.003)	(0.003)	(0.002)	(0.002)	(0.002)	(0.001)	(0.001)	(0.001)	(0.003)	(0.003)	(0.003)
Year fixed effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Observations	899	899	899	899	899	899	899	899	899	899	899	899
Adjusted R ²	0.163	0.157	0.155	0.194	0.193	0.184	0.191	0.191	0.181	0.203	0.202	0.192
Number of countries	79	79	79	79	79	79	79	79	79	79	79	79

Note: Panel fixed effects model. Δ indicates change from one period to the next. Robust standard errors, clustered at the country level in parentheses. Significance level *** p < 0.01, ** p < 0.05, * p < 0.1.

Source: World Bank's PovcalNet database and IMF. Coverage: 2004-2018 period.

We find some positive association between the changes in financial inclusion and changes in poverty (columns 1 and 4). However, the net effect of financial inclusion¹⁴ is negative, given that the interaction between the change in financial inclusion and change in the Gini index is negative and large in magnitude. This interaction is statistically significant for all the measures of poverty used, suggesting that the detrimental effect of inequality on poverty is strongly dampened in countries that have a greater level of financial inclusion. Also, the interactions between changes in Gini and financial outreach are statistically significant for all poverty measures. This interaction with the financial usage index is however, only significant in columns 9 and 12, for the squared poverty gap and the Watts index. These results suggest that the financial outreach dimension is the one that contributes the most to reducing the detrimental effect of inequality on poverty. As a robustness check, we have also estimated alternative specifications where we interacted the GDP growth rate with changes in the financial inclusion index (or either of its two dimensions). These alternative interactions turned statistically insignificant (reason why we do not report them), findings that suggest financial inclusion helps poverty reduction via dampening the damaging effect of inequality but not through boosting the effect of economic growth.

In Fig. 4, we plot the marginal effects of all the regression coefficients shown in column 1. These marginal effects illustrate a strong association between poverty reduction and financial inclusion by reducing the detrimental effect of inequality.

4.3. Addressing endogeneity concerns

We acknowledge that the associations between changes in financial inclusion and poverty might suffer from endogeneity bias.

Therefore, to address this potential concern, we continue by presenting our instrumental variable specifications.

The first-stage relationship between our instruments (country's coloniser, origin of legal code, and two-year lagged mobile phone subscriptions) and our potential endogenous variables (change in financial inclusion and its interaction with Gini) is strong as expected (see Table A4). That is, in line with earlier literature, improvements in mobile phone subscription lead to higher levels of financial inclusion. Also, countries with the civil law tradition have worse levels of financial inclusion than countries with the common law legal tradition, and colonial origin influences financial inclusion. The two-stage panel random IV specifications are shown in Table 5. The bottom of this table shows the Sargan-Hansen overidentification test, which suggests our instruments are valid. That is, as the null hypothesis of this test suggests, the instruments are uncorrelated with the error term, and that the excluded instruments are correctly excluded from the second-stage regression. We also report the endogeneity test showing that there are unobserved factors correlated with the error, therefore the need for instrumental variable model specifications.

Despite finding evidence of endogeneity, Table 5, our secondstage IV estimations confirm our core messages. That is, changes in inequality are strongly associated with changes in poverty, whereas economic growth is not. However, this time we find that only the interaction between changes in the Gini index and the financial inclusion outreach index is statistically significant, and again negative in sign, for each of our four dependent variables. The magnitude of these IV interactions, although with the same sign, are larger than the panel fixed specifications shown earlier. We conclude that our earlier panel fixed effects regression offers more conservative estimates of the association between financial inclusion outreach and poverty reduction.

To assess the magnitude of the IV effects, Table A5 re-estimates our results but using standardised coefficients. That is, we de-mean all our continuous dependent and independent variables and divide them by their standard deviation. In this way, the regression

¹⁴ The net effect of financial inclusion is given by the sum of the regression coefficient for financial inclusion and its interaction term with the change in the Gini index.



Fig. 4. Marginal effect of growth, change in inequality, and change in financial inclusion on poverty in low- and low-middle income countries.

coefficients represent effects in terms of standard deviations. The standardised coefficients for changes in the Gini coefficient on poverty change are mostly between 0.50 and 0.80, therefore would be considered of medium magnitude following Cohen's (1988) guide-lines. The standardised interaction coefficients between the change in the Gini coefficient and the financial inclusion outreach indices are greater than 0.90, therefore of large magnitude.

5. Additional checks

5.1. Using alternative poverty lines

All our earlier findings focused on the poverty line of \$1.90 dollars a day. In this section, we focus instead on those living under \$3.20 and \$5.50 dollars a day. Although, we do not necessarily expect to find exactly the same relationship between our key explanatory variables and changes in poverty, it is worth assessing whether improvements in financial inclusion could lead to similar reductions in less extreme forms of poverty, and whether through similar channels, such as improvements in financial outreach.

Table A6 shows a substantial percentage of the population lives under \$3.20 and \$5.50 dollars a day in low- and low-middle income countries, that is 48% and 70% respectively. In contrast, only a small percentage of the population lives under these poverty lines in upper-middle and high-income countries, only 6.5% and 15.4% respectively.

Tables A7 and A8 show the poverty decomposition using the \$3.20 and \$5.50 a day poverty lines respectively using Eq. (3). Columns 1–4 refer to all countries in our sample, columns 5–9 to lowand low-middle income countries only, and columns 10–12 to wealthier countries only. As before, we find that economic growth has a null effect on poverty changes in low-middle income countries. Similarly, increases in inequality lead to higher levels of poverty. However, this time we find that the regression coefficient of changes in inequality is not statistically significant when using as dependent variable the change in the poverty headcount ratio in low- and low-middle income countries, but it is in other countries, as well as for the other three poverty measures.

Then we proceed to estimate the relationship between changes in financial inclusion and poverty, using the same instrumental variables we used earlier on, again focusing on low- and low-middle income countries.

The first-stage regression for the \$3.20 dollar a day poverty line is shown in Table A9, and the first-stage regression for the \$5.50 dollar a day poverty line is in Table A11. The respective secondstage IV panel random effects regressions are reported in Tables A10 and A12. As shown at the bottom of these tables, once again we find evidence of endogeneity, and that our instruments are valid according to the Sargan-Hansen overidentification statistics.

The interaction between changes in financial inclusion index and the Gini index is statistically insignificant for all our poverty measures using the \$3.20 and \$5.50 dollars a day poverty lines (Tables A10 and A12). Focusing on the \$3.20 dollars a day, we find that the interaction between changes in Gini and financial outreach is statistically significant for all poverty measures except for the poverty headcount. For the \$5.50 dollars a day, we find that the interaction between the Gini index and financial outreach is once again statistically significant, and negative, but only for the poverty gap square and the Watts index. These results strongly suggest that improvements in financial outreach help to mitigate the detrimental effect of inequality, particularly for the poorest of the poor.

5.2. Alternative measure of financial inclusion

As an additional robustness check, we use an alternative measure of financial inclusion, focusing this time on the so-called demand-side measures. That is the percentage of account ownership at a financial institution.

Growth-redistribution decomposition of those living under \$1.90 dollars a day in low- and low-middle income countries: the role of financial inclusion. Second-stage IV random effects specification.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)
Dependent variable:	Δ Headco	ount		ΔPoverty	/ gap		ΔPoverty	/ gap square	ed	Δ Watts		
ΔGini	1.240***	1.352***	0.858**	0.757***	0.832***	0.555***	0.517***	0.576***	0.386***	1.414***	1.584***	1.049***
	(0.429)	(0.394)	(0.377)	(0.223)	(0.211)	(0.202)	(0.153)	(0.144)	(0.138)	(0.423)	(0.401)	(0.380)
GDP growth rate	-0.029	-0.046	-0.033	-0.031	-0.043	-0.032	-0.026	-0.034	-0.026	-0.066	-0.087	-0.068
	(0.036)	(0.038)	(0.036)	(0.025)	(0.027)	(0.026)	(0.020)	(0.021)	(0.021)	(0.053)	(0.056)	(0.054)
Δ Financial inclusion	-0.037			0.163			0.165			0.423		
	(0.275)			(0.197)			(0.163)			(0.432)		
Δ Gini $\times \Delta$ Financial inclusion	-80.883			-44.397			-27.062			-70.913		
	(52.948)			(27.490)			(19.413)			(52.801)		
ΔFinancial inclusion outreach		0.285			0.429			0.360			0.936	
		(0.433)			(0.292)			(0.235)			(0.624)	
Δ Gini $ imes \Delta$ Financial inclusion outreach		-105.063**			-60.947**			-40.397**			-109.589**	
		(40.849)			(24.109)			(16.336)			(42.700)	
∆Financial inclusion usage		. ,	0.002		. ,	0.124		. ,	0.126		. ,	0.338
			(0.230)			(0.169)			(0.139)			(0.375)
Δ Gini $ imes \Delta$ Financial inclusion usage			14.254			5.731			5.697			20.213
5			(48.192)			(28.639)			(20.780)			(57.328)
Constant	0.002	0.002	0.003	0.001	0.002	0.001	0.001	0.001	0.001	0.002	0.003	0.002
	(0.002)	(0.002)	(0.003)	(0.001)	(0.001)	(0.002)	(0.001)	(0.001)	(0.002)	(0.003)	(0.003)	(0.004)
Year fixed effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Observations	701	701	701	701	701	701	701	701	701	701	701	701
Number of countries	66	66	66	66	66	66	66	66	66	66	66	66
Test of overidentifying restrictions:												
Sargan-Hansen statistic Chi-sq	5.848	4.543	5.059	6.663	4.129	6.972	5.993	3.784	6.468	6.218	3.903	6.262
Sargan-Hansen statistic P-value	0.321	0.474	0.409	0.247	0.531	0.223	0.307	0.581	0.263	0.286	0.563	0.282
Endogenity test:												
Chi-sa	14.690	10.980	14.510	17.850	14.040	15.380	17.830	15.240	15.480	17.750	15.160	15.480
P-value	0.001	0.004	0.001	0.000	0.001	0.001	0.000	0.001	0.000	0.000	0.001	0.000

Note: Δ indicates change from one period to the next. Robust standard errors, clustered at the country level in parentheses. Significance level *** p < 0.01, ** p < 0.05, * p < 0.1. Respective first-stage IV regression in Table A4. Source: World Bank's PovcalNet database and IMF. Coverage: 2004–2018 period.

We re-run our instrumental panel random effects specifications, shown in Eqs. (5)–(7) replacing our measures of financial inclusion with the information on account ownership. As the Global Findex database starts reporting this information from the year 2011, we restrict our analysis for the period 2011–2018.¹⁵ As before, we re-run our results using first the \$1.90 a day, and then separately for the \$3.20 and \$5.50 a day poverty lines.

Table A13 shows the first-stage of the instrumental variable panel random-effects regression for our instrumented endogenous variables.¹⁶ Table 6 shows the second-stage IV panel random effects using as dependent variable the poverty statistics for the \$1.90 a day poverty line.¹⁷ We use the indicator of account ownership in columns 1–4, and those accounts owned by males and females in columns 5–8 and 9–12, respectively. The results are consistent with our earlier findings. We find that the detrimental effect of changes

in inequality on poverty is dampened in countries that have higher account ownership at a financial institution. We also find that higher account ownership by men or women reduces poverty in a country where there is a higher level of income inequality, across all four statistics of poverty used. However, the magnitude of the coefficients for females is larger. To assess the magnitude of these effects, Table A14 re-estimates our results but using standardised coefficients. All the interactions between the change in account ownership and the Gini index are above 0.80, therefore of large magnitude.

As before, we re-run our instrumental variable panel regressions using instead the poverty statistics for those living under \$3.20 and \$5.50 dollars a day. These additional results help us gauge whether improvements in account ownership would lead to similar reductions in less extreme forms of poverty and whether the impacts are also larger for females than males.

Tables A16 and A18 show the second-stage IV panel random estimates for the poverty statistics using the \$3.20 and \$5.50 a day poverty lines. The respective first-stage IV regressions, using the same instruments we just used, are reported in Tables A15 and A17.

We find that our results remain fairly consistent for the \$3.20 dollars a day line. Again, the interaction between changes in inequality and account ownership is negative and statistically significant. Similarly, the interaction coefficient is larger for the bank account ownership of females than males, with the exception of the poverty headcount ratio measure (Table A16).

For the \$5.50 dollar a day poverty line, we find again that the interaction between changes in inequality and account ownership is negative and statistically significant, except for the poverty headcount ratio statistic. This is perhaps not surprising since we had shown in the earlier poverty decomposition analysis

¹⁵ For any gap year on account ownership, we use values from earlier waves of the survey. For instance, for the year 2012 and 2013, we use data from the 2011 survey.

¹⁶ As before, we use as instruments the two-year lagged number of mobile phone subscriptions and the countries' origin of legal code, both correlated to account ownership over time. Since for this first-stage regression we find that our instrument on colonial origin is not statistically significant for any of the regressors, we remove this instrument to avoid using very weak instruments, and use instead the number of airports per squared kilometre, at the beginning of our series, taken from the CIA World Factbook. Airports are key components of modern infrastructure that contributes to reducing costs and fostering interlinkages among firms, suppliers and regions (Pradhan, Arvin, & Norman, 2015). Thus, a higher number of airports is associated with imporvements in financial inclusion for similar reasons as changes in mobile phone subscriptions as it allows a higher level of inclusion across citizens and reduces costs of financial outreach.

¹⁷ The bottom of Table 6 shows evidence of endogeneity suggesting that IV-regressions should be preferred, to non-IV specifications. The Sargan-Hansen overidentifying tests also suggest that our instruments are valid.

12

Growth-redistribution decomposition of those livin	g under \$1.90 dollars a dav:	the role of financial inclusion using ar	alternative measure of financial inclusion	. Second-stage IV random effects specifi	ication.
	,				

Dependent variable:	(1) Δ Headcount	(2) ≤ ∆Poverty gap	(3) Δ Poverty gap squared	(4) ΔWatts	(5) ∆Headcount	(6) ≤ ∆Poverty gap	(7) Δ Poverty gap squared	(8) ∆Watts	(9) ∆Headcount	(10) ∆Poverty gap	(11) Δ Poverty gap squared	(12) ∆Watts
ΔGini	1.766***	1.220***	0.881***	2.406***	1.539***	1.029***	0.736***	2.015***	1.830***	1.300***	0.949***	2.584***
	(0.567)	(0.384)	(0.278)	(0.758)	(0.527)	(0.341)	(0.243)	(0.673)	(0.568)	(0.402)	(0.298)	(0.806)
GDP growth rate	0.139	0.087	0.057	0.157	0.121	0.070	0.045	0.123	0.133	0.086	0.058	0.158
	(0.149)	(0.100)	(0.071)	(0.194)	(0.130)	(0.082)	(0.057)	(0.156)	(0.148)	(0.103)	(0.075)	(0.203)
Δ Account ownership	-0.338	-0.164	-0.092	-0.252								
	(0.420)	(0.295)	(0.217)	(0.585)								
$\Delta Gini \times \Delta Account$ ownership	-34.057*	-24.040*	-17.245*	-46.603*								
	(18.391)	(13.729)	(10.204)	(27.376)	0.000	0.450	0.000	0.0.44				
Account ownership male					-0.326	-0.156	-0.088	-0.241				
A Cini A Associate supportable male					(0.344)	(0.233)	(0.171)	(0.462)				
∆GIII × ∆Account ownership male					-27.178	-17.724	-12.352	-33.491				
A Account ownership formale					(15.654)	(10.510)	(7.414)	(20.048)	0.215	0.140	0.090	0.220
									(0.467)	(0.339)	(0.251)	(0.678)
AGini × AAccount ownership female									-32 422**	-24 026**	-17 559**	-47 282**
									(13.677)	(10.949)	(8,495)	(22.626)
Constant	-0.022*	-0.012*	-0.007	-0.020	-0.021*	-0.011*	-0.006	-0.018	-0.022*	-0.012	-0.007	-0.020
	(0.012)	(0.007)	(0.005)	(0.013)	(0.011)	(0.006)	(0.004)	(0.011)	(0.012)	(0.007)	(0.005)	(0.014)
Year fixed effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Observations	400	400	400	400	400	400	400	400	400	400	400	400
Number of countries	53	53	53	53	53	53	53	53	53	53	53	53
Test of overidentifying restrictions:												
Sargan-Hansen statistic Chi-so	0.450	1 505	1 872	1 735	1 906	3 594	3 862	3 680	0.413	0.472	0 499	0 4 3 4
Sargan-Hansen statistic P-value	0.798	0.471	0.392	0.420	0.386	0.166	0.145	0.159	0.814	0.790	0.779	0.805
Endogenity test:												
Chi-sq	11.500	9.240	8.060	8.090	11.540	9.920	8.630	8.620	10.700	8.640	7.590	7.630
P-value	0.003	0.010	0.018	0.018	0.003	0.007	0.013	0.013	0.005	0.013	0.023	0.022

Note: Δ indicates change from one period to the next. Robust standard errors, clustered at the country level in parentheses. Significance level *** p < 0.01, ** p < 0.05, * p < 0.1. Respective first-stage IV regression in Table A13. Source: World Bank's PovcalNet database and Global Findex. Coverage: 2011–2018 period.

Usage of emergency funds and constraints for financial access by quintiles.

	Poorest	Wealthiest			
Quintiles:	I	II	III	IV	V
Panel A: Main source of emergency funds Savings	Percent 26.05	28.70	30.99	32.95	35.96
Family or friends	39.56	36.27	33.30	29.81	25.07
Money from working	19.43	21.78	24.06	26.11	29.78
Borrowing from a bank	6.74	6.36	5.72	5.60	4.45
Selling assets	4.57	4.01	3.42	3.01	2.17
Some other source	2.09	1.83	1.71	1.69	1.93
Panel B: If received government transfers into an account					
Yes	56.41	58.55	61.04	63.80	65.34
No	42.56	40.63	38.15	35.61	33.97
Panel C: If received government transfers through a mobile phone					
Yes	4.07	3.41	4.44	4.51	5.34
No	95.07	95.72	94.78	94.63	93.82
Panel D: Main reason for not having an account is lack of money					
Yes	67.08	66.55	64.62	61.63	57.56
No	30.03	31.20	33.12	36.13	39.82
Panel F: Main reason for not having an account is because it is too expensive					
Yes	32.05	31 30	29 54	27.28	26 78
No	59.99	61.65	63.85	65.79	66.83
Panel F: Main reason for not having an account is because institution is too far away					
Yes	24 56	21.89	21 21	18 69	1737
No	70.93	74 22	75.40	77.85	78 73
	, 0.00	,	10110	11100	70175
Panel G: Main reason for not having an account is because lacks documentation	22.00	22.22	22.12	22.20	22.00
res	22.89	23.33	23.13	22.20	22.80
NO	72.50	/3.52	/3.63	/4.56	/3.82
Panel H: Main reason for not having an account is because of religious reasons					
Yes	7.50	6.85	6.90	6.43	7.10
No	88.20	89.55	89.42	90.20	89.38

Source: Own estimates using World Bank's 2017 Global Findex.

Table 8

Main source of emergency funds by males and females.

	Male	Female
Main source of emergency funds	Percent	
Savings	30.87	32.66
Family or friends	27.27	36.61
Money from working	30.26	19.15
Borrowing from a bank	5.47	5.75
Selling assets	3.52	2.94
Some other source	1.77	1.90

Source: Own estimates using World Bank's 2017 Global Findex.

(Table A8) that the association between changes in inequality and changes in poverty is not significant for the poverty headcount ratio if measured using the \$5.50 dollars a day. For this poverty line, we also find evidence that interaction coefficient between the Gini index and the account ownership tends to be larger for females and males, particularly for the changes in poverty gap squared and the Watts index.

5.3. What are the main constraints for financial inclusion?

Here, we try to unravel further what are the most important barriers to financial inclusion, using this time the micro-data from the latest 2017 Global Findex survey. According to this dataset, we find that poorer households have major differences in the main sources of emergency funds. The poorest households (in the bottom quintiles) rely more on family and friends than wealthier households (in top quintiles) who rely more on savings (Table 7). These differences are also quite relevant between males and females (Table 8). Women rely more on friends and family than men for emergency funds (37% versus 27%). Moreover, money from working is the main source of emergency for only 19% of women compared to 30% of men.

In Table 7 (Panels B and C) we learn that the majority of the population (about 60%) has received government transfers into one of their accounts, but surprisingly <5% of households have received a government transfer through a mobile phone. This information is potentially quite relevant for how government should provide welfare assistance for COVID-19 response. To reach the poorest households, more reliance is needed on digital and remote technologies such as mobile banking.

Table 7 also shows the main barriers preventing people from having a financial account. By far, the biggest constraint is lack of money (60% of households) across the board regardless of household income (Panel D). Also, roughly 30% of households state that the main reason for them not to have an account is financial institution fees (Panel E). About 21% say that the main reason for not having an account is that the financial institution is too far away (Panel F). A similar percentage also claim that the main constraint in opening an account is not having the documentation required (Panel G). These constraints prevent millions of households from benefiting from financial inclusion and limit countries' ability to make a significant dent in poverty.

6. Forecasts

Another way to inform policy is to forecast the potential shortterm effects of the fall in global economic growth due to the COVID-19 crisis on poverty. In this section, we forecast changes in the poverty headcount ratio using the \$1.90, \$3.20 and \$5.50 dollars a day poverty lines.

At the time of this writing (August 2020) the IMF had released two forecasts for the World Economic Outlook (WEO), the first one released in April and an updated one in June 2020, which is the one we used in this analysis (IMF, 2020b).¹⁸ According to this latest IMF projection, global growth is projected at -4.9% in 2020, and at 5.4% in 2021. Despite this expected recovery in 2021, the sudden drop of household income and GDP is likely to adversely impact low-income households particularly, potentially wiping out years of progress made in reducing poverty. In June 2020, the World Bank also released predictions for global growth in 2020, predicting a 5.2% contraction, the deepest global recession in eight decades, despite unprecedented social welfare and fiscal support (World Bank, 2020). Both the IMF and the World Bank estimate that the global recession will be deeper than anticipated at the beginning of the pandemic. But the question remains, how severe would these economic impacts be on poverty? More importantly, what tools could policymakers use to smooth the damaging impact of the fall in GDP and the likely rise in inequality? To shed light on these two questions, we forecast the likely impacts on extreme levels of poverty as well as to what extent improvements in financial inclusion could mitigate some of these impacts.

6.1. Impact of COVID-19 on people living under \$1.90 dollars a day

We start by forecasting the expected global change in poverty using the \$1.90 dollars a day poverty line. We use the poverty headcount ratio drawn from the PovcalNet dataset during the period 2004–2018. We forecast changes in poverty for *all* 121 countries in our sample for which we have both data on financial inclusion and the poverty headcount ratio. We focus on the extent to which improvements in the sub-index of the financial outreach would curb poverty since that was the dimension that both our panel fixed effects and IV specifications suggest is the most important in reducing poverty. We use Eq. (4), the panel fixed effects specifications, as the basis for our forecasts as they offer more conservative estimates than the IV-specifications.¹⁹ These IV specifications control for a remaining endogeneity bias, however they are also subject to upward bias depending on the strength of the instruments used.

In this global forecast for poverty we focus on two scenarios only. The first one assumes the change in GDP growth for each country as estimated by the IMF for the years 2020 and 2021 (IMF, 2020a). As a robustness test, the second scenario assumes the World Bank global economic growth projections. In both these scenarios, we are assuming that inequality and financial inclusion outreach will remain at their pre-COVID levels. Fig. 5 shows that both scenarios lead to almost identical forecasts. That is, globally, the percentage of people living under \$1.90 a day would increase from 13.1% in 2019, to about 13.8% in 2020 and 14.5% by 2021. This represents an increase of 107.8 million people in poverty, using the \$1.90 dollars a day poverty line.

Since extreme levels of poverty are concentrated in poorer countries, we next explore to what extent improvements in financial inclusion could reduce the impacts on the poverty headcount ratio for the 79 low- and low-middle income countries included in our sample. Fig. 6 presents four forecast scenarios using the IMF growth projections, whereas Fig. 7 uses instead the World Bank growth projections. Our first forecast shows the baseline sce-



Fig. 5. Forecast poverty headcount \$1.90 a day, all countries.



Fig. 6. Forecast poverty headcount \$1.90 a day, including low- and lower-middleincome countries only using IMF forecast GDP growth.



Fig. 7. Forecast poverty headcount \$1.90 a day, including low- and lower-middleincome countries only using World Bank forecast GDP growth.

nario, assuming only the forecasted change in GDP growth for each country for the years 2020 and 2021. The second scenario, on top of these changes in GDP growth also assumes that inequality (proxied

¹⁸ The first WEO forecast in April 2020 assumed that the COVID-19 pandemic would fade away in the second half of 2020, and the containment efforts would gradually ease, having an expected reduction of -3.0% in the global economic growth (IMF, 2020a). That earlier scenario proved too optimistic since the COVID-19 pandemic had a more negative impact on activity in the first semester of 2020 than anticipated.

¹⁹ Our forecasts offer the same conclusions if we use instead the overall financial inclusion index, this is not surprising given its strong correlation to the financial outreach sub-index.

by the Gini index) will increase by 1% in each country.²⁰ The third scenario assumes that on top of the fall in GDP growth, there is an improvement of 5% in the financial outreach index, assuming no changes in the Gini indices. The fourth scenario is the same as the third one, but assumes instead an improvement of 10% in the financial outreach index. We acknowledge that these scenarios are rather arbitrary but are, to some extent guided by the recent changes in the financial inclusion and financial outreach indices.²¹

The baseline scenario in both Figs. 6 and 7 suggests that the percentage of people living under \$1.90 a day would increase from 25.2% in 2019 to 26.2% by 2020, and to 27.1% by 2021 in lowand lower-middle-income countries. If on top of the change in GDP there is a rise in inequality, then poverty would increase even further to near 28% by 2021. Poverty rises could be curbed with substantial improvements in financial inclusion, as suggested by our third and fourth scenarios in both Figs. 6 and 7.

6.2. Impact of COVID-19 on people living under \$3.20 and \$5.50 dollar a day

We continue by forecasting the expected global change in poverty using the \$3.20 and \$5.50 dollars a day poverty lines for *all* 121 countries in our sample using Eq. (4) specification as before. Fig. A1 suggests that the percentage of people worldwide living under \$3.20 a day would increase from 24.8% in 2019 to nearly 27% by 2021, pushing nearly 169.4 million people in poverty. Fig. A2 shows that in terms of the percentage of people living under \$5.50 a day, poverty would increase from 40% in 2019 to nearly 43% by 2021, pushing nearly 231 million people in poverty.

If focused only on the 79 low- and low-middle income countries included in our sample, Fig. A3 suggests that the percentage of people living under \$3.20 a day would increase from 45.5% in 2019 to nearly 48.0% by 2021.²² However, that rise of poverty could be smoothed to 46.4% with an increase of 10% in the financial inclusion index. Similarly, Fig. A4 suggests that the percentage of people living under \$5.50 a day would increase from 68.3% in 2019 to 70.0% by 2021. However, that rise of poverty could be smoothed to 68.7% with an increase of 10% in the financial inclusion outreach sub-index alone.

6.3. Limitations and policy implications of our analysis

Overall, our poverty forecasts are in the same range of other studies also using PovcalNet (Sumner et al., 2020). Nonetheless, we acknowledge the limitations of these types of forecasts. They rely on economic growth projections which understandably involve subjective judgements made by forecasters within the IMF and the World Bank and a considerable margin of error (Sandefur & Subramanian, 2020). This is particularly so given that significant slowdowns have historically proven to be difficult to predict (Ravallion, 2013). Plus, the world has never seen such a sudden and widespread drop in economic activity. Although over 181 countries have implemented urgent welfare assistance that is likely to continue in months to come and perhaps mitigate further some welfare impacts (Gentilini et al., 2020), the perception that the pandemic has been mismanaged in some countries, together with economic downturn are an explosive combination for political crises. For instance, ten of the elections held soon after the lockdowns have suffered from electoral violence, which could prolong the economic downturn even further in some countries (Birch et al., 2020). None of these important considerations is directly modelled in our analysis nor on how soon a COVID-19 vaccine would be available. Still, from our analysis we can shed light into five key implications for policymaking.

First, economic growth recovery alone is unlikely to reverse the expected rise in extreme poverty, as growth gains do not automatically trickle down to the bottom quintiles. Second, poor households are likely to suffer the double burden of rising levels of inequality which requires urgent tailored policies to address it. Although most of low-income and lower-income countries have implemented urgent COVID-19 welfare assistance, in some countries, these funds might be used for political clientelism, risking benefiting only a fraction of those who need it (Birch et al., 2020). Moreover, most of these packages have been translated in urgent temporary cash transfers, but not in any short- or medium-term plans for job recovery urgently needed for reducing both poverty and inequality (Gutiérrez-Romero, 2020).²³

Third, our analysis suggests that the poorest households would benefit from having much wider financial inclusion outreach. However, we are not under the illusion that simply building more bank branches and ATMs near poorer households could be a weapon to reduce poverty per se.²⁴ A very high share of the poorest population still does not have access to financial services because of lack of money, financial services are not available either physically or remotely and fees to open or use accounts are too high. All these constraints need to be addressed to enable the poor to benefit from financial inclusion.

Fourth, for financial inclusion to continue to expand, despite the ongoing pandemic, greater collaboration between governments and financial institutions would be needed to find cheaper forms of financial inclusion such as remote or mobile banking, and reduce the red tape to open and use financial accounts. This is a worthy task as earlier studies have suggested that improving financial inclusion towards the poor is likely to create fewer distortions to the economy than potential taxes (Beck, Demirgüç-Kunt, & Levine, 2007; Beck, Demirgüç-Kunt, & Martinez Peria, 2007).

Fifth, drawing from the extensive randomised-control literature, it is clear that the type of financial services offered is important. If the goal is poverty reduction of those living under extreme levels of poverty, offering more inclusive micro-saving and microinsurance services is far more promising, than simply offering the extreme poor micro-credit for opening up small businesses (Cai et al., 2009; Dupas & Robinson, 2013).²⁵

7. Conclusion

Our findings showed that over the last two decades, financial inclusion, in particular financial outreach, has been instrumental

²⁰ Although in some recessions inequality might even be reduced, for COVID-19 it is highly likely that the households with lowest income will be more severely affected than the most wealthy given differences in vulnerabilities to price shocks and income shocks due to unemployment or occupational health-risks of COVID.

²¹ For instance, the financial outreach sub-index increased by about 4% between 2016 and 2018, and by about 8% between 2014 and 2018 in the analysed low-and lower-middle income countries. A 10% increase in the overall financial outreach sub-index could be accomplished by an increase of roughly 2% in either of the Fo1, Fo2 or Fo3 components of the financial outreach sub-index or by a 10% increase in the Fo4 component.

²² The IMF and World Bank growth projections suggest remarkably similar poverty trends, thus we present the IMF growth projections only.

²³ We acknowledge that our forecast analysis has not directly modelled the impact of unemployment on poverty, but we have done so indirectly, by using the forecast change in global economic growth that does take into account those predicted unemployment trends.

²⁴ Similarly, although some important case studies, such as in Kenya, have shown that mobile banking can significantly contribute to poverty reduction (Suri & Jack, 2016), again increasing these services alone is not sufficient to reduce poverty.

²⁵ These micro-credits are usually taken up by people with already viable businesses, but *meta*-analyses have shown that micro-credits have a negligible effect on poverty reduction (Banerjee, Duflo, et al., 2015; Duvendack et al., 2014; Stewart et al., 2012; Yang & Stanley, 2014). Credits to firms are of course greatly needed to enable jobs and economic recovery, but different tools would be needed when it comes to reducing extreme forms of poverty.

R. Gutiérrez-Romero and M. Ahamed

in reducing poverty. Unlike other recent cross-country studies, we have shown that financial inclusion has an important role in poverty reduction, even in low- and lower-middle-income countries. These findings are robust to using different measures of financial inclusion, measures of poverty, and instrumental variable analysis to address potential endogeneity concerns (e.g. Goksu et al., 2017; Park & Mercado, 2018).

Another important aspect of our results is that financial outreach reduces poverty by offsetting the detrimental effect that inequality has on poverty. In other words, financial outreach helps reduce existing inequalities in financial services that are likely to enable poor people to smooth their consumption from significant shocks like the ongoing pandemic. Our findings are in line with the experimental literature which has found that financial inclusion helps to break inequality barriers among the most vulnerable groups (e.g. Banerjee, Karlan, & Zinman, 2015; Dupas & Robinson, 2013; Koomson et al., 2020; Li, 2018).

Furthermore, our forecast analysis shows that nearly 107.8 million people could be pushed into extreme poverty, wiping out years of progress in poverty reduction if adequate measures are not taken. Our analysis suggests that financial inclusion, in particular, financial outreach could help curb some of the increases in poverty. Financial inclusion has been regarded as a key complementary tool to address the Sustainable Development Goals (Chibba, 2009). The developing world has never before needed more urgent acceleration in using financial inclusion as a complementary tool to reduce poverty. COVID-19 crises will require flexibility from all sectors involved, particularly from the financial industry. As millions of poor people and small firms have lost their livelihoods suddenly, they will need rapid access to government

Table A1

cfindox

Financial inclusion index

ennaen																
Country	Income Level	2004	2005	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016	2017	2018
Afghanistan	Low income	0.012	0.013	0.014	0.015	0.016	0.019	0.019	0.018	0.024	0.021	0.026	0.027	0.027	0.026	0.028
Albania	Upper-middle income	0.258	0.273	0.288	0.314	0.347	0.355	0.357	0.359	0.360	0.358	0.352	0.352	0.350	0.311	0.296
Algeria	Upper-middle income	0.085	0.086	0.091	0.098	0.099	0.071	0.079	0.082	0.086	0.088	0.090	0.090	0.089	0.090	0.090
Angola	Lower-middle income	0.006	0.008	0.013	0.016	0.020	0.024	0.032	0.037	0.042	0.047	0.049	0.050	0.050	0.051	0.049
Argentina	Upper-middle income	0.130	0.143	0.151	0.166	0.180	0.191	0.205	0.223	0.251	0.264	0.277	0.304	0.331	0.355	0.378
Armenia	Lower-middle income	0.079	0.104	0.115	0.144	0.163	0.183	0.204	0.243	0.280	0.312	0.357	0.374	0.391	0.397	0.426
Austria	High income	0.375	0.363	0.364	0.366	0.368	0.364	0.360	0.373	0.374	0.378	0.417	0.410	0.393	0.381	0.389
Azerbaijan	Upper-middle income	0.083	0.084	0.087	0.095	0.103	0.108	0.126	0.145	0.173	0.224	0.304	0.336	0.313	0.324	0.350
Bahamas, The	High income	0.495	0.511	0.523	0.542	0.544	0.529	0.497	0.502	0.483	0.500	0.494	0.471	0.490	0.468	0.471
Bangladesh	Lower-middle income	0.135	0.138	0.142	0.145	0.150	0.158	0.176	0.196	0.206	0.221	0.233	0.251	0.266	0.277	0.292
Belgium	High income	0.933	0.933	0.933	0.943	0.945	0.941	0.947	0.948	0.946	0.943	0.935	0.930	0.921	0.909	0.888
Belize	Upper-middle income	0.233	0.234	0.230	0.240	0.244	0.243	0.242	0.241	0.233	0.234	0.225	0.207	0.209	0.197	0.205
Bhutan	Lower-middle income	0.090	0.089	0.091	0.096	0.103	0.112	0.146	0.190	0.103	0.202	0.248	0.270	0.299	0.280	0.274
Bolivia	Lower-middle income	0.038	0.041	0.043	0.072	0.079	0.092	0.106	0.126	0.139	0.159	0.178	0.197	0.217	0.238	0.244
Bosnia and Herzegovina	Upper-middle income	0.224	0.235	0.255	0.287	0.313	0.331	0.316	0.334	0.343	0.363	0.370	0.367	0.361	0.364	0.371
Botswana	Upper-middle income	0.165	0.166	0.170	0.186	0.185	0.184	0.186	0.183	0.187	0.190	0.192	0.183	0.182	0.184	0.219
Brazil	Upper-middle income	0.279	0.285	0.288	0.297	0.317	0.338	0.361	0.385	0.530	0.567	0.571	0.562	0.623	0.601	0.597
Brunei Darussalam	High income	0.420	0.447	0.443	0.458	0.476	0.501	0.489	0.485	0.568	0.538	0.486	0.487	0.443	0.428	0.429
Bulgaria	Upper-middle income	0.437	0.448	0.496	0.544	0.591	0.604	0.598	0.579	0.580	0.574	0.563	0.553	0.538	0.540	0.542
Burundi	Low income	0.008	0.009	0.010	0.011	0.011	0.013	0.016	0.018	0.022	0.025	0.025	0.026	0.026	0.026	0.026
Cambodia	Lower-middle income	0.018	0.018	0.019	0.022	0.025	0.032	0.035	0.039	0.044	0.050	0.062	0.071	0.081	0.089	0.101
Cameroon	Lower-middle income	0.000	0.002	0.005	0.009	0.011	0.013	0.016	0.018	0.020	0.022	0.020	0.024	0.025	0.025	0.033
Central African Republic	Low income	0.000	0.000	0.000	0.002	0.003	0.005	0.007	0.008	0.009	0.012	0.009	0.010	0.011	0.011	0.011
Chad	Low income	0.004	0.004	0.004	0.004	0.004	0.005	0.005	0.005	0.005	0.006	0.007	0.007	0.008	0.008	0.008
Chile	High income	0.296	0.314	0.353	0.385	0.412	0.427	0.451	0.479	0.499	0.509	0.506	0.508	0.510	0.515	0.524
China	Upper-middle income	0.049	0.049	0.049	0.053	0.059	0.066	0.075	0.084	0.097	0.113	0.128	0.167	0.177	0.182	0.205
Colombia	Upper-middle income	0.205	0.220	0.235	0.240	0.245	0.261	0.257	0.272	0.288	0.304	0.320	0.335	0.340	0.350	0.361
Comoros	Low income	0.011	0.011	0.011	0.015	0.014	0.017	0.018	0.038	0.037	0.041	0.043	0.043	0.047	0.047	0.048
Congo, Dem. Rep.	Low income	0.001	0.001	0.001	0.001	0.001	0.001	0.003	0.003	0.006	0.008	0.008	0.009	0.009	0.009	0.009
Congo, Rep.	Lower-middle income	0.003	0.003	0.003	0.008	0.010	0.012	0.014	0.017	0.020	0.023	0.031	0.035	0.037	0.040	0.040
Costa Rica	Upper-middle income	0.226	0.280	0.294	0.326	0.334	0.335	0.336	0.366	0.374	0.411	0.431	0.422	0.473	0.502	0.517
Croatia	Upper-middle income	0.333	0.340	0.352	0.372	0.391	0.406	0.412	0.412	0.401	0.399	0.382	0.385	0.379	0.362	0.357
Cyprus	High income	0.555	0.554	0.565	0.566	0.573	0.580	0.582	0.577	0.564	0.501	0.496	0.487	0.487	0.669	0.603
Czech Republic	High income	0.288	0.299	0.308	0.326	0.344	0.354	0.364	0.383	0.399	0.424	0.444	0.361	0.361	0.366	0.375

assistance and financial services such as savings and credit instruments. Financial institutions will have a crucial role to play to keep the economy afloat, contain potential regional contagion of financial collapse and help resuscitate small businesses once social distancing measures are eased.

CRediT authorship contribution statement

Roxana Gutiérrez-Romero: Conceptualization, Methodology, Software, Validation, Formal analysis, Investigation, Resources, Data curation, Writing – original draft, Writing – review & editing, Visualization, Project administration. **Mostak Ahamed:** Investigation, Data curation, Writing – original draft.

Declaration of Competing Interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Acknowledgements

We are grateful for the constructive and insightful comments provided by the Editor, and the three anonymous referees.

Appendix A

R. Gutiérrez-Romero and M. Ahamed

Table A1 (continued)

cfindex

 Country	Income Level	2004	2005	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016	2017	2018
 Djibouti	Lower-middle income	0.013	0.013	0.014	0.014	0.016	0.021	0.026	0.029	0.032	0.041	0.048	0.055	0.051	0.067	0.067
Dominican Republic	Upper-middle income	0.177	0.186	0.187	0.194	0.197	0.201	0.205	0.213	0.216	0.236	0.244	0.269	0.280	0.280	0.281
Ecuador	Upper-middle income	0.108	0.120	0.112	0.146	0.160	0.175	0.185	0.178	0.175	0.169	0.183	0.187	0.187	0.187	0.192
Egypt, Arab Rep.	Lower-middle income	0.065	0.066	0.068	0.071	0.074	0.075	0.076	0.076	0.079	0.090	0.093	0.097	0.104	0.128	0.138
El Salvador	Lower-middle income	0.218	0.222	0.232	0.222	0.232	0.251	0.246	0.254	0.256	0.269	0.270	0.276	0.289	0.316	0.321
Equatorial Guinea	Upper-middle income	0.016	0.018	0.020	0.023	0.026	0.027	0.029	0.034	0.039	0.037	0.041	0.047	0.050	0.063	0.072
Estonia	High income	0.482	0.496	0.559	0.613	0.579	0.568	0.587	0.570	0.480	0.472	0.466	0.444	0.437	0.434	0.447
Eswatini	Lower-middle income	0.103	0.109	0.113	0.114	0.119	0.120	0.124	0.128	0.146	0.152	0.169	0.174	0.162	0.159	0.164
Fiji	Upper-middle income	0.131	0.152	0.186	0.170	0.185	0.185	0.199	0.210	0.208	0.227	0.255	0.276	0.302	0.318	0.314
Finland	High income	0.331	0.335	0.342	0.339	0.339	0.335	0.336	0.335	0.331	0.321	0.318	0.312	0.292	0.160	0.251
Gabon	Upper-middle income	0.027	0.027	0.028	0.030	0.041	0.043	0.048	0.054	0.074	0.107	0.107	0.107	0.107	0.107	0.107
Gambia, The	Low income	0.031	0.034	0.039	0.045	0.066	0.066	0.062	0.081	0.083	0.085	0.086	0.091	0.094	0.084	0.091
Georgia	Lower-middle income	0.072	0.101	0.141	0.190	0.254	0.255	0.279	0.323	0.381	0.424	0.475	0.479	0.518	0.570	0.577
Ghana	Lower-middle income	0.034	0.035	0.039	0.044	0.051	0.054	0.065	0.071	0.078	0.084	0.086	0.112	0.110	0.129	0.137
Greece	High income	0.657	0.661	0.678	0.694	0.709	0.707	0.702	0.690	0.680	0.645	0.625	0.619	0.612	0.608	0.602
Guatemala	Lower-Initiatie Income	0.203	0.216	0.217	0.269	0.297	0.313	0.320	0.347	0.371	0.381	0.404	0.408	0.400	0.385	0.341
Guillea	Low income	0.004	0.004	0.005	0.000	0.008	0.009	0.011	0.015	0.015	0.010	0.020	0.022	0.021	0.020	0.020
Guyana	Low income	0.120	0.125	0.105	0.124	0.124	0.145	0.140	0.154	0.156	0.101	0.100	0.158	0.150	0.155	0.152
Honduras	Low income	0.042	0.040	0.045	0.052	0.055	0.058	0.034	0.055	0.055	0.033	0.031	0.031	0.052	0.034	0.030
Hungary	High income	0.144	0.140	0.155	0.105	0.151	0.155	0.105	0.150	0.210	0.220	0.231	0.230	0.241	0.240	0.241
Iceland	High income	0.682	0.501	0.510	0.550	0.551	0.555	0.555	0.550	0.551	0.550	0.540	0.552	0.555	0.550	0.555
India	Lower-middle income	0.136	0.138	0.142	0.150	0.162	0.177	0.194	0.210	0.232	0.258	0.300	0.335	0.368	0.393	0.402
Indonesia	Lower-middle income	0.098	0.104	0.103	0.107	0.116	0.126	0.136	0.176	0.223	0.254	0.272	0.281	0.296	0.352	0.352
Ireland	High income	0.423	0.417	0.418	0.471	0.479	0.474	0.457	0.431	0.413	0.399	0.470	0.476	0.475	0.442	0.421
Italy	High income	0.545	0.550	0.564	0.584	0.624	0.622	0.613	0.619	0.610	0.591	0.589	0.605	0.593	0.579	0.553
Jamaica	Upper-middle income	0.232	0.228	0.227	0.221	0.222	0.221	0.217	0.212	0.219	0.222	0.230	0.235	0.241	0.290	0.301
Japan	High income	0.955	0.954	0.954	0.954	0.954	0.956	0.956	0.956	0.957	0.956	0.956	0.957	0.957	0.957	0.956
Jordan	Lower-middle income	0.184	0.186	0.187	0.187	0.188	0.191	0.173	0.172	0.171	0.165	0.162	0.160	0.167	0.170	0.174
Kenya	Lower-middle income	0.023	0.027	0.037	0.045	0.059	0.071	0.089	0.099	0.105	0.133	0.163	0.193	0.216	0.230	0.250
Kiribati	Lower-middle income	0.059	0.059	0.059	0.059	0.059	0.059	0.059	0.059	0.061	0.071	0.071	0.071	0.071	0.071	0.071
Korea, Rep.	High income	0.861	0.866	0.871	0.877	0.881	0.878	0.879	0.881	0.883	0.881	0.875	0.872	0.869	0.862	0.861
Kosovo	Upper-middle income	0.183	0.188	0.200	0.222	0.251	0.275	0.286	0.312	0.321	0.323	0.316	0.313	0.321	0.283	0.265
Lao PDR	Lower-middle income	0.054	0.054	0.054	0.054	0.054	0.057	0.062	0.064	0.067	0.072	0.085	0.095	0.100	0.109	0.114
Latvia	High income	0.340	0.374	0.409	0.510	0.533	0.526	0.528	0.539	0.511	0.496	0.431	0.409	0.394	0.409	0.401
Lebanon	Upper-middle income	0.393	0.400	0.412	0.429	0.456	0.478	0.496	0.498	0.508	0.496	0.488	0.488	0.488	0.497	0.507
Lesotho	Lower-middle income	0.047	0.048	0.047	0.049	0.050	0.054	0.062	0.067	0.062	0.072	0.073	0.078	0.081	0.083	0.088
Liberia	Low income	0.011	0.011	0.011	0.012	0.013	0.020	0.027	0.030	0.034	0.044	0.045	0.048	0.052	0.053	0.052
Libya	Upper-middle income	0.039	0.039	0.043	0.043	0.044	0.048	0.049	0.047	0.048	0.049	0.049	0.049	0.049	0.048	0.048
Madagascar	Low income	0.005	0.006	0.007	0.008	0.010	0.011	0.018	0.013	0.014	0.016	0.019	0.021	0.022	0.025	0.029
Malawi	Low income	0.037	0.038	0.038	0.042	0.043	0.045	0.048	0.042	0.047	0.050	0.050	0.058	0.058	0.058	0.058
Maldivos	Upper-middle income	0.375	0.390	0.400	0.414	0.423	0.430	0.443	0.452	0.462	0.483	0.479	0.458	0.452	0.448	0.445
Malta	Uigh incomo	0.328	0.335	0.302	0.429	0.438	0.444	0.444	0.447	0.433	0.407	0.409	0.012	0.323	0.524	0.327
Mauritania	Lower-middle income	0.020	0.033	0.040	0.077	0.032	0.034	0.030	0.304	0.300	0.858	0.032	0.050	0.0057	0.075	0.057
Mauritius	Upper-middle income	0.513	0.547	0.549	0.581	0.052	0.627	0.632	0.653	0.672	0.681	0.670	0.665	0.640	0.628	0.600
Mexico	Upper-middle income	0.125	0.130	0.145	0.156	0.205	0.209	0.229	0.203	0.214	0.215	0.212	0.218	0.229	0.020	0.231
Micronesia, Fed. Sts.	Lower-middle income	0.108	0.108	0.109	0.108	0.109	0.126	0.138	0.140	0.149	0.148	0.148	0.133	0.129	0.125	0.123
Moldova	Lower-middle income	0.236	0.259	0.276	0.308	0.344	0.348	0.361	0.384	0.395	0.413	0.418	0.286	0.306	0.317	0.327
Mongolia	Lower-middle income	0.164	0.178	0.195	0.240	0.248	0.255	0.274	0.301	0.327	0.353	0.377	0.400	0.461	0.431	0.463
Montenegro	Upper-middle income	0.234	0.246	0.267	0.308	0.456	0.451	0.445	0.465	0.480	0.465	0.470	0.456	0.502	0.459	0.489
Morocco	Lower-middle income	0.117	0.125	0.130	0.140	0.151	0.200	0.208	0.219	0.230	0.239	0.249	0.258	0.270	0.286	0.291
Mozambique	Low income	0.013	0.014	0.018	0.021	0.023	0.027	0.030	0.034	0.037	0.045	0.049	0.056	0.062	0.058	0.058
Myanmar	Lower-middle income	0.017	0.017	0.016	0.017	0.017	0.018	0.018	0.018	0.021	0.026	0.032	0.038	0.042	0.046	0.057
Namibia	Upper-middle income	0.085	0.089	0.103	0.108	0.161	0.182	0.193	0.204	0.201	0.218	0.245	0.268	0.306	0.280	0.306
Nepal	Low income	0.076	0.075	0.076	0.080	0.080	0.085	0.087	0.097	0.103	0.107	0.118	0.130	0.144	0.171	0.209
Netherlands	High income	0.697	0.671	0.685	0.698	0.697	0.695	0.663	0.632	0.600	0.584	0.594	0.549	0.525	0.504	0.497
Nicaragua	Lower-middle income	0.065	0.074	0.080	0.088	0.094	0.085	0.076	0.083	0.084	0.091	0.100	0.110	0.128	0.135	0.125
North Macedonia	Upper-middle income	0.274	0.271	0.291	0.314	0.402	0.434	0.447	0.440	0.448	0.462	0.476	0.492	0.495	0.497	0.499
Norway	High income	0.200	0.200	0.215	0.219	0.244	0.215	0.211	0.212	0.207	0.207	0.202	0.198	0.186	0.180	0.177
Oman	High income	0.169	0.172	0.177	0.187	0.197	0.202	0.204	0.211	0.203	0.214	0.202	0.200	0.200	0.196	0.197
Pakistan	Lower-middle income	0.055	0.058	0.065	0.070	0.074	0.074	0.077	0.081	0.086	0.091	0.098	0.104	0.110	0.116	0.119
ralldilld Dapua Now Cuince	opper-inidale income	0.251	0.251	0.251	0.251	0.252	0.254	0.276	0.293	0.316	0.344	0.364	0.3/9	0.387	0.387	0.388
Paraguay	Lower-middle income	0.020	0.020	0.030	0.049	0.030	0.051	0.052	0.050	0.000	0.033	0.037	0.057	0.002	0.007	0.009
r ai dgudy Doru	Upper-middle income	0.030	0.029	0.031	0.031	0.037	0.044	0.057	0.002	0.008	0.074	0.073	0.081	0.083	0.131	0.152
Philippines	Lower-middle income	0.115	0.005	0.095	0.114	0.141	0.144	0.101	0.102	0.190	0.211	0.244	0.330	0.528	0.520	0.342
Poland	High income	0.476	0.105	0.107	0.506	0.530	0.115	0.120	0.130	0.135	0.132	0.556	0.109	0.583	0.104	0.192
Portugal	High income	0.797	0.801	0.409	0.500	0.550	0.345	0.344	0.545	0.787	0.779	0.550	0.508	0.535	0.500	0.007
Romania	Upper-middle income	0.206	0.214	0.228	0.240	0.255	0.257	0.256	0.261	0.253	0.244	0.241	0.238	0.233	0.228	0.224
Rwanda	Low income	0.002	0.005	0.005	0.008	0.048	0.055	0.063	0.072	0.084	0.085	0.081	0.079	0.084	0.083	0.099
Samoa	Upper-middle income	0.134	0.151	0.163	0.176	0.188	0.204	0.207	0.233	0.247	0.306	0.306	0.305	0.356	0.362	0.367
San Marino	High income	0.988	1.000	0.981	1.000	1.000	1.000	0.989	0.979	0.975	1.000	0.994	0.934	0.934	0.932	0.926
	-															

(continued on next page)

R. Gutiérrez-Romero and M. Ahamed

Table A1 (continued)

cfindex																
Country	Income Level	2004	2005	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016	2017	2018
Sao Tome and Principe	Lower-middle income	0.234	0.234	0.234	0.234	0.234	0.234	0.234	0.234	0.269	0.293	0.284	0.319	0.261	0.274	0.257
Saudi Arabia	High income	0.164	0.165	0.174	0.182	0.188	0.193	0.198	0.199	0.201	0.225	0.232	0.242	0.253	0.263	0.267
Serbia	Upper-middle income	0.119	0.154	0.182	0.217	0.246	0.244	0.239	0.234	0.270	0.253	0.242	0.247	0.256	0.244	0.246
Seychelles	High income	0.410	0.433	0.448	0.445	0.490	0.506	0.509	0.523	0.576	0.596	0.633	0.630	0.650	0.681	0.691
Solomon Islands	Lower-middle income	0.056	0.058	0.059	0.060	0.066	0.066	0.061	0.058	0.060	0.052	0.076	0.079	0.081	0.085	0.085
South Africa	Upper-middle income	0.148	0.164	0.182	0.190	0.215	0.241	0.259	0.267	0.297	0.316	0.333	0.330	0.340	0.342	0.313
South Sudan	Low income	0.004	0.004	0.004	0.004	0.004	0.004	0.004	0.004	0.005	0.008	0.012	0.014	0.016	0.014	0.013
Spain	High income	0.717	0.735	0.743	0.762	0.763	0.767	0.762	0.744	0.726	0.717	0.696	0.682	0.662	0.671	0.663
Suriname	Upper-middle income	0.231	0.234	0.238	0.242	0.242	0.248	0.251	0.252	0.256	0.261	0.266	0.262	0.290	0.285	0.298
Sweden	High income	0.524	0.525	0.526	0.529	0.532	0.530	0.529	0.538	0.550	0.547	0.545	0.540	0.529	0.522	0.518
Switzerland	High income	0.755	0.758	0.754	0.755	0.749	0.764	0.764	0.775	0.770	0.790	0.787	0.753	0.702	0.688	0.671
Syrian Arab Republic	Lower-middle income	0.039	0.039	0.039	0.040	0.044	0.055	0.057	0.059	0.058	0.058	0.058	0.058	0.058	0.058	0.058
Tanzania	Low income	0.023	0.023	0.023	0.026	0.029	0.029	0.034	0.038	0.040	0.041	0.064	0.047	0.047	0.047	0.047
Thailand	Upper-middle income	0.203	0.233	0.269	0.294	0.326	0.342	0.363	0.373	0.398	0.419	0.447	0.456	0.459	0.464	0.474
Tonga	Upper-middle income	0.219	0.217	0.199	0.203	0.220	0.205	0.197	0.201	0.210	0.231	0.228	0.299	0.282	0.339	0.354
Trinidad and Tobago	High income	0.362	0.362	0.362	0.362	0.362	0.374	0.386	0.385	0.391	0.399	0.402	0.401	0.403	0.392	0.389
Turkey	Upper-middle income	0.271	0.274	0.285	0.304	0.307	0.314	0.464	0.487	0.511	0.558	0.553	0.566	0.588	0.622	0.624
Uganda	Low income	0.017	0.019	0.020	0.022	0.029	0.034	0.037	0.036	0.040	0.041	0.043	0.048	0.051	0.061	0.075
Ukraine	Lower-middle income	0.404	0.416	0.455	0.510	0.536	0.551	0.553	0.561	0.570	0.475	0.469	0.444	0.438	0.469	0.410
United Arab Emirates	High income	0.231	0.258	0.254	0.259	0.266	0.287	0.289	0.296	0.302	0.296	0.327	0.350	0.359	0.366	0.358
United Kingdom	High income	0.426	0.422	0.421	0.432	0.432	0.421	0.419	0.419	0.414	0.438	0.438	0.438	0.438	0.438	0.415
Uzbekistan	Lower-middle income	0.167	0.180	0.195	0.199	0.206	0.208	0.218	0.240	0.244	0.264	0.267	0.241	0.260	0.321	0.320
Vanuatu	Lower-middle income	0.164	0.164	0.164	0.173	0.182	0.185	0.192	0.193	0.203	0.205	0.207	0.217	0.218	0.216	0.215
Venezuela, RB	Upper-middle income	0.095	0.095	0.110	0.130	0.134	0.140	0.146	0.151	0.156	0.175	0.189	0.225	0.225	0.225	0.225
Vietnam	Lower-middle income	0.047	0.050	0.052	0.061	0.070	0.077	0.082	0.117	0.130	0.143	0.157	0.168	0.182	0.181	0.201
West Bank and Gaza	Lower-middle income	0.1973	0.198	0.202	0.2128	0.2273	0.241	0.249	0.276	0.2878	0.2927	0.3046	0.3173	0.3349	0.3484	0.3982
Yemen, Rep.	Lower-middle income	0.0173	0.0183	0.0189	0.0197	0.0202	0.0207	0.021	0.0201	0.0221	0.0235	0.0257	0.0249	0.0249	0.0249	0.0249
Zambia	Lower-middle income	0.0246	0.0247	0.0257	0.0273	0.0299	0.0324	0.0336	0.0364	0.0436	0.0541	0.0594	0.0616	0.0622	0.0634	0.0574
Zimbabwe	Low income	0.0726	0.0757	0.0749	0.0431	0.0415	0.0321	0.0388	0.0388	0.0652	0.0726	0.0681	0.0433	0.044	0.0793	0.1086

Source: Own estimates using Financial Access Survey, 2004–2018 (IMF, 2019).

coutreach

Financial outreach index.

Country	2004	2005	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016	2017	2018
 Afghanistan	0.000	0.002	0.004	0.005	0.007	0.011	0.012	0.012	0.011	0.012	0.013	0.012	0.012	0.012	0.014
Albania	0.075	0.102	0.128	0.175	0.232	0.247	0.251	0.255	0.257	0.253	0.241	0.243	0.237	0.225	0.214
Algeria	0.024	0.025	0.028	0.031	0.033	0.035	0.035	0.036	0.037	0.037	0.039	0.041	0.042	0.042	0.043
Angola	0.009	0.012	0.021	0.026	0.033	0.040	0.055	0.064	0.073	0.082	0.084	0.087	0.087	0.087	0.083
Argentina	0.069	0.069	0.070	0.074	0.083	0.090	0.101	0.116	0.131	0.148	0.152	0.156	0.162	0.168	0.178
Armenia	0.080	0.095	0.111	0.143	0.176	0.198	0.212	0.247	0.273	0.286	0.307	0.315	0.315	0.322	0.326
Austria	0.397	0.369	0.372	0.372	0.375	0.376	0.375	0.404	0.416	0.418	0.494	0.486	0.462	0.461	0.465
Azerbaijan Rahamas Tho	0.085	0.087	0.091	0.100	0.119	0.130	0.137	0.147	0.149	0.158	0.168	0.170	0.161	0.160	0.162
Bangladesh	0.337	0.301	0.300	0.303	0.303	0.330	0.333	0.373	0.309	0.451	0.442	0.424	0.440	0.407	0.419
Belgium	0.880	0.880	0.880	0.899	0.903	0.896	0.905	0.907	0.904	0.898	0.884	0.875	0.858	0.838	0.800
Belize	0.177	0.179	0.172	0.190	0.196	0.194	0.193	0.192	0.186	0.194	0.191	0.179	0.175	0.171	0.180
Bhutan	0.072	0.070	0.068	0.070	0.073	0.076	0.093	0.115	0.082	0.126	0.129	0.138	0.150	0.168	0.177
Bolivia	0.044	0.047	0.050	0.098	0.103	0.118	0.134	0.160	0.174	0.197	0.220	0.238	0.261	0.272	0.274
Bosnia and Herzegovina	0.208	0.208	0.213	0.248	0.280	0.304	0.280	0.294	0.299	0.312	0.325	0.315	0.308	0.311	0.316
Botswana	0.060	0.061	0.068	0.097	0.096	0.093	0.098	0.092	0.092	0.096	0.099	0.103	0.103	0.103	0.124
Brazil	0.300	0.307	0.303	0.308	0.317	0.322	0.327	0.327	0.332	0.339	0.342	0.335	0.327	0.314	0.306
Brunel Darussalam	0.203	0.252	0.244	0.271	0.303	0.334	0.332	0.330	0.356	0.336	0.319	0.319	0.304	0.281	0.287
Burundi	0.429	0.449	0.495	0.030	0.004	0.382	0.380	0.040	0.040	0.343	0.330	0.330	0.317	0.329	0.327
Cambodia	0.012	0.013	0.014	0.020	0.026	0.033	0.022	0.025	0.023	0.048	0.060	0.069	0.079	0.086	0.095
Cameroon	0.002	0.005	0.006	0.006	0.007	0.009	0.011	0.013	0.014	0.016	0.017	0.018	0.019	0.018	0.021
Central African Republic	0.000	0.000	0.000	0.001	0.001	0.003	0.004	0.005	0.005	0.006	0.005	0.003	0.004	0.004	0.004
Chad	0.000	0.000	0.000	0.000	0.001	0.002	0.002	0.003	0.002	0.004	0.005	0.005	0.006	0.006	0.006
Chile	0.140	0.155	0.169	0.183	0.196	0.200	0.210	0.215	0.221	0.213	0.198	0.193	0.186	0.179	0.171
China	0.086	0.086	0.086	0.093	0.104	0.116	0.131	0.148	0.169	0.197	0.224	0.294	0.310	0.319	0.358
Colombia	0.127	0.127	0.127	0.127	0.131	0.134	0.147	0.147	0.157	0.168	0.171	0.176	0.175	0.174	0.172
Comoros Congo Dom Bon	0.006	0.006	0.006	0.013	0.013	0.017	0.019	0.049	0.046	0.049	0.051	0.050	0.055	0.056	0.057
Congo, Deni, Kep.	0.001	0.001	0.001	0.001	0.001	0.001	0.002	0.003	0.003	0.003	0.005	0.006	0.006	0.006	0.006
Costa Rica	0.003	0.003	0.003	0.007	0.009	0.010	0.012	0.010	0.017	0.022	0.028	0.031	0.034	0.030	0.030
Croatia	0.342	0.354	0.377	0.412	0.447	0.473	0.483	0.493	0.499	0.495	0.497	0.509	0.514	0.518	0.507
Cyprus	0.529	0.528	0.548	0.549	0.562	0.574	0.578	0.570	0.547	0.433	0.426	0.409	0.393	0.508	0.431
Czech Republic	0.249	0.260	0.273	0.273	0.284	0.289	0.294	0.305	0.315	0.328	0.329	0.329	0.324	0.324	0.329
Djibouti	0.011	0.011	0.013	0.013	0.017	0.021	0.027	0.031	0.034	0.035	0.042	0.048	0.050	0.063	0.063
Dominican Republic	0.135	0.151	0.154	0.160	0.164	0.170	0.174	0.184	0.188	0.205	0.209	0.227	0.234	0.240	0.231
Ecuador	0.075	0.076	0.091	0.142	0.142	0.162	0.168	0.130	0.135	0.133	0.139	0.142	0.140	0.128	0.137
Egypt, Arab Rep.	0.027	0.029	0.032	0.037	0.041	0.044	0.047	0.049	0.051	0.053	0.056	0.060	0.066	0.072	0.076
El Salvador	0.187	0.196	0.201	0.207	0.217	0.222	0.215	0.224	0.227	0.246	0.248	0.257	0.280	0.300	0.307
Equatorial Guinea	0.012	0.012	0.014	0.016	0.019	0.019	0.022	0.027	0.032	0.030	0.033	0.040	0.044	0.049	0.056
Eswatini	0.049	0.060	0.067	0.069	0.078	0.079	0.087	0.094	0.104	0.106	0.121	0.128	0.124	0.117	0.128
Fiji	0.089	0.093	0.102	0.109	0.118	0.119	0.131	0.137	0.136	0.151	0.166	0.171	0.181	0.178	0.180
Finland	0.140	0.146	0.159	0.154	0.153	0.147	0.149	0.146	0.137	0.131	0.130	0.097	0.088	0.069	0.082
Gabon	0.028	0.028	0.028	0.030	0.029	0.033	0.038	0.043	0.064	0.067	0.067	0.067	0.067	0.067	0.067
Gambia, The	0.027	0.029	0.034	0.041	0.050	0.058	0.070	0.071	0.072	0.077	0.077	0.079	0.079	0.079	0.082
Georgia	0.063	0.080	0.112	0.158	0.221	0.218	0.230	0.250	0.303	0.341	0.345	0.341	0.354	0.360	0.353
Ghana	0.026	0.027	0.030	0.036	0.037	0.040	0.042	0.042	0.048	0.056	0.058	0.070	0.073	0.084	0.086
Greece	0.388	0.396	0.420	0.454	0.480	0.478	0.468	0.447	0.429	0.307	0.332	0.320	0.307	0.301	0.290
Guatemala	0.185	0.185	0.182	0.230	0.270	0.288	0.295	0.302	0.318	0.329	0.555	0.559	0.339	0.337	0.291
Guvana	0.047	0.048	0.050	0.053	0.055	0.056	0.057	0.062	0.065	0.066	0.068	0.070	0.073	0.075	0.076
Haiti	0.034	0.034	0.034	0.033	0.033	0.033	0.029	0.031	0.032	0.033	0.033	0.034	0.037	0.039	0.039
Honduras	0.095	0.103	0.119	0.145	0.160	0.161	0.148	0.162	0.175	0.180	0.180	0.170	0.174	0.174	0.173
Hungary	0.202	0.217	0.235	0.259	0.275	0.275	0.272	0.272	0.266	0.262	0.265	0.260	0.263	0.267	0.262
Iceland	0.434	0.443	0.434	0.434	0.425	0.384	0.388	0.378	0.359	0.344	0.335	0.307	0.330	0.326	0.306
India	0.107	0.108	0.111	0.117	0.126	0.135	0.151	0.166	0.186	0.205	0.244	0.265	0.285	0.296	0.297
Indonesia	0.057	0.060	0.066	0.071	0.080	0.091	0.092	0.149	0.219	0.243	0.266	0.277	0.279	0.279	0.274
Iteldiu	0.414	0.404	0.404	0.409	0.420	0.423	0.376	0.300	0.342	0.332	0.359	0.345	0.341	0.334	0.309
Iamaica	0.729	0.737	0.739	0.792	0.840	0.807	0.839	0.845	0.850	0.757	0.790	0.758	0.770	0.750	0.097
Japan	0.920	0.918	0.917	0.917	0.917	0.921	0.921	0.922	0.923	0.922	0.921	0.924	0.924	0.923	0.922
Jordan	0.149	0.147	0.143	0.144	0.148	0.149	0.156	0.157	0.157	0.156	0.157	0.154	0.161	0.162	0.168
Kenya	0.017	0.017	0.020	0.029	0.035	0.039	0.045	0.047	0.050	0.051	0.053	0.054	0.052	0.052	0.050
Kiribati	0.067	0.067	0.067	0.067	0.067	0.067	0.067	0.067	0.066	0.076	0.076	0.076	0.076	0.076	0.076
Korea, Rep.	0.752	0.761	0.771	0.780	0.788	0.783	0.785	0.788	0.791	0.789	0.778	0.773	0.766	0.754	0.752
Kosovo	0.209	0.208	0.211	0.219	0.237	0.258	0.274	0.283	0.286	0.280	0.269	0.279	0.271	0.243	0.239
Lao PDR	0.016	0.016	0.016	0.016	0.016	0.020	0.029	0.033	0.038	0.048	0.053	0.059	0.062	0.066	0.067
Latvia	0.268	0.270	0.287	0.331	0.342	0.342	0.342	0.318	0.283	0.257	0.238	0.226	0.217	0.215	0.203
Lepanon	0.469	0.482	0.490	0.503	0.516	0.530	0.545	0.555	0.565	0.5/5	0.585	0.594	0.602	0.013	0.625
Lesollio	0.018	0.020	0.018	0.021	0.023	0.020	0.033	0.030	0.038	0.043	0.044	0.049	0.054	0.053	0.033
Liberta	0.005	0.005	0.005	0.007	0.010	0.017	0.021	0.021	0.021	0.022	0.022	0.019	0.019	0.020	0.022
Libya	0.04/	0.04/	0.000	0.050	0.000	0.037	0.000	0.030	0.000	0.002	0.002	0.001	0.001	0.000	0.000

(continued on next page)

Table A2 (continued)

coutreach															
Country	2004	2005	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016	2017	2018
Madagascar	0.005	0.005	0.006	0.007	0.008	0.009	0.009	0.010	0.011	0.011	0.013	0.014	0.016	0.017	0.017
Malawi	0.007	0.009	0.008	0.016	0.018	0.021	0.027	0.016	0.034	0.035	0.036	0.037	0.037	0.037	0.037
Malaysia	0.146	0.139	0.137	0.160	0.167	0.192	0.196	0.199	0.199	0.202	0.197	0.194	0.188	0.184	0.183
Maldives	0.283	0.295	0.344	0.462	0.479	0.489	0.489	0.494	0.534	0.552	0.603	0.618	0.636	0.649	0.653
Malta	0.799	0.800	0.785	0.800	0.808	0.811	0.814	0.829	0.821	0.818	0.808	0.809	0.795	0.784	0.768
Mauritania	0.026	0.026	0.026	0.026	0.027	0.027	0.027	0.027	0.030	0.043	0.047	0.051	0.058	0.059	0.070
Mauritius	0.481	0.506	0.522	0.557	0.566	0.606	0.622	0.644	0.654	0.661	0.665	0.666	0.653	0.619	0.594
Mexico	0.123	0.130	0.139	0.155	0.169	0.171	0.179	0.181	0.193	0.193	0.197	0.199	0.204	0.207	0.214
Micronesia, Fed. Sts.	0.094	0.094	0.093	0.093	0.093	0.115	0.130	0.129	0.144	0.141	0.139	0.137	0.136	0.134	0.133
Moldova	0.226	0.251	0.265	0.290	0.329	0.334	0.344	0.372	0.381	0.393	0.412	0.279	0.283	0.286	0.290
Mongolia	0.207	0.215	0.236	0.256	0.256	0.265	0.272	0.295	0.319	0.328	0.344	0.370	0.398	0.407	0.432
Montenegro	0.148	0.171	0.209	0.281	0.349	0.375	0.362	0.370	0.379	0.394	0.390	0.407	0.419	0.422	0.435
Morocco	0.076	0.095	0.096	0.106	0.120	0.158	0.169	0.180	0.191	0.199	0.205	0.211	0.214	0.216	0.219
Mozambique	0.011	0.013	0.016	0.018	0.020	0.023	0.027	0.030	0.032	0.034	0.038	0.042	0.043	0.043	0.042
Myanmar	0.010	0.009	0.007	0.007	0.008	0.008	0.008	0.009	0.010	0.017	0.023	0.024	0.026	0.038	0.044
Nonal	0.070	0.009	0.074	0.077	0.114	0.150	0.149	0.155	0.150	0.157	0.100	0.109	0.195	0.170	0.164
Nothorlands	0.039	0.039	0.040	0.047	0.047	0.030	0.000	0.078	0.000	0.091	0.090	0.102	0.112	0.130	0.109
Nicaragua	0.732	0.707	0.733	0.750	0.755	0.755	0.070	0.047	0.013	0.371	0.338	0.003	0.472	0.449	0.415
North Macedonia	0.032	0.042	0.040	0.033	0.000	0.000	0.034	0.038	0.002	0.070	0.075	0.085	0.099	0.100	0.100
Norway	0.151	0.120	0.102	0.204	0.205	0.262	0.285	0.280	0.275	0.231	0.230	0.307	0.308	0.011	0.002
Oman	0.171	0.170	0.172	0.175	0.105	0.105	0.150	0.150	0.145	0.140	0.135	0.125	0.104	0.030	0.032
Pakistan	0.060	0.062	0.066	0.071	0.077	0.080	0.082	0.086	0.091	0.098	0.145	0.143	0.145	0.144	0.130
Panama	0.187	0.187	0.187	0.187	0.197	0.205	0.211	0.222	0.233	0.253	0.267	0.280	0.283	0.280	0.277
Papua New Guinea	0.014	0.014	0.013	0.014	0.015	0.015	0.017	0.020	0.020	0.021	0.021	0.020	0.023	0.023	0.022
Paraguay	0.043	0.043	0.043	0.044	0.052	0.058	0.077	0.082	0.087	0.095	0.090	0.098	0.097	0.102	0.106
Peru	0.041	0.043	0.048	0.057	0.072	0.080	0.087	0.097	0.115	0.122	0.152	0.283	0.258	0.250	0.258
Philippines	0.110	0.114	0.117	0.116	0.121	0.125	0.132	0.141	0.153	0.171	0.182	0.193	0.204	0.213	0.219
Poland	0.265	0.271	0.289	0.319	0.361	0.379	0.380	0.391	0.408	0.407	0.421	0.422	0.433	0.418	0.418
Portugal	0.762	0.778	0.793	0.785	0.802	0.811	0.815	0.807	0.793	0.780	0.765	0.648	0.695	0.695	0.666
Romania	0.289	0.304	0.328	0.349	0.377	0.380	0.381	0.394	0.380	0.367	0.364	0.359	0.349	0.340	0.327
Rwanda	0.003	0.007	0.007	0.011	0.048	0.054	0.058	0.070	0.082	0.089	0.091	0.096	0.099	0.099	0.094
Samoa	0.125	0.129	0.144	0.170	0.172	0.196	0.199	0.163	0.169	0.209	0.233	0.219	0.242	0.248	0.257
San Marino	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000
Sao Tome and Principe	0.265	0.265	0.265	0.265	0.265	0.265	0.265	0.265	0.298	0.301	0.301	0.305	0.262	0.271	0.241
Saudi Arabia	0.089	0.092	0.107	0.121	0.132	0.140	0.146	0.150	0.153	0.159	0.170	0.180	0.182	0.182	0.182
Serbia	0.177	0.240	0.289	0.347	0.395	0.394	0.382	0.371	0.355	0.326	0.303	0.301	0.311	0.290	0.294
Seychelles	0.446	0.430	0.447	0.450	0.524	0.531	0.537	0.545	0.574	0.628	0.663	0.671	0.692	0.718	0.726
Solomon Islands	0.038	0.041	0.047	0.049	0.050	0.050	0.042	0.040	0.040	0.038	0.040	0.040	0.041	0.043	0.043
South Sudan	0.084	0.087	0.092	0.093	0.129	0.154	0.107	0.174	0.173	0.176	0.196	0.201	0.201	0.200	0.197
South Suddi	0.003	0.003	0.003	0.003	0.003	0.003	0.003	0.003	0.005	0.007	0.010	0.009	0.008	0.007	0.000
Suriname	0.750	0.760	0.775	0.787	0.790	0.760	0.772	0.750	0.756	0.715	0.079	0.070	0.050	0.041	0.027
Sweden	0.001	0.000	0.035	0.102	0.101	0.112	0.117	0.115	0.127	0.155	0.145	0.157	0.145	0.147	0.142
Switzerland	0.155	0.758	0.150	0.202	0.207	0.205	0.201	0.201	0.787	0.152	0.764	0.757	0.742	0.729	0.724
Svrian Arab Republic	0.014	0.015	0.016	0.017	0.024	0.039	0.043	0.045	0.047	0.047	0.047	0.047	0.047	0.047	0.047
Tanzania	0.010	0.010	0.010	0.011	0.013	0.016	0.018	0.020	0.022	0.024	0.025	0.026	0.026	0.026	0.026
Thailand	0.112	0.144	0.183	0.208	0.259	0.286	0.312	0.331	0.356	0.382	0.420	0.430	0.431	0.429	0.428
Tonga	0.225	0.202	0.192	0.202	0.240	0.228	0.223	0.223	0.213	0.225	0.200	0.286	0.306	0.349	0.341
Trinidad and Tobago	0.244	0.244	0.244	0.244	0.244	0.253	0.264	0.260	0.268	0.277	0.286	0.286	0.289	0.287	0.286
Turkey	0.149	0.154	0.169	0.187	0.214	0.223	0.243	0.266	0.285	0.315	0.329	0.335	0.327	0.326	0.329
Uganda	0.009	0.010	0.011	0.013	0.019	0.023	0.025	0.026	0.029	0.032	0.032	0.033	0.032	0.030	0.030
Ukraine	0.072	0.093	0.114	0.153	0.196	0.200	0.202	0.216	0.233	0.260	0.236	0.215	0.218	0.240	0.238
United Arab Emirates	0.128	0.176	0.170	0.177	0.190	0.227	0.227	0.240	0.253	0.262	0.271	0.281	0.282	0.275	0.270
United Kingdom	0.732	0.726	0.723	0.745	0.744	0.724	0.724	0.726	0.718	0.761	0.761	0.761	0.761	0.761	0.719
Uzbekistan	0.225	0.232	0.246	0.242	0.240	0.240	0.240	0.282	0.285	0.298	0.278	0.211	0.220	0.290	0.287
Vanuatu	0.106	0.106	0.106	0.123	0.137	0.143	0.156	0.158	0.177	0.180	0.182	0.187	0.186	0.199	0.197
Venezuela, RB	0.127	0.127	0.135	0.143	0.147	0.153	0.169	0.174	0.176	0.180	0.181	0.185	0.185	0.185	0.185
Vietnam	0.036	0.042	0.044	0.060	0.077	0.089	0.098	0.112	0.113	0.124	0.130	0.134	0.138	0.136	0.145
West Bank and Gaza	0.1423	0.1436	0.1507	0.1699	0.1958	0.2208	0.2261	0.2427	0.2568	0.2698	0.2943	0.3106	0.3352	0.3547	0.4308
Yemen, Rep.	0.009	0.0106	0.0118	0.0132	0.0141	0.0149	0.0155	0.0157	0.0165	0.0178	0.0211	0.0208	0.0208	0.0208	0.0208
Zambia	0.0152	0.0155	0.0171	0.02	0.0248	0.0292	0.0311	0.0323	0.0361	0.0392	0.0427	0.0441	0.0446	0.0441	0.0405
Zimbabwe	0.0278	0.0331	0.0319	0.035	0.0311	0.0379	0.0343	0.0398	0.0762	0.08	0.088	0.0444	0.0433	0.042	0.0407

Source: Own estimates using Financial Access Survey, 2004–2018 (IMF, 2019).

Financial usage index.

cusage															
Country	2004	2005	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016	2017	2018
Afghanistan	0.029	0.029	0.029	0.029	0.029	0.029	0.029	0.026	0.042	0.033	0.044	0.047	0.047	0.044	0.045
Albania	0.493	0.493	0.493	0.493	0.493	0.493	0.493	0.493	0.493	0.493	0.493	0.493	0.493	0.422	0.402
Algeria	0.163	0.165	0.173	0.185	0.185	0.119	0.136	0.142	0.150	0.154	0.156	0.154	0.149	0.151	0.152
Angola	0.004	0.004	0.004	0.004	0.004	0.004	0.004	0.004	0.004	0.004	0.004	0.004	0.004	0.006	0.006
Argentina Armenia	0.208	0.239	0.255	0.284	0.303	0.321	0.338	0.301	0.405	0.414	0.437	0.494	0.547	0.595	0.634
Austria	0.349	0.356	0.353	0.358	0.359	0.349	0.342	0.235	0.285	0.345	0.420	0.313	0.307	0.279	0.292
Azerbaijan	0.082	0.082	0.082	0.082	0.082	0.082	0.114	0.144	0.205	0.310	0.476	0.547	0.507	0.535	0.591
Bahamas, The	0.671	0.703	0.731	0.768	0.775	0.751	0.681	0.668	0.630	0.589	0.560	0.531	0.554	0.546	0.538
Bangladesh	0.111	0.114	0.117	0.118	0.122	0.124	0.144	0.159	0.168	0.179	0.184	0.198	0.209	0.222	0.238
Belgium	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000
BellZe Bhutan	0.306	0.306	0.306	0.306	0.306	0.306	0.306	0.306	0.293	0.286	0.270	0.245	0.253	0.231	0.238
Bolivia	0.032	0.034	0.036	0.040	0.050	0.059	0.070	0.083	0.095	0.230	0.127	0.144	0.162	0.196	0.208
Bosnia and Herzegovina	0.244	0.270	0.308	0.337	0.356	0.364	0.362	0.386	0.400	0.429	0.429	0.434	0.430	0.431	0.442
Botswana	0.300	0.300	0.300	0.300	0.300	0.300	0.300	0.300	0.309	0.311	0.312	0.285	0.284	0.288	0.342
Brazil	0.253	0.259	0.269	0.283	0.318	0.359	0.404	0.460	0.784	0.857	0.862	0.852	1.000	0.966	0.968
Brunei Darussalam	0.696	0.696	0.696	0.696	0.696	0.715	0.689	0.684	0.839	0.795	0.700	0.702	0.621	0.616	0.612
Burundi	0.447	0.447	0.499	0.006	0.025	0.033	0.015	0.020	0.025	0.012	0.599	0.576	0.565	0.555	0.361
Cambodia	0.025	0.025	0.025	0.025	0.025	0.032	0.035	0.040	0.046	0.053	0.065	0.074	0.086	0.095	0.110
Cameroon	0.000	0.000	0.006	0.013	0.018	0.021	0.022	0.026	0.028	0.031	0.026	0.033	0.034	0.034	0.050
Central African Republic	0.001	0.001	0.001	0.004	0.007	0.009	0.011	0.013	0.014	0.020	0.016	0.018	0.021	0.020	0.020
Chad	0.009	0.009	0.009	0.009	0.009	0.009	0.009	0.009	0.009	0.009	0.010	0.011	0.012	0.011	0.011
Chile	0.495	0.518	0.589	0.643	0.689	0.718	0.758	0.815	0.853	0.887	0.899	0.910	0.924	0.945	0.974
Colombia	0.005	0.005	0.005	0.005	0.005	0.004	0.004	0.005	0.005	0.000	0.007	0.007	0.008	0.009	0.010
Comoros	0.018	0.018	0.018	0.018	0.018	0.018	0.018	0.025	0.027	0.030	0.033	0.035	0.037	0.037	0.038
Congo, Dem. Rep.	0.003	0.003	0.003	0.003	0.003	0.003	0.004	0.005	0.010	0.014	0.014	0.014	0.014	0.014	0.014
Congo, Rep.	0.003	0.005	0.005	0.011	0.012	0.015	0.017	0.020	0.024	0.026	0.036	0.041	0.041	0.046	0.046
Costa Rica	0.312	0.410	0.427	0.467	0.480	0.473	0.467	0.483	0.495	0.525	0.545	0.582	0.692	0.762	0.769
CUDILIA	0.322	0.322	0.322	0.322	0.322	0.322	0.322	0.308	0.278	0.277	0.237	0.228	0.207	0.164	0.167
Czech Republic	0.338	0.350	0.354	0.394	0.420	0.436	0.455	0.483	0.507	0.547	0.591	0.402	0.409	0.421	0.435
Djibouti	0.016	0.016	0.016	0.016	0.016	0.021	0.026	0.029	0.031	0.049	0.057	0.066	0.052	0.074	0.074
Dominican Republic	0.231	0.231	0.231	0.237	0.240	0.241	0.246	0.250	0.253	0.276	0.290	0.324	0.339	0.332	0.344
Ecuador	0.152	0.177	0.140	0.151	0.185	0.192	0.208	0.241	0.228	0.216	0.242	0.246	0.248	0.263	0.263
Egypt, Arab Kep. Fl Salvador	0.115	0.115	0.115	0.116	0.118	0.115	0.113	0.112	0.116	0.139	0.141	0.144	0.152	0.200	0.218
Equatorial Guinea	0.023	0.026	0.029	0.034	0.035	0.038	0.040	0.044	0.049	0.046	0.051	0.056	0.058	0.082	0.094
Estonia	0.794	0.794	0.908	1.000	0.921	0.922	0.972	0.949	0.765	0.773	0.779	0.746	0.744	0.745	0.776
Eswatini	0.172	0.172	0.172	0.172	0.172	0.172	0.172	0.172	0.200	0.211	0.232	0.234	0.210	0.213	0.209
Fiji Finland	0.184	0.229	0.295	0.250	0.271	0.270	0.286	0.303	0.300	0.323	0.368	0.410	0.457	0.497	0.485
Gabon	0.576	0.576	0.576	0.576	0.576	0.576	0.576	0.576	0.578	0.564	0.558	0.587	0.554	0.277	0.467
Gambia. The	0.026	0.020	0.020	0.052	0.086	0.079	0.053	0.095	0.007	0.096	0.100	0.100	0.115	0.092	0.100
Georgia	0.085	0.129	0.178	0.231	0.297	0.304	0.341	0.417	0.482	0.530	0.641	0.656	0.728	0.837	0.862
Ghana	0.045	0.045	0.052	0.055	0.070	0.073	0.095	0.107	0.118	0.122	0.123	0.167	0.157	0.188	0.203
Greece	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000
Guinea	0.230	0.258	0.204	0.295	0.552	0.545	0.300	0.400	0.440	0.449	0.492	0.495	0.479	0.448	0.405
Guyana	0.214	0.225	0.176	0.215	0.214	0.259	0.264	0.272	0.277	0.283	0.278	0.272	0.262	0.257	0.249
Haiti	0.053	0.061	0.070	0.077	0.085	0.090	0.087	0.086	0.086	0.079	0.074	0.074	0.072	0.074	0.079
Honduras	0.207	0.202	0.212	0.232	0.232	0.235	0.242	0.246	0.270	0.272	0.297	0.321	0.328	0.324	0.329
Hungary	0.397	0.408	0.423	0.439	0.450	0.457	0.463	0.465	0.461	0.462	0.437	0.423	0.424	0.430	0.428
India	0.174	0.177	0.181	0.193	0.210	0.231	0.251	0.268	0.292	0.326	0.372	0.424	0.475	0.517	0.536
Indonesia	0.153	0.161	0.150	0.155	0.163	0.172	0.194	0.200	0.232	0.268	0.280	0.288	0.318	0.445	0.452
Ireland	0.437	0.433	0.437	0.550	0.555	0.539	0.561	0.516	0.503	0.484	0.613	0.644	0.646	0.580	0.565
Italy	0.310	0.313	0.316	0.321	0.349	0.311	0.326	0.334	0.329	0.328	0.332	0.360	0.369	0.380	0.371
Jamaica	0.371	0.357	0.344	0.328	0.323	0.316	0.305	0.288	0.302	0.307	0.320	0.325	0.332	0.415	0.427
Japan Jordan	0.230	0.236	0.243	0.243	0.240	0.246	0.195	0 1 9 1	0.189	0.178	0.169	0.168	0.176	0.181	0.182
Kenya	0.033	0.041	0.060	0.068	0.092	0.112	0.145	0.166	0.177	0.238	0.303	0.370	0.425	0.457	0.506
Kiribati	0.051	0.051	0.051	0.051	0.051	0.051	0.051	0.051	0.057	0.066	0.066	0.066	0.066	0.066	0.066
Korea, Rep.	1.000	0.999	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000
Kosovo	0.152	0.162	0.188	0.228	0.269	0.297	0.301	0.351	0.366	0.379	0.377	0.357	0.386	0.335	0.299
LaO PDK Latvia	0.104	0.104	0.104	0.104	0.104	0.104	0.104	0.104	0.104	0.104	0.127	0.142	0.151	0.164	0.1/6
Lebanon	0.297	0.297	0.313	0.336	0.380	0.412	0.433	0.820	0.802	0.395	0.364	0.354	0.343	0.349	0.356
Lesotho	0.085	0.085	0.085	0.085	0.085	0.092	0.100	0.108	0.093	0.109	0.111	0.114	0.116	0.122	0.133
Liberia	0.019	0.019	0.019	0.019	0.019	0.025	0.035	0.043	0.052	0.072	0.075	0.085	0.095	0.095	0.091
Libya	0.030	0.029	0.029	0.028	0.029	0.037	0.038	0.034	0.034	0.034	0.034	0.034	0.034	0.034	0.034

(continued on next page)

Table A3 (continued)

cusage															
Country	2004	2005	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016	2017	2018
Madagascar	0.007	0.009	0.009	0.010	0.014	0.015	0.031	0.018	0.020	0.023	0.028	0.031	0.031	0.037	0.044
Malawi	0.076	0.076	0.076	0.076	0.076	0.076	0.076	0.076	0.065	0.070	0.070	0.086	0.086	0.086	0.086
Malaysia	0.667	0.723	0.737	0.738	0.751	0.747	0.760	0.774	0.798	0.841	0.839	0.795	0.788	0.785	0.781
Maldives	0.387	0.387	0.387	0.387	0.387	0.387	0.387	0.387	0.309	0.359	0.344	0.377	0.381	0.364	0.368
Malta	0.866	0.880	0.911	0.976	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	1.000	0.924
Mauritania	0.039	0.039	0.039	0.039	0.039	0.039	0.039	0.042	0.043	0.046	0.046	0.050	0.055	0.059	0.062
Mauritius	0.554	0.600	0.584	0.613	0.640	0.656	0.668	0.688	0.696	0.708	0.677	0.665	0.624	0.640	0.633
Mexico	0.129	0.131	0.154	0.159	0.251	0.258	0.294	0.231	0.241	0.242	0.231	0.244	0.262	0.243	0.255
Micronesia, Fed. Sts.	0.127	0.128	0.131	0.128	0.129	0.142	0.149	0.155	0.157	0.158	0.159	0.129	0.121	0.114	0.112
Moldova	0.249	0.269	0.291	0.332	0.364	0.366	0.384	0.400	0.413	0.440	0.426	0.296	0.335	0.357	0.374
Mongolia	0.110	0.132	0.142	0.221	0.238	0.244	0.278	0.310	0.338	0.385	0.419	0.438	0.543	0.463	0.504
Morecco	0.343	0.343	0.343	0.343	0.592	0.548	0.550	0.587	0.010	0.557	0.574	0.210	0.008	0.506	0.359
Morombiguo	0.170	0.105	0.174	0.185	0.192	0.254	0.259	0.270	0.280	0.290	0.300	0.320	0.342	0.376	0.384
Myanmar	0.015	0.015	0.021	0.020	0.029	0.034	0.035	0.040	0.045	0.039	0.005	0.075	0.080	0.077	0.078
Namibia	0.028	0.029	0.029	0.050	0.031	0.031	0.032	0.030	0.030	0.039	0.044	0.037	0.004	0.037	0.074
Nepal	0.123	0.123	0.123	0.123	0.123	0.123	0.123	0.123	0.123	0.129	0.147	0.551	0.187	0.225	0.260
Netherlands	0.627	0.627	0.623	0.625	0.624	0.646	0.648	0.615	0.585	0.602	0.665	0.606	0.594	0.575	0.601
Nicaragua	0.109	0.117	0.122	0.133	0.140	0.118	0.106	0.114	0.113	0.118	0.133	0.146	0.165	0.181	0.157
North Macedonia	0.456	0.456	0.456	0.456	0.578	0.629	0.649	0.644	0.663	0.680	0.705	0.728	0.733	0.735	0.750
Norway	0.238	0.238	0.272	0.278	0.339	0.281	0.279	0.284	0.282	0.293	0.291	0.293	0.292	0.289	0.287
Oman	0.160	0.174	0.188	0.206	0.225	0.234	0.244	0.265	0.256	0.281	0.269	0.271	0.273	0.265	0.271
Pakistan	0.050	0.053	0.065	0.070	0.071	0.067	0.071	0.076	0.080	0.083	0.089	0.094	0.098	0.104	0.106
Panama	0.333	0.333	0.333	0.333	0.323	0.317	0.361	0.384	0.422	0.460	0.488	0.505	0.520	0.525	0.530
Papua New Guinea	0.098	0.098	0.097	0.095	0.096	0.097	0.097	0.103	0.102	0.101	0.103	0.103	0.111	0.126	0.130
Paraguay	0.013	0.013	0.016	0.016	0.020	0.027	0.032	0.038	0.044	0.048	0.052	0.060	0.065	0.169	0.211
Peru	0.206	0.135	0.155	0.188	0.229	0.225	0.257	0.292	0.299	0.326	0.362	0.390	0.417	0.425	0.450
Philippines	0.089	0.094	0.097	0.095	0.100	0.103	0.120	0.134	0.122	0.130	0.132	0.139	0.144	0.147	0.159
Poland	0.745	0.745	0.745	0.745	0.745	0.757	0.754	0.742	0.717	0.713	0.729	0.755	0.774	0.801	0.847
Portugal	0.831	0.831	0.831	0.831	0.849	0.845	0.850	0.849	0.769	0.777	0.781	0.784	0.729	0.724	0.744
Rwanda	0.102	0.102	0.102	0.102	0.102	0.101	0.037	0.092	0.092	0.088	0.085	0.085	0.085	0.080	0.094
Samoa	0.002	0.002	0.005	0.000	0.048	0.050	0.070	0.323	0.087	0.001	0.070	0.055	0.007	0.005	0.509
San Marino	0.972	1 000	0.105	1 000	1 000	1 000	0.217	0.952	0.944	1 000	0.987	0.849	0.851	0.846	0.831
Sao Tome and Principe	0.196	0.196	0.196	0.196	0.196	0.196	0.196	0.196	0.233	0.284	0.262	0.337	0.260	0.277	0.277
Saudi Arabia	0.260	0.260	0.260	0.260	0.260	0.260	0.265	0.263	0.263	0.309	0.311	0.321	0.344	0.366	0.376
Serbia	0.045	0.045	0.045	0.053	0.057	0.054	0.058	0.059	0.163	0.161	0.165	0.180	0.185	0.186	0.187
Seychelles	0.365	0.436	0.449	0.440	0.447	0.474	0.473	0.495	0.580	0.556	0.595	0.579	0.596	0.633	0.646
Solomon Islands	0.080	0.082	0.075	0.075	0.088	0.087	0.085	0.083	0.086	0.071	0.122	0.130	0.134	0.139	0.139
South Africa	0.230	0.264	0.299	0.316	0.326	0.352	0.377	0.385	0.455	0.495	0.509	0.494	0.518	0.524	0.461
South Sudan	0.005	0.005	0.005	0.005	0.005	0.005	0.005	0.005	0.008	0.010	0.016	0.021	0.027	0.024	0.022
Spain	0.675	0.703	0.705	0.731	0.728	0.750	0.750	0.728	0.711	0.722	0.717	0.697	0.679	0.711	0.710
Suriname	0.422	0.422	0.422	0.422	0.422	0.422	0.422	0.422	0.422	0.422	0.422	0.422	0.478	0.463	0.498
Sweden	0.947	0.947	0.947	0.947	0.947	0.947	0.947	0.968	1.000	1.000	1.000	1.000	1.000	1.000	1.000
Switzerland	0.761	0.757	0.742	0.732	0.713	0.744	0.738	0.752	0.748	0.807	0.816	0.747	0.650	0.636	0.603
	0.071	0.071	0.071	0.071	0.071	0.075	0.076	0.077	0.074	0.075	0.075	0.075	0.075	0.075	0.075
Thailand	0.040	0.040	0.040	0.040	0.031	0.040	0.033	0.002	0.004	0.005	0.115	0.075	0.075	0.075	0.075
Tonga	0.520	0.348	0.278	0.400	0.415	0.415	0.425	0.420	0.455	0.400	0.461	0.450	0.450	0.303	0.352
Trinidad and Tobago	0.513	0.513	0.513	0.513	0.513	0.528	0.542	0.545	0.548	0.554	0.550	0.550	0.548	0.527	0.520
Turkev	0.427	0.427	0.434	0.453	0.427	0.432	0.747	0.769	0.800	0.867	0.840	0.861	0.921	1.000	1.000
Uganda	0.028	0.030	0.033	0.034	0.044	0.048	0.053	0.050	0.056	0.053	0.058	0.068	0.075	0.102	0.132
Ukraine	0.828	0.828	0.891	0.966	0.969	1.000	1.000	1.000	1.000	0.750	0.766	0.735	0.718	0.762	0.630
United Arab Emirates	0.363	0.363	0.363	0.363	0.363	0.363	0.368	0.368	0.365	0.340	0.398	0.437	0.458	0.484	0.472
United Kingdom	0.037	0.036	0.035	0.034	0.034	0.035	0.031	0.029	0.028	0.027	0.027	0.027	0.027	0.027	0.027
Uzbekistan	0.094	0.116	0.130	0.145	0.163	0.168	0.191	0.187	0.193	0.221	0.254	0.280	0.313	0.362	0.362
Vanuatu	0.239	0.239	0.239	0.239	0.239	0.239	0.239	0.239	0.239	0.239	0.239	0.255	0.260	0.239	0.240
Venezuela, RB	0.056	0.056	0.079	0.114	0.118	0.123	0.116	0.124	0.131	0.171	0.200	0.275	0.275	0.275	0.275
Vietnam	0.062	0.062	0.062	0.062	0.062	0.062	0.062	0.125	0.153	0.168	0.193	0.212	0.239	0.238	0.273
West Bank and Gaza	0.2683	0.2683	0.2683	0.2683	0.2683	0.2674	0.2791	0.3192	0.3282	0.3227	0.3186	0.3266	0.3354	0.3411	0.3573
Yemen, Kep.	0.0291	0.0291	0.0291	0.0291	0.0291	0.0291	0.0291	0.0269	0.0302	0.0317	0.0326	0.0312	0.0312	0.0312	0.0312
Zambia	0.03/6	0.03/6	0.03/6	0.03/6	0.03/6	0.03/6	0.03//	0.0426	0.0541	0.0642	0.0818	0.0849	0.085/	0.0891	0.0798
ZIIIDaDwe	0.1308	0.131	0.1307	0.0544	0.0559	0.0258	0.0456	0.0386	0.0523	0.0642	0.0437	0.0429	0.046	0.12//	0.190

Source: Own estimates using Financial Access Survey, 2004–2018 (IMF, 2019).

Table A4
Growth-redistribution decomposition of those living under \$1.90 dollars a day in low- and low-middle income countries. First-stage IV specification.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)	(13)	(14)	(15)	(16)	(17)	(18)	(19)	(20)	(21)	(22)	(23)	(24)
First-stage model for:	ΔHeadco	unt					ΔPoverty	gap					ΔPoverty	gap squared					ΔWatts					
Dependent variable:	ΔFindex	$\Delta Gini \times \Delta findex$	ΔFindex outreach	$\begin{array}{l} \Delta Gini \times \Delta Financial \\ outreach \end{array}$	ΔFinancial usage	$\begin{array}{l} \Delta Gini \times \Delta Financial \\ usage \end{array}$	ΔFindex	$\Delta \text{Gini} \times \Delta \text{findex}$	ΔFindex outreach	$\Delta Gini \times \Delta Financial outreach$	∆Financial usage	$\begin{array}{l} \Delta Gini \times \Delta Financial \\ usage \end{array}$	ΔFindex	$\Delta Gini \times \Delta findes$	x ΔFindex outreach	$\Delta Gini \times \Delta Financial outreach$	∆Financial usage	$\begin{array}{l} \Delta Gini \times \Delta Financial \\ usage \end{array}$	ΔFindex	$\Delta Gini \times \Delta findex$	ΔFindex outreach	$\Delta Gini \times \Delta Financial outreach$	∆Financial usage	$\begin{array}{l} \Delta Gini \times \Delta Financial \\ usage \end{array}$
Mobile subscription lagged	2 0.014***	-0.000	0.009***	0.000	0.020***	-0.000	0.014***	-0.000	0.009***	0.000	0.020***	-0.000	0.014***	-0.000	0.009***	0.000	0.020***	-0.000	0.014***	-0.000	0.009***	0.000	0.020***	-0.000
periods	(0.004)	(0.000)	(0.002)	(0.000)	(0.006)	(0.000)	(0.004)	(0.000)	(0.002)	(0.000)	(0.006)	(0.000)	(0.004)	(0.000)	(0.002)	(0.000)	(0.006)	(0.000)	(0.004)	(0.000)	(0.002)	(0.000)	(0.006)	(0.000)
Origin of																								
legal																								
French	-0.004*	-0.000	-0.007*	-0.000**	-0.005*	0.000	-0.004*	-0.000	-0.002*	-0.000**	-0.005*	0.000	-0.004*	-0.000	-0.002*	-0.000**	-0.005*	0.000	-0.004*	-0.000	-0.002*	-0.000**	-0.005*	0.000
	(0.002)	(0.000)	(0.001)	(0.000)	(0.003)	(0.000)	(0.002)	(0.000)	(0.001)	(0.000)	(0.003)	(0.000)	(0.002)	(0.000)	(0.001)	(0.000)	(0.003)	(0.000)	(0.002)	(0.000)	(0.001)	(0.000)	(0.003)	(0.000)
Socialist	-0.002	0.000	-0.002	-0.000	-0.002	0.000***	-0.002	0.000	-0.002	-0.000	-0.002	0.000***	-0.002	0.000	-0.002	-0.000	-0.002	0.000***	-0.002	0.000	-0.002	-0.000	-0.002	0.000***
	(0.002)	(0.000)	(0.001)	(0.000)	(0.003)	(0.000)	(0.002)	(0.000)	(0.001)	(0.000)	(0.003)	(0.000)	(0.002)	(0.000)	(0.001)	(0.000)	(0.003)	(0.000)	(0.002)	(0.000)	(0.001)	(0.000)	(0.003)	(0.000)
German	-0.001	-0.000	0.000	-0.000	-0.003	-0.000	-0.001	-0.000	0.000	-0.000	-0.003	-0.000	-0.001	-0.000	0.000	-0.000	-0.003	-0.000	-0.001	-0.000	0.000	-0.000	-0.003	-0.000
	(0.008)	(0.000)	(0.005)	(0.000)	(0.012)	(0.000)	(0.008)	(0.000)	(0.005)	(0.000)	(0.012)	(0.000)	(0.008)	(0.000)	(0.005)	(0.000)	(0.012)	(0.000)	(0.008)	(0.000)	(0.005)	(0.000)	(0.012)	(0.000)
Colony																								
Dutch	0.009***	0.000***	0.009***	0.000***	0.009***	-0.000	0.009***	0.000***	0.009***	0.000***	0.009***	-0.000	0.009***	0.000***	0.009***	0.000***	0.009***	-0.000	0.009***	0.000***	0.009***	0.000***	0.009***	-0.000
colony	(0.002)	(0.000)	(0.001)	(0.000)	(0.003)	(0.000)	(0.002)	(0.000)	(0.001)	(0.000)	(0.003)	(0.000)	(0.002)	(0.000)	(0.001)	(0.000)	(0.003)	(0.000)	(0.002)	(0.000)	(0.001)	(0.000)	(0.003)	(0.000)
Spanish	0.001	-0.000	0.001	-0.000	-0.000	-0.000**	0.001	-0.000	0.001	-0.000	-0.000	-0.000**	0.001	-0.000	0.001	-0.000	-0.000	-0.000**	0.001	-0.000	0.001	-0.000	-0.000	-0.000**
colony	(0.002)	(0.000)	(0.002)	(0.000)	(0.003)	(0.000)	(0.002)	(0.000)	(0.002)	(0.000)	(0.003)	(0.000)	(0.002)	(0.000)	(0.002)	(0.000)	(0.003)	(0.000)	(0.002)	(0.000)	(0.002)	(0.000)	(0.003)	(0.000)
USA colony	-0.004*	-0.000	0.000	-0.000**	-0.009***	-0.000	-0.004*	-0.000	0.000	-0.000**	-0.009***	-0.000	-0.004*	-0.000	0.000	-0.000**	-0.009***	-0.000	-0.004*	-0.000	0.000	-0.000**	-0.009***	-0.000
ACini	(0.002)	(0.000)	(0.001)	(0.000)	(0.003)	(0.000)	(0.002)	(0.000)	(0.001)	(0.000)	(0.003)	(0.000)	(0.002)	(0.000)	(0.001)	(0.000)	(0.003)	(0.000)	(0.002)	(0.000)	(0.001)	(0.000)	(0.003)	(0.000)
Admi	(0.010)	(0.001)	(0.018)	(0.004	(0.024)	(0.003)	(0.010)	(0.001)	(0.018)	(0.004	(0.024)	(0.004	(0.010)	(0.004	(0.012)	(0.001)	(0.024)	(0.003)	(0.010)	(0.004	(0.018)	(0.004)	(0.024)	(0.003)
CDP	0.060***	0.001)	0.048***	-0.000	0.074***	0.002)	0.060***	0.0001)	0.048***	-0.000	0.074***	0.0002)	0.060***	0.000	0.048***	-0.000	0.074***	0.002)	0.050***	0.000	0.048***	-0.000	0.074***	0.002)
growth	(0.015)	(0.000)	(0.012)	(0.000)	(0.022)	(0.000)	(0.015)	(0.000)	(0.012)	(0.000)	(0.022)	(0.000)	(0.015)	(0.000)	(0.012)	(0.000)	(0.022)	(0.000)	(0.015)	(0.000)	(0.012)	(0.000)	(0.022)	(0.000)
rate																								
Constant	-0.008**	0.000	-0.007***	-0.000	-0.008*	0.000	-0.008**	0.000	-0.007***	-0.000	-0.008*	0.000	-0.008**	0.000	-0.007***	-0.000	-0.008*	0.000	-0.008**	0.000	-0.007***	-0.000	-0.008*	0.000
	(0.003)	(0.000)	(0.002)	(0.000)	(0.005)	(0.000)	(0.003)	(0.000)	(0.002)	(0.000)	(0.005)	(0.000)	(0.003)	(0.000)	(0.002)	(0.000)	(0.005)	(0.000)	(0.003)	(0.000)	(0.002)	(0.000)	(0.005)	(0.000)
Year	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
fixed																								
effects																								
Observations	701	701	701	701	701	701	701	701	701	701	701	701	701	701	701	701	701	701	701	701	701	701	701	701
R-squared	0.143	0.221	0.131	0.262	0.120	0.098	0.143	0.221	0.131	0.262	0.120	0.098	0.143	0.221	0.131	0.262	0.120	0.098	0.143	0.221	0.131	0.262	0.120	0.098

Note: Δ indicates change from one period to the next. Robust standard errors, clustered at the country level in parentheses. Significance level *** p < 0.01, ** p < 0.05, * p < 0.1. Respective second-stage IV random effects regression in Table 5. Source: World Bank's PovcalNet database and IMF. Coverage: 2004–2018 period.

-

Growth-redistribution decomposition of those living under \$1.90 dollars a day in low- and low-middle income countries: the role of financial inclusion. Standardised coefficients of second-stage IV random effects specification.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)
Dependent variable:	Δ Headco	unt		Δ Poverty	gap		Δ Poverty	gap square	ed	Δ Watts		
ΔGini	0.749***	0.817***	0.518**	0.769***	0.845***	0.564***	0.731***	0.814***	0.545***	0.760***	0.852***	0.564***
	(0.259)	(0.238)	(0.228)	(0.226)	(0.214)	(0.206)	(0.216)	(0.203)	(0.195)	(0.228)	(0.216)	(0.204)
GDP growth rate	-0.039	-0.062	-0.045	-0.071	-0.097	-0.072	-0.081	-0.107	-0.082	-0.080	-0.105	-0.082
AFinancial inclusion	(0.049) -0.025	(0.052)	(0.048)	(0.057) 0.180	(0.061)	(0.059)	(0.064) 0.253	(0.067)	(0.065)	(0.064) 0.248	(0.067)	(0.066)
	(0.181)			(0.218)			(0.250)			(0.253)		
Δ Gini × Δ Financial inclusion	-0.854			-0.788			-0.668			-0.667		
	(0.559)			(0.488)			(0.479)			(0.496)		
Δ Financial inclusion outreach		0.165			0.417			0.487			0.482	
		(0.251)			(0.284)			(0.318)			(0.321)	
Δ Gini $\times \Delta$ Financial inclusion outreach		-1.022**			-0.997**			-0.919**			-0.949**	
		(0.398)	0.000		(0.394)	0.004		(0.372)	0.000		(0.370)	0.000
Δ Financial inclusion usage			0.002			0.234			0.332			0.338
A Cini A Financial inclusion was as			(0.258)			(0.319)			(0.365)			(0.375)
			0.252			(0.850)			(0.255)			(0.002)
Constant	0 173***	0 188***	0.172**	0 177***	0 224***	(0.850)	0 162***	0 218**	0.132	0 1 5 9***	0 214**	(0.902)
constant	(0.048)	(0.055)	(0.080)	(0.059)	(0.224)	(0.093)	(0.060)	(0.086)	(0.090)	(0.060)	(0.086)	(0.092)
Year fixed effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Observations	701	701	701	701	701	701	701	701	701	701	701	701
Number of countries	66	66	66	66	66	66	66	66	66	66	66	66

Note: Δ indicates change from one period to the next. Robust standard errors, clustered at the country level in parentheses. Significance level *** p < 0.01, ** p < 0.05, * p < 0.1. Respective first-stage IV regression (of non-standardised coefficients) in Table A4. Source: World Bank's PovcalNet database and IMF. Coverage: 2004–2018 period.

Table A6

Summary statistics of poverty under \$3.20 and \$5.50 dollars a day over 2004-2018.

	All countries		Low- and low income coun	ver-middle- tries	Upper-middle high-income	e and countries
	Mean	St dev.	Mean	St dev.	Mean	St dev.
Headcount under \$3.2 a day	0.282	0.297	0.476	0.278	0.065	0.107
Poverty gap under \$3.2 a day	0.125	0.163	0.217	0.177	0.024	0.043
Poverty gap squared under \$3.2 a day	0.073	0.109	0.128	0.126	0.013	0.024
Watts index under \$3.2 a day	0.202	0.294	0.355	0.334	0.033	0.063
Headcount under \$5.5 a day	0.440	0.346	0.697	0.241	0.154	0.183
Poverty gap under \$5.5 a day	0.226	0.226	0.377	0.207	0.059	0.085
Poverty gap squared under \$5.5 a day	0.145	0.168	0.247	0.172	0.032	0.052
Watts index under \$5.5 a day	0.397	0.454	0.674	0.460	0.088	0.137
Number of countries	121		79		66	
Total number of observations	1679		884		795	

Source: Own estimates using PovcalNet.

Poverty growth-redistribution decomposition of those living under \$3.20 dollars a day.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)
_	All countries				Low- and low	wer-middle	-income countrie	es	Upper-midle	and high-i	ncome countries	
	ΔHeadcount	Δ Poverty gap	Δ Poverty gap squared	∆Watts	∆Headcount	Δ Poverty gap	Δ Poverty gap squared	∆Watts	∆Headcount	∆Poverty gap	Δ Poverty gap squared	∆Watts
∆Gini	0.565***	0.463***	0.377***	0.972***	0.422	0.534**	0.473***	1.298***	0.742***	0.358***	0.240***	0.511***
	(0.174)	(0.120)	(0.093)	(0.261)	(0.263)	(0.205)	(0.158)	(0.435)	(0.109)	(0.063)	(0.053)	(0.085)
GDP growth	-0.030	-0.011	-0.007	-0.017	0.004	0.004	-0.000	-0.001	-0.066***	-0.025^{*}	-0.014	-0.028
rate	(0.026)	(0.016)	(0.013)	(0.034)	(0.046)	(0.027)	(0.021)	(0.055)	(0.023)	(0.014)	(0.010)	(0.025)
Constant	-0.004^{*}	-0.003^{*}	-0.002	-0.005^{*}	-0.006**	-0.004^{*}	-0.003*	-0.007	-0.002	-0.001	-0.001	-0.003
	(0.002)	(0.001)	(0.001)	(0.003)	(0.003)	(0.002)	(0.002)	(0.004)	(0.002)	(0.002)	(0.001)	(0.003)
Year fixed effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Observations	1.679	1.679	1.679	1.679	884	884	884	884	795	795	795	795
Adjusted R ²	0.107	0.155	0.162	0.152	0.0477	0.139	0.164	0.170	0.351	0.340	0.330	0.284
Number of countries	121	121	121	121	79	79	79	79	66	66	66	66

Note: Panel fixed effects model. Δ indicates change from one period to the next. Robust standard errors, clustered at the country level in parentheses. Significance level *** p < 0.01, ** p < 0.05, * p < 0.1.

Source: World Bank's PovcalNet database and IMF. Coverage: 2004–2018 period.

Table A8

Poverty growth-redistribution decomposition of those living under \$5.50 dollars a day.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)
_	All countries	;			Low- and lov	wer-middle	e-income countri	ies	Upper-midle	and high-	income countrie	s
	∆Headcount	ΔPoverty gap	∆Poverty gap squared	∆Watts	∆Headcount	∆Poverty gap	Δ Poverty gap squared	∆Watts	∆Headcount	Δ Poverty gap	∆Poverty gap squared	∆Watts
∆Gini	0.453**	0.485***	0.446***	1.238***	0.062	0.401**	0.468**	1.425***	0.959***	0.584***	0.407***	0.954***
	(0.187)	(0.131)	(0.113)	(0.315)	(0.145)	(0.199)	(0.190)	(0.537)	(0.200)	(0.095)	(0.069)	(0.141)
GDP growth	-0.086***	-0.031	-0.017	-0.047	-0.026	-0.003	0.001	-0.007	-0.152***	-0.061***	-0.034**	-0.084^{**}
rate	(0.028)	(0.019)	(0.016)	(0.044)	(0.049)	(0.033)	(0.026)	(0.073)	(0.034)	(0.019)	(0.014)	(0.036)
Constant	-0.003	-0.003^{*}	-0.003*	-0.007^{*}	-0.005^{*}	-0.005^{**}	-0.004^{**}	-0.010^{*}	-0.003	-0.002	-0.001	-0.005
	(0.002)	(0.002)	(0.001)	(0.004)	(0.003)	(0.002)	(0.002)	(0.006)	(0.003)	(0.002)	(0.002)	(0.004)
Year fixed effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Observations	1,679	1,679	1,679	1,679	884	884	884	884	795	795	795	795
Adjusted R ²	0.0675	0.135	0.157	0.154	-0.00320	0.0739	0.122	0.139	0.285	0.357	0.364	0.322
Number of countries	121	121	121	121	79	79	79	79	66	66	66	66

Note: Panel fixed effects model. Δ indicates change from one period to the next. Robust standard errors, clustered at the country level in parentheses. Significance level *** p < 0.01, ** p < 0.05, * p < 0.1.

Source: World Bank's PovcalNet database and IMF. Coverage: 2004–2018 period.

Table A9 Growth-redistribution decomposition of those living under \$3.20 dollars a day in low- and low-middle income countries. First-stage IV specification.

	(1)	(2)	(3)	(4)	(5)	(6)	(7) (8	i) ((9)	(10)	(11)	(12)	(13)	(14)	(15)	(16)	(17)	(18)	(19)	(20)	(21)	(22)	(23)	(24)
First-stage model for:	ΔHeadcor	int					ΔPoverty ga	p					ΔPowert	gap squared					ΔWatts					
Dependent variable:	ΔFindex	$\Delta Gini \times \Delta findex$	ΔFindex	$\Delta Gini \times \Delta Financial$	∆Financial	$\Delta Gini \times \Delta Financial$	ΔFindex Δ	Gini × Δfindex	ΔFindex	$\Delta Gini \times \Delta Financial$	ΔFinancial	$\Delta Gini \times \Delta Financial$	ΔFindex	$\Delta Gini \times \Delta findex$	κ ΔFindex	$\Delta Gini \times \Delta Financial$	ΔFinancial	$\Delta Gini \times \Delta Financial$	ΔFindex	$\Delta Gini \times \Delta findex$	ΔFindex	$\Delta Gini \times \Delta Financial$	ΔFinancial	$\Delta Gini \times \Delta Financial$
			outreach	outreach	usage	usage			outreach	outreach	usage	usage			outreach	outreach	usage	usage			outreach	outreach	usage	usage
Mobile subscription lagged	2 0.014***	-0.000	0.009***	0.000	0.020***	-0.000	0.014*** -0	0.000 0	0.009***	0.000	0.020***	-0.000	0.014***	-0.000	0.009***	0.000	0.020***	-0.000	0.014***	-0.000	0.009***	0.000	0.020***	-0.000
periods	(0.004)	(0.000)	(0.002)	(0.000)	(0.006)	(0.000)	(0.004) (0	.000) ((0.002)	(0.000)	(0.006)	(0.000)	(0.004)	(0.000)	(0.002)	(0.000)	(0.006)	(0.000)	(0.004)	(0.000)	(0.002)	(0.000)	(0.006)	(0.000)
Origin of legal system																								
French	-0.003*	-0.000	-0.002	-0.000**	-0.005	0.000	-0.003* -0	0.000	-0.002	-0.000**	-0.005	0.000	-0.003*	-0.000	-0.002	-0.000**	-0.005	0.000	-0.003*	-0.000	-0.002	-0.000**	-0.005	0.000
	(0.002)	(0.000)	(0.001)	(0.000)	(0.003)	(0.000)	(0.002) (0	.000) ((0.001)	(0.000)	(0.003)	(0.000)	(0.002)	(0.000)	(0.001)	(0.000)	(0.003)	(0.000)	(0.002)	(0.000)	(0.001)	(0.000)	(0.003)	(0.000)
Socialist	-0.002	0.000	-0.002	-0.000	-0.002	0.000***	-0.002 0.0	000 -	-0.002	-0.000	-0.002	0.000***	-0.002	0.000	-0.002	-0.000	-0.002	0.000***	-0.002	0.000	-0.002	-0.000	-0.002	0.000***
	(0.002)	(0.000)	(0.001)	(0.000)	(0.003)	(0.000)	(0.002) (0	.000) ((0.001)	(0.000)	(0.003)	(0.000)	(0.002)	(0.000)	(0.001)	(0.000)	(0.003)	(0.000)	(0.002)	(0.000)	(0.001)	(0.000)	(0.003)	(0.000)
German	-0.001	-0.000	0.000	-0.000	-0.003	-0.000	-0.001 -0	0.000 (0.000	-0.000	-0.003	-0.000	-0.001	-0.000	0.000	-0.000	-0.003	-0.000	-0.001	-0.000	0.000	-0.000	-0.003	-0.000
	(0.008)	(0.000)	(0.005)	(0.000)	(0.012)	(0.000)	(0.008) (0	.000) ((0.005)	(0.000)	(0.012)	(0.000)	(0.008)	(0.000)	(0.005)	(0.000)	(0.012)	(0.000)	(0.008)	(0.000)	(0.005)	(0.000)	(0.012)	(0.000)
Colony																								
Dutch colony	0.009***	0.000***	0.009***	0.000***	0.009***	-0.000	0.009*** 0.0	000***	0.009***	0.000***	0.009***	-0.000	0.009***	0.000***	0.009***	0.000***	0.009***	-0.000	0.009***	0.000***	0.009***	0.000***	0.009***	-0.000
	(0.002)	(0.000)	(0.001)	(0.000)	(0.003)	(0.000)	(0.002) (0	.000) ((0.001)	(0.000)	(0.003)	(0.000)	(0.002)	(0.000)	(0.001)	(0.000)	(0.003)	(0.000)	(0.002)	(0.000)	(0.001)	(0.000)	(0.003)	(0.000)
Spanish colony	0.000	-0.000	0.001	-0.000	-0.001	-0.000**	0.000 -0	0.000 0	0.001	-0.000	-0.001	-0.000**	0.000	-0.000	0.001	-0.000	-0.001	-0.000**	0.000	-0.000	0.001	-0.000	-0.001	-0.000**
	(0.002)	(0.000)	(0.002)	(0.000)	(0.003)	(0.000)	(0.002) (0	.000) ((0.002)	(0.000)	(0.003)	(0.000)	(0.002)	(0.000)	(0.002)	(0.000)	(0.003)	(0.000)	(0.002)	(0.000)	(0.002)	(0.000)	(0.003)	(0.000)
USA colony	-0.004*	-0.000	0.000	-0.000**	-0.009***	-0.000	-0.004* -0	0.000 (0.000	-0.000**	-0.009***	-0.000	-0.004*	-0.000	0.000	-0.000**	-0.009***	-0.000	-0.004^{*}	-0.000	0.000	-0.000**	-0.009***	-0.000
	(0.002)	(0.000)	(0.001)	(0.000)	(0.003)	(0.000)	(0.002) (0	.000) ((0.001)	(0.000)	(0.003)	(0.000)	(0.002)	(0.000)	(0.001)	(0.000)	(0.003)	(0.000)	(0.002)	(0.000)	(0.001)	(0.000)	(0.003)	(0.000)
ΔGini	-0.006	0.004***	-0.002	0.004***	-0.011	0.004**	-0.006 0.0	004***	-0.002	0.004***	-0.011	0.004**	-0.006	0.004***	-0.002	0.004***	-0.011	0.004**	-0.006	0.004***	-0.002	0.004***	-0.011	0.004**
	(0.019)	(0.001)	(0.018)	(0.001)	(0.035)	(0.002)	(0.019) (0	.001) ((0.018)	(0.001)	(0.035)	(0.002)	(0.019)	(0.001)	(0.018)	(0.001)	(0.035)	(0.002)	(0.019)	(0.001)	(0.018)	(0.001)	(0.035)	(0.002)
GDP growth rate	0.062***	-0.000	0.048***	-0.000	0.078***	0.000	0.062*** -0	0.000 0	0.048***	-0.000	0.078***	0.000	0.062***	-0.000	0.048***	-0.000	0.078***	0.000	0.062***	-0.000	0.048***	-0.000	0.078***	0.000
	(0.016)	(0.000)	(0.014)	(0.000)	(0.023)	(0.000)	(0.016) (0	.000) ((0.014)	(0.000)	(0.023)	(0.000)	(0.016)	(0.000)	(0.014)	(0.000)	(0.023)	(0.000)	(0.016)	(0.000)	(0.014)	(0.000)	(0.023)	(0.000)
Constant	0.001	-0.000	0.001	-0.000	0.001	-0.000	0.001 -0	0.000 0	0.001	-0.000	0.001	-0.000	0.001	-0.000	0.001	-0.000	0.001	-0.000	0.001	-0.000	0.001	-0.000	0.001	-0.000
	(0.002)	(0.000)	(0.001)	(0.000)	(0.003)	(0.000)	(0.002) (0	.000) ((0.001)	(0.000)	(0.003)	(0.000)	(0.002)	(0.000)	(0.001)	(0.000)	(0.003)	(0.000)	(0.002)	(0.000)	(0.001)	(0.000)	(0.003)	(0.000)
Year fixed effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes Ye	15 1	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Observations	687	687	687	687	687	687	687 68	87 (687	687	687	687	687	687	687	687	687	687	687	687	687	687	687	687
R-squared	0.140	0.221	0.129	0.263	0.119	0.098	0.140 0.3	221 (0.129	0.263	0.119	0.098	0.140	0.221	0.129	0.263	0.119	0.098	0.140	0.221	0.129	0.263	0.119	0.098

Note: Δ indicates change from one period to the next. Robust standard errors, clustered at the country level in parentheses. Significance level *** p < 0.01, ** p < 0.05, * p < 0.1. Respective second-stage IV random effects regression in Table A10. Source: World Bank's PovcalNet database and IMF. Coverage: 2004–2018 period.

Growth-redistribution decomposition of those living under \$3.20 dollars a day in low- and low-middle income countries: the role of financial inclusion. Second-stage IV random effects specification.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)
Dependent variable:	∆Headco	unt		ΔPoverty	y gap		ΔPoverty	/ gap square	d	Δ Watts		
ΔGini	0.790	0.890**	0.465	0.889***	0.974***	0.614**	0.740***	0.813***	0.530***	1.985***	2.207***	1.421***
	(0.503)	(0.418)	(0.404)	(0.306)	(0.275)	(0.265)	(0.226)	(0.208)	(0.201)	(0.624)	(0.572)	(0.547)
GDP growth rate	0.004	0.000	-0.011	-0.022	-0.033	-0.027	-0.027	-0.037	-0.029	-0.073	-0.099	-0.082
	(0.049)	(0.050)	(0.047)	(0.031)	(0.032)	(0.031)	(0.027)	(0.027)	(0.027)	(0.074)	(0.076)	(0.076)
Δ Financial inclusion	-0.693**			-0.044			0.104			0.269		
	(0.344)			(0.220)			(0.189)			(0.539)		
Δ Gini $\times \Delta$ Financial inclusion	-36.988			-54.303			-43.909			-107.402		
	(89.220)			(40.761)			(28.240)			(80.245)		
Δ Financial inclusion outreach		-0.809			0.166			0.335			0.858	
		(0.499)			(0.331)			(0.280)			(0.788)	
Δ Gini $ imes \Delta$ Financial inclusion outreacl	1	-59.370			-72.985***			-59.948***			-157.207***	
		(55.898)			(27.549)			(21.890)			(56.353)	
Δ Financial inclusion usage			-0.356			-0.001			0.090			0.260
			(0.273)			(0.185)			(0.161)			(0.465)
Δ Gini $ imes \Delta$ Financial inclusion usage			46.409			14.346			8.244			33.711
			(62.187)			(35.566)			(27.955)			(78.086)
Constant	-0.007	-0.007	-0.006	-0.002	-0.003	-0.001	-0.001	-0.001	0.001	-0.002	-0.004	0.001
	(0.006)	(0.006)	(0.006)	(0.003)	(0.003)	(0.003)	(0.002)	(0.002)	(0.002)	(0.004)	(0.005)	(0.005)
Year fixed effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Observations	687	687	687	687	687	687	687	687	687	687	687	687
Number of countries	66	66	66	66	66	66	66	66	66	66	66	66
Test of overidentifying restrictions:												
Sargan-Hansen statistic Chi-so	4310	4 2 3 4	3 059	5 202	3 805	4 064	6 1 5 2	3 840	5 601	5 864	3 837	4 872
Sargan-Hansen statistic P-value	0.506	0.516	0.691	0.392	0.578	0.540	0.152	0.573	0347	0 320	0.573	0.432
Sargan mansen statistic i -value	0.500	0.510	0.031	0.552	0.570	0.540	0.232	0.575	0.547	0.520	0.575	0.752
Endogenity test:												
Chi-sq	23.580	24.020	25.680	21.620	18.450	21.530	21.200	17.070	19.150	23.460	19.900	21.570
P-value	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000

Note: Δ indicates change from one period to the next. Robust standard errors, clustered at the country level in parentheses. Significance level *** p < 0.01, ** p < 0.05, * p < 0.1. Respective first-stage IV regression in Table A11. Source: World Bank's PovcalNet database and IMF. Coverage: 2004–2018 period.

R.
Gutiérrez-Romero
and
₫.
Ahamed

Table A11	
Growth-redistribution decomposition of those living under \$5.50 dollars a day in low- and low-middle income countries. First-stage IV speci	fication.

		-			-										-									
_	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)	(13)	(14)	(15)	(16)	(17)	(18)	(19)	(20)	(21)	(22)	(23)	(24)
First-stage model for:	ΔHeadc	ount					ΔPoverty	gap					ΔPovert	y gap squared					ΔWatts					
Dependent variable:	ΔFinder	$\Delta Gini \times \Delta finde$	ΔFindex outreach	$\begin{array}{l} \Delta Gini \times \Delta Financial \\ outreach \end{array}$	ΔFinancial usage	$\begin{array}{l} \Delta Gini \times \Delta Financial \\ usage \end{array}$	ΔFindex	$\Delta Gini \times \Delta findex$	ΔFindex outreach	$\Delta Gini \times \Delta Financial outreach$	ΔFinancial usage	$\begin{array}{l} \Delta Gini \times \Delta Financial \\ usage \end{array}$	ΔFindex	$\Delta Gini \times \Delta findex$	ΔFindex outreach	$\Delta Gini \times \Delta Financial outreach$	ΔFinancial usage	$\begin{array}{l} \Delta Gini \times \Delta Financial \\ usage \end{array}$	ΔFindex	$\Delta Gini \times \Delta findex$	ΔFindex outreach	$\Delta Gini \times \Delta Financial outreach$	ΔFinancial usage	$\begin{array}{l} \Delta Gini \times \Delta Financial \\ usage \end{array}$
Mobile subscription lagged periods Origin of legal	2 0.014*** (0.004)	-0.000 (0.000)	0.009*** (0.002)	0.000 (0.000)	0.020*** (0.006)	-0.000 (0.000)	0.014*** (0.004)	-0.000 (0.000)	0.009*** (0.002)	0.000 (0.000)	0.020*** (0.006)	-0.000 (0.000)	0.014*** (0.004)	-0.000 (0.000)	0.009*** (0.002)	0.000 (0.000)	0.020*** (0.006)	-0.000 (0.000)	0.014*** (0.004)	-0.000 (0.000)	0.009*** (0.002)	0.000 (0.000)	0.020*** (0.006)	-0.000 (0.000)
system French	-0.003*	-0.000	-0.002	-0.000**	-0.005	0.000	-0.003*	-0.000	-0.002	-0.000**	-0.005	0.000	-0.003*	-0.000	-0.002	-0.000**	-0.005	0.000	-0.003*	-0.000	-0.002	-0.000**	-0.005	0.000
Socialist	(0.002) -0.002 (0.002)	(0.000) (0.000)	(0.001) 0.002 (0.001)	-0.000 (0.000)	(0.003) -0.002 (0.003)	0.000*** (0.000)	-0.002) (0.002)	0.000 (0.000)	-0.002 (0.001)	-0.000 (0.000)	-0.002 (0.003)	0.000*** (0.000)	-0.002 (0.002)	(0.000) (0.000)	-0.002	-0.000 (0.000)	-0.002 (0.003)	0.000*** (0.000)	-0.002 (0.002)	0.000 (0.000)	(0.001) 0.002 (0.001)	-0.000 (0.000)	-0.002 (0.003)	0.000*** (0.000)
German	-0.001 (0.008)	-0.000 (0.000)	0.000 (0.005)	-0.000 (0.000)	-0.003 (0.012)	-0.000 (0.000)	-0.001 (0.008)	-0.000 (0.000)	0.000 (0.005)	-0.000 (0.000)	-0.003 (0.012)	-0.000 (0.000)	-0.001 (0.008)	-0.000 (0.000)	0.000 (0.005)	-0.000 (0.000)	-0.003 (0.012)	-0.000 (0.000)	-0.001 (0.008)	-0.000 (0.000)	0.000 (0.005)	-0.000 (0.000)	-0.003 (0.012)	-0.000 (0.000)
Colony	0.000111	0.000111	0.000111	0.000111	0.0003111	0.000	0.000111	0.000111	0.000181	0.000111	0.000111	0.000	0.000111	0.000111	0.000111	0.000111	0.000111	0.000	0.000444	0.000111	0.000111	0.000111	0.000111	0.000
colony	(0.002)	(0.000)	(0.001)	(0.000)	(0.003)	-0.000	(0.002)	(0.000)	(0.001)	(0.000)	(0.003)	-0.000	(0.007)	(0.000)	(0.001)	(0.000)	(0.003)	-0.000	(0.002)	(0.000)	(0.001)	(0.000)	(0.003)	-0.000
Snanish	0.000	-0.000	0.001	-0.000	-0.001	-0.000**	0.000	-0.000	0.001	-0.000	-0.001	-0.000**	0.000	-0.000	0.001	-0.000	-0.001	-0.000**	0.000	-0.000	0.001	-0.000	-0.001	-0.000**
colony	(0.002)	(0.000)	(0.002)	(0.000)	(0.003)	(0.000)	(0.002)	(0.000)	(0.002)	(0.000)	(0.003)	(0.000)	(0.002)	(0.000)	(0.002)	(0.000)	(0.003)	(0.000)	(0.002)	(0.000)	(0.002)	(0.000)	(0.003)	(0.000)
USA colony	-0.004*	-0.000	0.000	-0.000**	-0.009***	-0.000	-0.004*	-0.000	0.000	-0.000**	-0.009***	-0.000	-0.004*	-0.000	0.000	-0.000**	-0.009***	-0.000	-0.004*	-0.000	0.000	-0.000**	-0.009***	-0.000
	(0.002)	(0.000)	(0.001)	(0.000)	(0.003)	(0.000)	(0.002)	(0.000)	(0.001)	(0.000)	(0.003)	(0.000)	(0.002)	(0.000)	(0.001)	(0.000)	(0.003)	(0.000)	(0.002)	(0.000)	(0.001)	(0.000)	(0.003)	(0.000)
ΔGini	-0.006	0.004***	-0.002	0.004***	-0.011	0.004**	-0.006	0.004***	-0.002	0.004***	-0.011	0.004**	-0.006	0.004***	-0.002	0.004***	-0.011	0.004**	-0.006	0.004***	-0.002	0.004***	-0.011	0.004**
	(0.019)	(0.001)	(0.018)	(0.001)	(0.035)	(0.002)	(0.019)	(0.001)	(0.018)	(0.001)	(0.035)	(0.002)	(0.019)	(0.001)	(0.018)	(0.001)	(0.035)	(0.002)	(0.019)	(0.001)	(0.018)	(0.001)	(0.035)	(0.002)
GDP	0.062***	-0.000	0.048***	-0.000	0.078***	0.000	0.062***	-0.000	0.048***	-0.000	0.078***	0.000	0.062***	-0.000	0.048***	-0.000	0.078***	0.000	0.062***	-0.000	0.048***	-0.000	0.078***	0.000
growth	(0.016)	(0.000)	(0.014)	(0.000)	(0.023)	(0.000)	(0.016)	(0.000)	(0.014)	(0.000)	(0.023)	(0.000)	(0.016)	(0.000)	(0.014)	(0.000)	(0.023)	(0.000)	(0.016)	(0.000)	(0.014)	(0.000)	(0.023)	(0.000)
rate																								
Constant	0.001	-0.000	0.001	-0.000	0.001	-0.000	0.001	-0.000	0.001	-0.000	0.001	-0.000	0.001	-0.000	0.001	-0.000	0.001	-0.000	0.001	-0.000	0.001	-0.000	0.001	-0.000
	(0.002)	(0.000)	(0.001)	(0.000)	(0.003)	(0.000)	(0.002)	(0.000)	(0.001)	(0.000)	(0.003)	(0.000)	(0.002)	(0.000)	(0.001)	(0.000)	(0.003)	(0.000)	(0.002)	(0.000)	(0.001)	(0.000)	(0.003)	(0.000)
Year fixed effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Observations	687	687	687	687	687	687	687	687	687	687	687	687	687	687	687	687	687	687	687	687	687	687	687	687
R-squared	0.140	0.221	0.129	0.263	0.119	0.098	0.140	0.221	0.129	0.263	0.119	0.098	0.140	0.221	0.129	0.263	0.119	0.098	0.140	0.221	0.129	0.263	0.119	0.098

Note: Δ indicates change from one period to the next. Robust standard errors, clustered at the country level in parentheses. Significance level *** p < 0.01, ** p < 0.05, * p < 0.1. Respective second-stage IV random effects regression in Table A10. Source: World Bank's PovcalNet database and IMF. Coverage: 2004–2018 period.

Growth-redistribution decomposition of those living under \$5.50 dollars a day in low- and low-middle income countries: the role of financial inclusion. Second-stage IV random effects specification.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)
Dependent variable:	ΔHeadcou	int		ΔPoverty	/ gap		Δ Poverty	/ gap square	ed	Δ Watts		
ΔGini	-0.214	-0.045	-0.209	0.614*	0.707**	0.413	0.754^{**}	0.837***	0.522^{**}	2.139***	2.414***	1.517**
GDP growth rate	(0.492) 0.024 (0.046)	(0.390) 0.050 (0.054)	(0.394) -0.004 (0.044)	(0.337) -0.006 (0.025)	(0.303) -0.008 (0.027)	(0.293) -0.016 (0.024)	(0.294) -0.017 (0.020)	(0.239) -0.025 (0.021)	(0.249) -0.024 (0.020)	(0.820) -0.065 (0.000)	(0.730) -0.085 (0.001)	(0.094) -0.082
ΔFinancial inclusion	(0.040) -0.974^{***} (0.378)	(0.054)	(0.044)	(0.033) -0.415^{*} (0.252)	(0.037)	(0.034)	(0.030) -0.139 (0.213)	(0.051)	(0.030)	(0.030) -0.227 (0.650)	(0.091)	(0.089)
$\Delta Gini \times \Delta Financial \ inclusion$	(0.970) 100.977 (106 569)			(0.232) -15.344 (60.254)			-38.608			-90.615 (118.456)		
Δ Financial inclusion outreach	(100.505)	-1.640^{**}		(00.234)	-0.477		(42.233)	-0.019		(110.450)	0.140	
$\Delta Gini imes \Delta Financial inclusion outreact$	1	(0.040) 58.729 (76.719)			-37.051			(0.510) -57.178^{**} (26.807)			(0.540) -153.547^{**} (73.520)	
Δ Financial inclusion usage		(70.713)	-0.434		(38.107)	-0.209		(20.807)	-0.059		(75.525)	-0.026
$\Delta Gini \times \Delta Financial \ inclusion \ usage$			(0.283) 102.559 (70,736)			(0.201) 36.256 (44.626)			(0.170) 19.447 (34.626)			(0.344) 66.128 (99.243)
Constant	-0.009	-0.009	-0.010 (0.007)	-0.005 (0.004)	-0.005 (0.004)	-0.004 (0.004)	-0.002	-0.003	-0.001 (0.003)	-0.006	-0.008	-0.003 (0.007)
Year fixed effects Observations Number of countries	Yes 687 66	Yes 687 66	Yes 687 66	Yes 687 66	Yes 687 66	Yes 687 66	Yes 687 66	Yes 687 66	Yes 687 66	Yes 687 66	Yes 687 66	Yes 687 66
Test of overidentifying restrictions: Sargan-Hansen statistic Chi-sq Sargan-Hansen statistic P-value	7.028 0.219	6.718 0.242	6.973 0.2227	3.99 0.5509	2.873 0.7195	2.638 0.756	4.399 0.494	3.012 0.6981	2.927 0.7113	4.11 0.5337	3.048 0.6926	2.826 0.7269
Endogenity test: Chi-sq P-value	25.320 0.000	27.720 0.000	26.970 0.000	27.150 0.000	26.840 0.000	29.040 0.000	26.170 0.000	23.950 0.000	26.870 0.000	30.390 0.000	28.150 0.000	30.450 0.000

Note: Δ indicates change from one period to the next. Robust standard errors, clustered at the country level in parentheses. Significance level *** p < 0.01, ** p < 0.05, * p < 0.1. Respective first-stage IV regression in Table A11. Source: World Bank's PovcalNet database and IMF. Coverage: 2004–2018 period.

Growth-redistribution decomposition of those living under \$1.90 dollars a day: the role of financial inclusion using an alternative measure of financial inclusion. First-stage IV specification.

	(1	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)	(13)	(14)	(15)	(16)	(17)	(18)	(19)	(20)	(21)	(22)	(23)	(24)
	ΔHeadcou	nt	ΔPoverty g	ар	ΔPoverty :	gap squared	∆Watts		ΔHeadcount		ΔPoverty gap		ΔPoverty gap	squared	ΔWatts		ΔHeadcount		ΔPoverty gap		ΔPoverty gap s	squared	ΔWatts	
Dependent variable	e: AAccount ownership	$\Delta Account$ ownership × ΔGir	ΔAccount ii ownership	ΔAccount ownership × 2	ΔAccount ΔGini ownership	$\begin{array}{l} \Delta Account \\ ownership \times \Delta Gini \end{array}$	∆Account i ownership	$\begin{array}{l} \Delta Account \\ ownership \times \Delta Gini \end{array}$	ΔAccount ownership male	$\Delta Account$ ownership male $\times \Delta Gini$	ΔAccount ownership male	$\Delta Account$ ownership male × $\Delta Gini$	ΔAccount ownership male	$\Delta Account$ ownership male × $\Delta Gini$	ΔAccount ownership male	$\Delta Account$ ownership male × $\Delta Gini$	ΔAccount ownership female	$\Delta Account$ ownership female × $\Delta Gini$	ΔAccount ownership female	$\Delta Account$ ownership female × $\Delta Gini$	ΔAccount ownership female	$\Delta Account$ ownership female × $\Delta Gini$	ΔAccount ownership female	ΔAccount ownership female × ΔGini
Mobile subscription	n 0.020***	-0.000	0.020***	-0.000	0.020***	-0.000	0.020***	-0.000	0.020***	-0.000	0.020***	-0.000	0.020***	-0.000	0.020***	-0.000	0.019***	-0.000	0.019***	-0.000	0.019***	-0.000	0.019***	-0.000
lagged 2	(0.005)	(0.000)	(0.005)	(0.000)	(0.005)	(0.000)	(0.005)	(0.000)	(0.007)	(0.000)	(0.007)	(0.000)	(0.007)	(0.000)	(0.007)	(0.000)	(0.005)	(0.000)	(0.005)	(0.000)	(0.005)	(0.000)	(0.005)	(0.000)
periods																								
Origin of legal syster	m 																							
French	-0.011***	0.000	-0.011***	0.000	-0.011***	0.000	-0.011***	0.000	-0.012***	0.000	-0.012***	0.000	-0.012***	0.000	-0.012***	0.000	-0.011**	0.000	-0.011**	0.000	-0.011**	0.000	-0.011**	0.000
Contaliet	(0.004)	(0.000)	0.004)	(0.000)	(0.004)	(0.000)	(0.004)	(0.000)	(0.004)	(0.000)	(0.004)	(0.000)	(0.004)	(0.000)	(0.004)	(0.000)	(0.004)	(0.000)	(0.004)	(0.000)	(0.004)	(0.000)	(0.004)	(0.000)
Socialist	(0.007)	(0.000)	(0.007)	(0.000)	(0.007)	(0.000)	(0.007)	(0.000)	(0.005)	(0.000)	(0.005)	(0.000)	(0.005)	(0.000)	(0.005)	(0.000)	(0.000)	-0.000	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
Airporte por	0.007)	0.000)	(0.007)	0.000)	(0.007)	(0.000)	0.007)	0.000)	(0.003)	(0.000)	0.003)	(0.000)	0.003)	0.000	(0.003)	0.000)	(0.003)	0.001	(0.003)	(0.000)	(0.003)	0.001	0.003)	0.000
kilometre ²	(0.018)	(0.001)	(0.018)	(0.001)	(0.018)	(0.001)	(0.018)	(0.001)	(0.021)	(0.001)	(0.021)	(0.001)	(0.021)	(0.001)	(0.021)	(0.001)	(0.070)	(0.001)	(0.070)	(0.001)	(0.020)	(0.001)	(0.070)	(0.001)
AGini	-0.098	0.020	-0.098	0.020	-0.098	0.020	-0.098	0.020	-0.017	0.016	-0.017	0.016	-0.017	0.016	-0.012	0.016	-0.178	0.024*	-0.178	0.024*	-0.178	0.024*	-0.178	0.024*
Lionn	(0.150)	(0.013)	(0.150)	(0.013)	(0.150)	(0.013)	(0.150)	(0.013)	(0.159)	(0.017)	(0.159)	(0.017)	(0.159)	(0.012)	(0.159)	(0.012)	(0.158)	(0.014)	(0.158)	(0.014)	(0.158)	(0.014)	(0.158)	(0.014)
GDP growth rate	0.081*	0.002	0.081*	0.002	0.081*	0.002	0.081*	0.002	0.086*	0.002	0.086*	0.002	0.086*	0.002	0.086*	0.002	0.079*	0.002	0.079*	0.002	0.079*	0.002	0.079*	0.002
	(0.043)	(0.002)	(0.043)	(0.002)	(0.043)	(0.002)	(0.043)	(0.002)	(0.044)	(0.002)	(0.044)	(0.002)	(0.044)	(0.002)	(0.044)	(0.002)	(0.045)	(0.002)	(0.045)	(0.002)	(0.045)	(0.002)	(0.045)	(0.002)
Constant	-0.006	-0.000	-0.005	-0.000	-0.005	-0.000	-0.006	-0.000	-0.005	-0.000	-0.005	-0.000	-0.005	-0.000	-0.005	-0.000	-0.005	-0.000	-0.005	-0.000	-0.005	-0.000	-0.005	-0.000
	(0.004)	(0.000)	(0.004)	(0.000)	(0.004)	(0.000)	(0.004)	(0.000)	(0.005)	(0.000)	(0.005)	(0.000)	(0.005)	(0.000)	(0.005)	(0.000)	(0.004)	(0.000)	(0.004)	(0.000)	(0.004)	(0.000)	(0.004)	(0.000)
Year fixed effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Observations	400	400	400	400	400	400	400	400	400	400	400	400	400	400	400	400	400	400	400	400	400	400	400	400
R-squared	0.428	0.128	0.428	0.128	0.428	0.128	0.428	0.128	0.425	0.106	0.425	0.106	0.425	0.106	0.425	0.106	0.394	0.145	0.394	0.145	0.394	0.145	0.394	0.145
-																								

Note: Δ indicates change from one period to the next. Robust standard errors, clustered at the country level in parentheses. Significance level *** p < 0.01, ** p < 0.05, * p < 0.1. Respective second-stage IV random effects regression in Table 6. Source: World Bank's PovcalNet database and Global Findex. Coverage: 2011–2018 period.

_

Growth-redistribution decomposition of those living under \$1.90 dollars a day: the role of financial inclusion using an alternative measure of financial inclusion. Standardised coefficients of second-stage IV random effects specification.

Dependent variable:	(1) ∆Headcount	(2) ∴ ∆Poverty gap	(3) ΔPoverty gap squared	(4) ∆Watts	(5) ∆Headcount	(6) ∆Poverty gap	(7) ΔPoverty gap squared	(8) ∆Watts	(9) ∆Headcount	(10) ∆Poverty gap	(11) ΔPoverty gap squared	(12) ∆Watts
ΔGini	1.080*** (0.346)	1.239*** (0.388)	1.236*** (0.387)	1.287*** (0.403)	0.939*** (0.321)	1.041*** (0.343)	1.030*** (0.338)	1.076*** (0.357)	1.122*** (0.348)	1.327*** (0.411)	1.340*** (0.421)	1.391*** (0.435)
GDP growth rate	0.188	0.194 (0.226)	0.177 (0.222)	0.184 (0.230)	0.160	0.152 (0.181)	0.135	0.141 (0.183)	0.183	0.199	0.186 (0.238)	0.193
∆Account ownership	-0.480 (0.598)	-0.381 (0.695)	-0.293 (0.707)	-0.307 (0.728)				((,	
Δ Gini × Δ Account ownership	-1.251* (0.676)	-1.460* (0.834)	-1.445* (0.855)	-1.488*								
ΔAccount ownership male			. ,	. ,	-0.492 (0.512)	-0.399 (0.576)	-0.312 (0.587)	-0.326 (0.604)				
Δ Gini × Δ Account ownership male					-0.987* (0.572)	-1.060^{*} (0.621)	-1.024* (0.608)	-1.058*				
ΔAccount ownership female						. ,		. ,	-0.454 (0.673)	-0.364 (0.815)	-0.274 (0.837)	-0.287 (0.860)
Δ Gini × Δ Account ownership female									-1.226** (0.517)	-1.515** (0.689)	-1.532** (0.739)	-1.573** (0.750)
Constant	-0.637* (0.371)	-0.552 (0.363)	-0.453 (0.337)	-0.473 (0.350)	-0.616* (0.343)	-0.524* (0.318)	-0.423 (0.289)	-0.443 (0.300)	-0.627* (0.375)	-0.553 (0.383)	-0.457 (0.363)	-0.477 (0.376)
Year fixed effects Observations	Yes 400	Yes 400	Yes 400	Yes 400	Yes 400	Yes 400	Yes 400	Yes 400	Yes 400	Yes 400	Yes 400	Yes 400
Number of countries	55	33	55	33	55	33	55	55	55	32	33	55

Note: Δ indicates change from one period to the next. Robust standard errors, clustered at the country level in parentheses. Significance level *** p < 0.01, ** p < 0.05, * p < 0.1. Respective first-stage IV regression (of non-standardised coefficients) in Table A13. Source: World Bank's PovcalNet database and Global Findex. Coverage: 2011–2018 period.

Growth-redistribution decomposition of those living under \$3.20 dollars a day: the role of financial inclusion using an alternative measure of financial inclusion. First-stage IV specification.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)	(13)	(14)	(15)	(16)	(17)	(18)	(19)	(20)	(21)	(22)	(23)	(24)
	ΔHeadcour	nt	ΔPoverty ga	ıp	ΔPoverty :	sap squared	ΔWatts		ΔHeadcount		ΔPoverty gap		ΔPoverty gap	squared	ΔWatts		ΔHeadcount		Δ Poverty gap		ΔPoverty gap	squared	ΔWatts	
Dependent variable:	ΔAccount ownership	$\begin{array}{l} \Delta Account \\ ownership \times \Delta Gini \end{array}$	ΔAccount ownership	$\Delta Account$ ownership × 2	ΔAccount ΔGini ownership	$\label{eq:shift} \begin{split} \Delta Account \\ ownership \times \Delta Gini \end{split}$	ΔAccount ownership	$\begin{array}{l} \Delta Account \\ ownership \times \Delta Gini \end{array}$	ΔAccount ownership male	ΔAccount ownership male × ΔGini	ΔAccount ownership male	$\Delta Account$ ownership male × $\Delta Gini$	ΔAccount ownership male	ΔAccount ownership male × ΔGini	ΔAccount ownership male	$\begin{array}{l} \Delta Account \\ ownership \\ male \times \Delta Gini \end{array}$	ΔAccount ownership female	$\Delta Account$ ownership female × $\Delta Gini$	ΔAccount ownership female	ΔAccount ownership female × ΔGini	ΔAccount ownership female	ΔAccount ownership female × ΔGini	ΔAccount ownership female	$\Delta Account$ ownership female $\times \Delta Gini$
Mobile subscription lagged 2 periods	0.019*** (0.005)	-0.000 (0.000)	0.019*** (0.005)	-0.000 (0.000)	0.019*** (0.005)	-0.000 (0.000)	0.019*** (0.005)	-0.000 (0.000)	0.019*** (0.007)	-0.000 (0.000)	0.019*** (0.007)	-0.000 (0.000)	0.019*** (0.007)	-0.000 (0.000)	0.019*** (0.007)	-0.000 (0.000)	0.019*** (0.005)	-0.000 (0.000)	0.019*** (0.005)	-0.000 (0.000)	0.019*** (0.005)	-0.000 (0.000)	0.019*** (0.005)	-0.000 (0.000)
Origin of legal system French	-0.010** (0.004)	0.000 (0.000)	-0.010** (0.004)	0.000	-0.010** (0.004)	0.000	-0.010** (0.004)	0.000 (0.000)	-0.011** (0.004)	0.000	-0.011** (0.004)	0.000	-0.011** (0.004)	0.000	-0.011** (0.004)	0.000 (0.000)	-0.010** (0.004)	0.000	-0.010** (0.004)	0.000 (0.000)	-0.010** (0.004)	0.000	-0.010** (0.004)	0.000 (0.000)
Socialist	-0.027*** (0.007)	0.000 (0.000)	-0.027*** (0.007)	0.000 (0.000)	-0.027*** (0.007)	0.000 (0.000)	-0.027*** (0.007)	0.000 (0.000)	-0.033*** (0.005)	0.000 (0.000)	-0.033*** (0.005)	0.000 (0.000)	-0.033*** (0.005)	0.000 (0.000)	-0.033*** (0.005)	0.000 (0.000)	-0.022** (0.009)	-0.000 (0.000)	-0.022** (0.009)	-0.000 (0.000)	-0.022** (0.009)	-0.000 (0.000)	-0.022** (0.009)	-0.000 (0.000)
kilometre ² ΔGini	(0.018) -0.094	-0.002 (0.001) 0.020	-0.017 (0.018) -0.094	-0.002 (0.001) 0.020	(0.018) -0.094	-0.002 (0.001) 0.020	-0.017 (0.018) -0.094	(0.001) 0.020	-0.021 (0.022) -0.008	-0.002 (0.001) 0.016	-0.021 (0.022) -0.008	(0.001) 0.016	-0.021 (0.022) -0.008	(0.001) 0.016	-0.021 (0.022) -0.008	-0.002 (0.001) 0.016	-0.010 (0.021) -0.175	-0.001 (0.001) 0.024*	-0.010 (0.021) -0.175	-0.001 (0.001) 0.024*	(0.021) -0.175	-0.001 (0.001) 0.024*	-0.010 (0.021) -0.175	(0.001) 0.024*
GDP growth rate	(0.151) 0.084* (0.046)	(0.013) 0.002 (0.002)	(0.151) 0.084* (0.046)	(0.013) 0.002 (0.002)	(0.151) 0.084* (0.046)	(0.013) 0.002 (0.002)	(0.151) 0.084* (0.046)	(0.013) 0.002 (0.002)	(0.160) 0.090* (0.048)	(0.012) 0.002 (0.002)	(0.160) 0.090* (0.048)	(0.012) 0.002 (0.002)	(0.160) 0.090* (0.048)	(0.012) 0.002 (0.002)	(0.160) 0.090* (0.048)	(0.012) 0.002 (0.002)	(0.158) 0.083* (0.048)	(0.014) 0.003 (0.002)	(0.158) 0.083* (0.048)	(0.014) 0.003 (0.002)	(0.158) 0.083* (0.048)	(0.014) 0.003 (0.002)	(0.158) 0.083* (0.048)	(0.014) 0.003 (0.002)
Constant	-0.006 (0.005)	-0.000 (0.000)	-0.006 (0.005)	-0.000 (0.000)	-0.006 (0.005)	-0.000 (0.000) Vet	-0.006 (0.005)	-0.000 (0.000) Vec	-0.006 (0.005)	-0.000 (0.000)	-0.006 (0.005)	-0.000 (0.000)	-0.006 (0.005)	-0.000 (0.000) Voc	-0.006 (0.005)	-0.000 (0.000)	-0.007 (0.005)	-0.000 (0.000)	-0.007 (0.005)	-0.000 (0.000)	-0.007 (0.005)	-0.000 (0.000)	-0.007 (0.005)	-0.000 (0.000)
Observations R-squared	391 0.436	391 0.129	391 0.436	391 0.129	391 0.436	391 0.129	391 0.436	391 0.129	391 0.433	391 0.106	391 0.433	391 0.106	391 0.433	391 0.106	391 0.433	391 0.106	391 0.401	391 0.146	391 0.401	391 0.146	391 0.401	391 0.146	391 0.401	391 0.146

Note: Δ indicates change from one period to the next. Robust standard errors, clustered at the country level in parentheses. Significance level *** p < 0.01, ** p < 0.05, * p < 0.1. Respective second-stage IV random effects regression in Table A16. Source: World Bank's PovcalNet database and Global Findex. Coverage: 2011–2018 period.

_

Growth-redistribution decomposition of those living under \$3.20 dollars a day in low- and low-middle income countries: the role of financial inclusion. Second-stage IV random effects specification.

Dependent variable:	(1) ∆Headcount	(2) Σ ΔPoverty gap	(3) ΔPoverty gap squared	(4) ∆Watts	(5) ∆Headcount	(6) ∆Poverty gap	(7) ΔPoverty gap squared	(8) ∆Watts	(9) ∆Headcount	(10) ∆Poverty gap	(11) ΔPoverty gap squared	(12) ∆Watts
ΔGini	1.050** (0.411)	1.316*** (0.419)	1.165*** (0.367)	3.193*** (1.007)	1.048** (0.420)	1.156*** (0.396)	0.993*** (0.333)	2.729*** (0.921)	0.962** (0.421)	1.346*** (0.417)	1.223*** (0.377)	3.347*** (1.034)
GDP growth rate	0.059 (0.097)	0.106 (0.115)	0.091 (0.101)	0.239 (0.273)	0.066 (0.102)	0.093 (0.102)	0.075 (0.085)	0.197 (0.230)	0.042 (0.088)	0.099 (0.112)	0.089 (0.102)	0.232 (0.276)
ΔAccount ownership	-0.322 (0.272)	-0.245 (0.319)	-0.173 (0.284)	-0.448 (0.778)								
ownership	(9.240)	(13.113)	-22.834 (12.699)	(34.120)	0.226	0.225	0.165	0 425				
ownership male					-0.556 (0.268) -18 246*	-0.255 (0.266) -20.424*	-0.165 (0.228) -17 322*	-0.425 (0.627) -46.687^*				
ownership male					(10.273)	(11.587)	(10.148)	(27.127)	-0.296	-0.226	-0.159	-0.406
ownership female									(0.261)	(0.349)	(0.322)	(0.879)
$\Delta Gini \times \Delta Account$ ownership female									-11.090 (8.961)	-23.139** (9.520)	-22.178** (9.771)	-59.511** (26.333)
Constant	-0.018^{*} (0.009)	-0.017* (0.009)	-0.013* (0.008)	-0.033* (0.020)	-0.018* (0.009)	-0.016* (0.009)	-0.012* (0.007)	-0.031* (0.018)	-0.017* (0.009)	-0.016* (0.009)	-0.013* (0.008)	-0.033 (0.020)
Year fixed effects Observations	Yes 391	Yes 391	Yes 391	Yes 391	Yes 391	Yes 391	Yes 391	Yes 391	Yes 391	Yes 391	Yes 391	Yes 391
Number of countries	52	52	52	52	52	52	52	52	52	52	52	52
Test of overidentifying restrictions:												
Sargan-Hansen statistic Chi-so	1.807	0.317	1.031	0.921	0.991	1.588	2.982	2.722	2.815	0.458	0.373	0.304
Sargan-Hansen statistic P-value	0.405	0.853	0.597	0.631	0.609	0.452	0.225	0.256	0.245	0.796	0.830	0.859
Endogenity test: Chi-sq	6.020	10.260	9.630	9.030	5.440	10.440	10.250	9.540	6.270	9.530	8.970	8.460
P-value	0.049	0.006	0.008	0.011	0.066	0.005	0.006	0.009	0.044	0.009	0.011	0.015

Note: Δ indicates change from one period to the next. Robust standard errors, clustered at the country level in parentheses. Significance level *** p < 0.01, ** p < 0.05, * p < 0.1. Respective first-stage IV regression in Table A15. Source: World Bank's PovcalNet database and IMF. Coverage: 2004–2018 period.

Growth-redistribution decom	position of those living	under \$5.50 dollars a d	av: the role of financial inclusion u	sing an alternative measure of	f financial inclusion. First-sta	ge IV specification.
		· · · · · · · · · · · · · · · · · · ·	· · · · · · · · · · · · · · · · · · ·			0

			-			-		-					-											
_	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)	(13)	(14)	(15)	(16)	(17)	(18)	(19)	(20)	(21)	(22)	(23)	(24)
	ΔHeadcour	ıt	ΔPoverty g	ър	ΔPoverty g	ap squared	ΔWatts		ΔHeadcount		Δ Poverty gap		ΔPoverty gap	squared	ΔWatts		∆Headcount		ΔPoverty gap		ΔPoverty gap	quared	ΔWatts	
Dependent variable:	ΔAccount ownership	$\Delta Account \\ ownership \times \Delta Gi$	ΔAccount ini ownership	$\begin{array}{l} \Delta Account \\ ownership \times \Delta Gini \end{array}$	ΔAccount ownership	$\begin{array}{l} \Delta Account \\ ownership \times \Delta Gini \end{array}$	AAccount ownership	$\label{eq:shift} \begin{split} \Delta & \text{Account} \\ & \text{ownership} \times \Delta & \text{Gini} \end{split}$	ΔAccount ownership male	ΔAccount ownership male × ΔGini	ΔAccount ownership male	$\Delta Account$ ownership male × $\Delta Gini$	ΔAccount ownership male	$\Delta Account$ ownership male × $\Delta Gini$	ΔAccount ownership male	$\Delta Account$ ownership male × $\Delta Gini$	ΔAccount ownership female	$\begin{array}{l} \Delta Account \\ ownership \\ female \times \Delta Gini \end{array}$	ΔAccount ownership female	$\begin{array}{l} \Delta Account \\ ownership \\ female \times \Delta Gini \end{array}$	ΔAccount ownership female	$\begin{array}{l} \Delta Account \\ ownership \\ female \times \Delta Gini \end{array}$	ΔAccount ownership female	ΔAccount ownership female × ΔGini
Mobile subscription lagged 2 periods	0.019*** a (0.005)	-0.000 (0.000)	0.019*** (0.005)	-0.000 (0.000)	0.019*** (0.005)	-0.000 (0.000)	0.019*** (0.005)	-0.000 (0.000)	0.019*** (0.007)	-0.000 (0.000)	0.019*** (0.007)	-0.000 (0.000)	0.019*** (0.007)	-0.000 (0.000)	0.019*** (0.007)	-0.000 (0.000)	0.019*** (0.005)	-0.000 (0.000)	0.019*** (0.005)	-0.000 (0.000)	0.019*** (0.005)	-0.000 (0.000)	0.019*** (0.005)	-0.000 (0.000)
Origin of legal system																								
French	-0.010**	0.000	-0.010**	0.000	-0.010**	0.000	-0.010**	0.000	-0.011**	0.000	-0.011**	0.000	-0.011**	0.000	-0.011**	0.000	-0.010**	0.000	-0.010**	0.000	-0.010**	0.000	-0.010**	0.000
Socialist	-0.027***	0.000	-0.027***	0.000	-0.027***	0.000	-0.027***	0.000	-0.033***	0.000	-0.033***	0.000	-0.033***	0.000	-0.033***	0.000	-0.022**	-0.000	-0.022**	-0.000	-0.022**	-0.000	-0.022**	-0.000
Airports	(0.007)	-0.002	-0.017	-0.002	(0.007) -0.017	(0.000)	(0.007)	(0.000) 0.002	(0.005) 0.021	(0.000) 0.002*	-0.021	(0.000) 0.002*	(0.005) -0.021	(0.000) 0.002*	(0.005) -0.021	(0.000) 0.002*	(0.009) 0.010	(0.000) 0.001	(0.009)	(0.000) 0.001	(0.009) 0.010	(0.000) -0.001	(0.009) 0.010	-0.001
per kilometre ²	(0.018)	(0.001)	(0.018)	(0.001)	(0.018)	(0.001)	(0.018)	(0.001)	(0.022)	(0.001)	(0.022)	(0.001)	(0.022)	(0.001)	(0.022)	(0.001)	(0.021)	(0.001)	(0.021)	(0.001)	(0.021)	(0.001)	(0.021)	(0.001)
ΔGini	-0.094	0.020	-0.094	0.020	-0.094	0.020	-0.094	0.020	-0.008	0.016	-0.008	0.016	-0.008	0.016	-0.008	0.016	-0.175	0.024*	-0.175	0.024*	-0.175	0.024*	-0.175	0.024*
	(0.151)	(0.013)	(0.151)	(0.013)	(0.151)	(0.013)	(0.151)	(0.013)	(0.160)	(0.012)	(0.160)	(0.012)	(0.160)	(0.012)	(0.160)	(0.012)	(0.158)	(0.014)	(0.158)	(0.014)	(0.158)	(0.014)	(0.158)	(0.014)
GDP growth rate	(0.084*	(0.002)	(0.084*	(0.002)	(0.084*	(0.002)	(0.084*	(0.002)	(0.048)	(0.002)	(0.048)	(0.002)	(0.048)	(0.002)	(0.048)	(0.002)	(0.048)	(0.003	(0.083*	(0.003)	(0.048)	(0.002)	(0.048)	(0.003)
Constant	-0.006	-0.000	-0.006	-0.000	-0.006	-0.000	-0.006	-0.000	-0.006	-0.000	-0.006	-0.000	-0.006	-0.000	-0.006	-0.000	-0.007	-0.000	-0.007	-0.000	-0.007	-0.000	-0.007	-0.000
Year fixed effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Observations R-squared	391 0.436	391 0.129	391 0.436	391 0.129	391 0.436	391 0.129	391 0.436	391 0.129	391 0.433	391 0.106	391 0.433	391 0.106	391 0.433	391 0.106	391 0.433	391 0.106	391 0.401	391 0.146	391 0.401	391 0.146	391 0.401	391 0.146	391 0.401	391 0.146

Note: Δ indicates change from one period to the next. Robust standard errors, clustered at the country level in parentheses. Significance level *** p < 0.01, ** p < 0.05, * p < 0.1. Respective second-stage IV random effects regression in Table A18. Source: World Bank's PovcalNet database and Global Findex. Coverage: 2011–2018 period.

Growth-redistribution decomposition of those living under \$5.50 dollars a day in low- and low-middle income countries: the role of financial inclusion. Second-stage IV random effects specification.

Dependent variable	(1) ∆Headcount	(2) ∆Poverty gap	(3) ΔPoverty gap squared	(4) ∆Watts	(5) ∆Headcount	(6) ΔPoverty gap	(7) ΔPoverty gap squared	(8) ∆Watts	(9) ∆Headcount	(10) ∆Poverty gap	(11) ∆Poverty gap squared	(12) ∆Watts
ΔGini	0.157 (0.364)	0.971*** (0.340)	1.147*** (0.367)	3.478*** (1.107)	0.311 (0.288)	0.936*** (0.348)	1.032*** (0.358)	3.093*** (1.066)	-0.021 (0.471)	0.921*** (0.341)	1.152*** (0.362)	3.529*** (1.106)
GDP growth rate	-0.036 (0.066)	0.059 (0.086)	0.086	0.243 (0.296)	-0.017 (0.066)	0.062 (0.088)	0.078 (0.092)	0.215 (0.268)	-0.049 (0.070)	0.047 (0.079)	0.077 (0.095)	0.223 (0.288)
ownership	-0.232 (0.179)	-0.250 (0.239)	-0.228 (0.277)	-0.599 (0.840)								
ownership	3.052 (16.796)	-15.667 (7.490)	(10.279)	(31.634)								
ΔAccount ownership male					-0.233 (0.187)	-0.258 (0.227)	-0.224 (0.240)	-0.586 (0.715)				
ownership male					-4.049 (12.324)	(8.326)	(9.732)	(28.286)				
ΔAccount ownership									-0.247 (0.190)	-0.226 (0.236)	-0.207 (0.295)	-0.537 (0.908)
$\Delta Gini \times \Delta Account$ ownership female									8.875 (19.343)	-11.951* (6.437)	-18.522** (7.298)	-57.142** (22.966)
Constant	-0.008 (0.005)	-0.015* (0.008)	-0.015* (0.008)	-0.040^{*} (0.023)	-0.009* (0.005)	-0.015* (0.008)	-0.014* (0.008)	-0.038* (0.021)	-0.007 (0.006)	-0.014* (0.007)	-0.014* (0.008)	-0.039* (0.023)
Year fixed effects Observations	Yes 391	Yes 391	Yes 391	Yes 391	Yes 391	Yes 391	Yes 391	Yes 391	Yes 391	Yes 391	Yes 391	Yes 391
Number of countries	52	52	52	52	52	52	52	52	52	52	52	52
Test of overidentifying restrictions:												
Sargan-Hansen statistic Chi-so	5.530	0.936	0.125	0.146	5.489	0.196	0.676	0.985	5.442	2.043	0.691	0.476
Sargan-Hansen statistic P-value	0.063	0.626	0.940	0.930	0.064	0.907	0.713	0.611	0.066	0.360	0.708	0.788
Endogenity test: Chi-sq	3.500	6.450	8.910	8.070	5.640	5.960	8.820	8.160	2.130	6.350	8.400	7.670
P-value	0.174	0.040	0.012	0.018	0.060	0.051	0.012	0.017	0.345	0.042	0.015	0.022

Note: Δ indicates change from one period to the next. Robust standard errors, clustered at the country level in parentheses. Significance level *** p < 0.01, ** p < 0.05, * p < 0.1. Respective first-stage IV regression in Table A17. Source: World Bank's PovcalNet database and IMF. Coverage: 2004–2018 period.



Fig. A1. Forecast poverty headcount \$3.20 a day, all countries.



Fig. A2. Forecast poverty headcount \$5.50 a day, all countries.



Fig. A3. Forecast poverty headcount \$3.20 a day, including low- and lower-middleincome countries only using IMF forecast GDP growth.



Fig. A4. Forecast poverty headcount \$5.50 a day, including low- and lower-middleincome countries only using IMF forecast GDP growth.

Appendix B. Supplementary data

Supplementary data to this article can be found online at https://doi.org/10.1016/j.worlddev.2020.105229.

References

- Ahamed, M. M., & Mallick, S. K. (2019). Is financial inclusion good for bank stability? International evidence. *Journal of Economic Behavior and Organization*, 157, 403–427.
- Aker, J. C., & Mbiti, I. M. (2010). Mobile phones and economic development in Africa. Journal of Economic Perspectives, 24(3), 207–232. https://doi.org/ 10.1257/jep.24.3.207.
- Allen, F., Carletti, E., Cull, R., Qian, J. 'QJ', Senbet, L., & Valenzuela, P. (2014). The African financial development and financial inclusion gaps. *Journal of African Economies*, 23(5), 614–642. https://doi.org/10.1093/jae/eju015.
- Amidžic, G., Massara, A., & Mialou, A. (2014). Assessing countries' financial inclusion standing - A new composite index (IMF working paper series No. 14/36).
- Atamanov, A., Azevedo, J. P. W. De, Castaneda Aguilar, R. A., Chen, S., Corral Rodas, P. A., Dewina, R., ... Zhao, Q. (2018). April 2018 povcalnet update : What's new. (World Bank).
- Banerjee, A., Duflo, E., Glennerster, R., & Kinnan, C. (2015). The miracle of microfinance? Evidence from a randomized evaluation. *American Economic Journal: Applied Economics*, 7(1), 22–53. https://doi.org/10.1257/app.20130533.
- Banerjee, A., Karlan, D., & Zinman, J. (2015). Six randomized evaluations of microcredit: Introduction and further steps. *American Economic Journal: Applied Economics*, 7(1), 1–21. https://doi.org/10.1257/app.20140287.

- Beck, T., Demirgüç-Kunt, A., & Levine, R. (2007). Finance, inequality and the poor. Journal of Economic Growth, 12(1), 27–49. https://doi.org/10.1007/s10887-007-9010-6.
- Beck, T., Demirgüç-Kunt, A., & Martinez Peria, M. S. (2007). Reaching out: Access to and use of banking services across countries. *Journal of Financial Economics*, 85 (1), 234–266. https://doi.org/10.1016/j.jfineco.2006.07.002.
- Beck, T., Levine, R., & Loayza, N. (2000). Finance and the sources of growth. *Journal of Financial Economics*, 58(1-2), 261-300. https://doi.org/10.1016/s0304-405x(00) 00072-6.
- Bell, A., & Jones, K. (2015). Explaining fixed effects: Random effects modeling of time-series cross-sectional and panel data. *Political Science Research and Methods*, 3(1), 133–153. https://doi.org/10.1017/psrm.2014.7.
- Birch, S., Buril, F., Cheeseman, N., Clark, A., Darnolf, S., Dodsworth, S., Garber, L., Gutiérrez-Romero, R., Hollstein, T., James, T. S., Mohan, V., & Sawyer, K. (2020). How to hold elections safely and democratically during the COVID-19 pandemic. *The British Academy*.
- Burgess, R., & Pande, R. (2005). Do rural banks matter? Evidence from the Indian social banking experiment. American Economic Review, 95(3), 780–795. https:// doi.org/10.1257/0002828054201242.
- Cai, H., Chen, Y., Fang, H., & Zhou, L.-A. (2009). Microinsurance, trust and economic development: evidence from a randomized natural field experiment (National Bureau of Economic Research). https://doi.org/10.3386/w15396.
- Chibba, M. (2009). Financial inclusion, poverty reduction and the millennium development goals. European Journal of Development Research, 21(2), 213–230. https://doi.org/10.1057/ejdr.2008.17.
- Churchill, S. A., & Marisetty, V. B. (2020). Financial inclusion and poverty: A tale of forty-five thousand households. *Applied Economics*, 52(16), 1777–1788. https:// doi.org/10.1080/00036846.2019.1678732.
- Cohen, J. (1988). Statistical power analysis for the behavioral sciences (2nd ed.). NJ: Erlbaum.: Hillsdale.
- Cole, S. (2009). Fixing market failures or fixing elections? Agricultural credit in India. American Economic Journal: Applied Economics, 1(1), 219–250. https://doi. org/10.1257/app.1.1.219.
- Datt, G., & Ravallion, M. (1992). Growth and redistribution components of changes in poverty measures. A decomposition with applications to Brazil and India in the 1980s. *Journal of Development Economics*, *38*(2), 275–295.
- Demirgüç-Kunt, A., Klapper, L., Singer, D., Ansar, S., & Hess, J. (2018). The global findex database 2017: Measuring financial inclusion and the fintech revolution. Washington D.C.
- Dercon, S. (2005). Risk, insurance, and poverty: A review. In Insurance Against Poverty. https://doi.org/10.1093/0199276838.003.0002.
- Dimova, R., & Adebowale, O. (2018). Does access to formal finance matter for welfare and inequality? Micro level evidence from Nigeria. *Journal of Development Studies*, 54(9), 1534–1550. https://doi.org/10.1080/ 00220388.2017.1371293.
- Dupas, P., & Robinson, J. (2013). Savings constraints and microenterprise development: Evidence from a field experiment in kenya. *American Economic Journal: Applied Economics*, 5(1), 163–192. https://doi.org/10.1257/ app.5.1.163.
- Duvendack, M., Palmer-Jones, R., & Vaessen, J. (2014). Meta-analysis of the impact of microcredit on women's control over household decisions: Methodological issues and substantive findings. *Journal of Development Effectiveness*, 6(2), 73–96. https://doi.org/10.1080/19439342.2014.903289.
- Flory, J. A. (2018). Formal finance and informal safety nets of the poor: Evidence from a savings field experiment. *Journal of Development Economics*, 135, 517–533. https://doi.org/10.1016/j.jdeveco.2018.07.015.
- Freije, S. (2014). Changes in poverty and inequality in Latin America during the great recession. In M. Grosh, M. Bussolo, & S. Freije (Eds.), Understanding the Poverty Impact of the Global Financial Crisis in Latin America and the Caribbean (pp. 59–114). Washington, DC: United States: World Bank.
- Gentilini, U., Almenfi, M., Dale, P., Blomquist, J., Palacios, R., Desai, V., & Moreira, V. (2020). Social protection and jobs responses to COVID-19: A real-time review of country measures (No. Living paper, Version 9, 15 May.). Retrieved from https://www.ugogentilini.net/wp-content/uploads/2020/05/Country-SP-COVID-responses_May15.pdf.
- Goksu, A., Deléchat, C., Newiak, M., & Yang, F. (2017). Inequality in financial inclusion and income inequality (IMF working paper series No. 236).
- Gutiérrez-Romero, R. (2020). Conflict in Africa during COVID-19: Social Distancing, Food Vulnerability and Welfare Response (Centre for Globalisation Research (GCR) working paper series). https://doi.org/10.2139/ssrn.3616421>.
- Gutiérrez-Romero, R., & Méndez-Errico, L. (2017). Does inequality foster or hinder the growth of entrepreneurship in the long run? *Research on Economic Inequality*, 25, 299–341. https://doi.org/10.1108/S1049-258520170000025009.
- Gutiérrez-Romero, R. (2021). Inequality, persistence of the informal economy, and club convergence. *World Development*. forthcoming.
- IMF. (2019). Financial access survey 2019 trends and developments.
- IMF. (2020a). World economic outlook. IMF.
- IMF. (2020b). World economic outlook update, June 2020: A crisis like no other, an uncertain recovery.
- Koomson, I., Villano, R. A., & Hadley, D. (2020). Effect of financial inclusion on poverty and vulnerability to poverty: Evidence using a multidimensional measure of financial inclusion. Social Indicators Research, 1–27. https://doi.org/ 10.1007/s11205-019-02263-0.
- La Porta, R., Lopez-de-Silanes, F., & Shleifer, A. (2008). The economic consequences of legal origins. *Journal of Economic Literature*, 46(2), 285–332. https://doi.org/ 10.1257/jel.46.2.285.

- La Porta, R., Lopez-De-Silanes, F., Shleifer, A., & Vishny, R. W. (1997). Legal determinants of external finance. *The Journal of Finance*, *52*(3), 1131–1150. https://doi.org/10.1111/j.1540-6261.1997.tb02727.x.
- Li, L. (2018). Financial inclusion and poverty: The role of relative income. China Economic Review, 52, 165–191. https://doi.org/10.1016/j.chieco.2018.07.006.
- Morduch, J. (1999). The microfinance promise. *Journal of Economic Literature*, 37(4), 1569–1614. https://doi.org/10.2307/2565486.
- Park, C.-Y., & Mercado, R. J. V. (2018). Financial inclusion: New measurement and cross-country impact assessment. (ADB Economics Working Paper Series). Retrieved from https://www.adb.org/publications/financial-inclusion-newmeasurement-cross-country-impact-assessment>.
- Pradhan, R. P., Arvin, M. B., & Norman, N. R. (2015). The dynamics of information and communications technologies infrastructure, economic growth, and financial development: Evidence from Asian countries. *Technology in Society*, 42, 135–149. https://doi.org/10.1016/j.techsoc.2015.04.002.
- Prina, S. (2015). Banking the poor via savings accounts: Evidence from a field experiment. Journal of Development Economics, 115, 16–31. https://doi.org/ 10.1016/j.jdeveco.2015.01.004.
- Ravallion, M. (2005). Inequality is bad for the poor (No. 3677). World Bank Policy Research Working Paper. https://dx.doi.org/10.2139/ssrn.780485>.
- Ravallion, M. (2013). How long will it take to lift one billion people out of poverty? Review. World Bank Research Observer, 28(2), 139–159. https://doi.org/10.1093/ wbro/lkt003.

- Sandefur, J., & Subramanian, A. (2020). The IMF's growth forecasts for poor countries don't match its COVID narrative (Cnter for Global Development No. 533).
- Stewart, R., van Rooyen, C., Korth, M., Chereni, A., Rebelo Da Silva, N., & de Wet, T. (2012). Do micro-credit, micro-savings, and micro-leasing serve as effective financial inclusion interventions enabling poor people, and especially women, to engage in meaningful economic opportunities in low-and middle-income countries. A systematic review of th (London: EPPI-Centre, Social Science Research Unit).
- Sumner, A., Hoy, C., & Ortiz-Juarez, E. (2020). Estimates of the impact of COVID-19 on global poverty. (WIDER Working Paper No. 2020/43). Helsinki.
- Suri, T., & Jack, W. (2016). The long-run poverty and gender impacts of mobile money. Science, 354(6317), 1288–1292. https://doi.org/10.1126/science. aah5309.
- van der Weide, R., & Branko, M. (2018). Inequality is bad for growth of the poor (but not for that of the rich). *The World Bank Economic Review*, 32(3), 507–530.
- World Bank (2018). *Clobal financial development report 2017/2018*: Bankers without borders. The World Bank.
- World Bank (2020). June 2020: Global economic prospects. Washington, DC: United States.
- Yang, M., & Stanley, T. D. (2014). Micro-credit and income: A literature review and meta-analysis. (Bulletin of Economics and Meta-Analysis).
- Yunus, M. (2013). Banker to the poor : The story of the Grameen bank. Aurum Press.